COOPER TIRE & RUBBER CO

FORM 10-Q (Quarterly Report)

Filed 11/8/2004 For Period Ending 9/30/2004

Address LIMA & WESTERN AVENUES

FINDLAY, Ohio 45840

Telephone 419-423-1321

CIK 0000024491

Industry Tires

Sector Consumer Cyclical

Fiscal Year 12/31



UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D. C. 20549 **FORM 10-Q**

(X) QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d)

OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2004

OR

() TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d)

OF THE SECURITIES AND EXCHANGE ACT OF 1934

Commission File No. 1-4329

COOPER TIRE & RUBBER COMPANY

(Exact name of registrant as specified in its charter)

DELAWARE (State or other jurisdiction of incorporation or organization)

34-4297750 (I.R.S. employer identification no.)

701 Lima Avenue, Findlay, Ohio 45840 (Address of principal executive offices)

(Zip code)

(419) 423-1321 (Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months, and (2) has been subject to such filing requirements for the past 90 days.

Yes (X) No ()

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act).

Yes (X) No ()

Number of shares of common stock of registrant outstanding at October 29, 2004: 74,866,947

Part I. FINANCIAL INFORMATION

Item 1. FINANCIAL STATEMENTS

COOPER TIRE & RUBBER COMPANY CONDENSED CONSOLIDATED BALANCE SHEETS

(Dollar amounts in thousands except per-share amounts)

	2003	SEPTEMBER 30, 2004 (UNAUDITED)
ASSETS		
Current assets:		
Cash and cash equivalents	\$ 66,426	\$ 24,932
Accounts receivable, less allowances of \$12,569 in 2003 and \$5,699 in 2004 Inventories at lower of cost (last-in, first-out) or market:	613,269	340,683
Finished goods	158,416	170,049
Work in process	35,485	
Raw materials and supplies	88,451	53,368
	282,352	
Prepaid expenses, income taxes refundable and deferred income taxes	62,362	53,303
Assets of discontinued operations		1,414,058
Total current assets		2,071,206
Property, plant and equipment: Land and land improvements	54,104	35 022
Buildings	431,659	35,022 266,811
Machinery and equipment		1,244,556
Molds, cores and rings	178,692	191,232
Less accumulated depreciation and amortization	2,563,246 1,355,348	1,737,621 1,041,157
Net property, plant and equipment Goodwill	1,207,898 429,792	696,464
Intangibles, net of accumulated amortization of \$20,642	429,192	43,224
in 2003 and \$14,289 in 2004 Other assets	47,634 159,134	34,879 109,082
	 \$ 2,868,867	\$ 2,956,855
		=========
LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities:		
Notes payable	\$ 2,770	\$ 161
Accounts payable	267,224	158,937
Accrued liabilities	197,169	
Income taxes	6,549	
Current portion of long-term debt Liabilities of discontinued operations	3,015	371,571
Diabilities of discontinued operations		
Total current liabilities	476,727	713,729 775,592
Long-term debt		
Postretirement benefits other than pensions Other long-term liabilities	220,723 255,580	154,195 202,783
Deferred income taxes	13,500	15,969
Stockholders' equity:		
Preferred stock, \$1 par value; 5,000,000 shares authorized; none issued		
Common stock, \$1 par value; 300,000,000 shares		
authorized; 85,268,000 shares issued in 2003 and 86,320,097 in 2004	85,268	86,320
Capital in excess of par value	24,813	
Retained earnings	1,226,999	
Cumulative other comprehensive loss	(109,679)	(105,224)
Less: 11,303,900 common shares in treasury	1,227,401	1,291,599
at cost in 2003 and 2004	(197,012)	(197,012)
Total stockholders' equity	1,030,389	1,094,587
	\$ 2,868,867 ========	\$ 2,956,855

COOPER TIRE & RUBBER COMPANY CONSOLIDATED STATEMENTS OF INCOME THREE MONTHS ENDED SEPTEMBER 30, 2003 AND 2004 (UNAUDITED)

(Dollar amounts in thousands except per-share amounts)

	2003	2004
Net sales Cost of products sold	\$ 519,171 457,863	\$
Gross profit	61,308	62,141
Selling, general and administrative Adjustments to class action warranty Restructuring	 36,152 (3,900) 45	39,247 (11,273) 8,432
Operating profit	29,011	25,735
Interest expense Other - net	 7,001 (318)	6,580 6
Income from continuing operations before income taxes	22,328	19,149
Provision for income taxes	 7,658	 5,974
Income from continuing operations	14,670	13,175
<pre>Income (loss) from discontinued operations, net of income taxes</pre>	 3,087	(3,305)
Net Income	17,757	9,870
Basic earnings per share: Income from continuing operations Income (loss) from discontinued operations	0.04	 (0.05)
Net Income	0.24	0.13
Diluted earnings per share: Income from continuing operations Income (loss) from discontinued operations	0.20 0.04	0.17
Net Income	\$ 0.24	\$ 0.13
Weighted average number of shares outstanding (000's): Basic	 73,723 ======	 74,928
Diluted	74,293 ======	75,935
Dividends per share	\$	\$ 0.105

See accompanying notes.

COOPER TIRE & RUBBER COMPANY CONSOLIDATED STATEMENTS OF INCOME NINE MONTHS ENDED SEPTEMBER 30, 2003 AND 2004 (UNAUDITED)

(Dollar amounts in thousands except per-share amounts)

	2003	2004
Net sales Cost of products sold	\$ 1,336,115	\$ 1,540,642 1,361,318
Gross profit	152,635	179,324
Selling, general and administrative Adjustments to class action warranty Restructuring	 106,855 (3,900) 2,190	125,622 (11,273) 9,111
Operating profit	47,490	55,864
Interest expense Other - net	 22,419 (976)	20,959 (311)
Income from continuing operations before income taxes	26,047	35,216
Provision for income taxes	 8,934	10,987
Income from continuing operations	17,113	24,229
Income from discontinued operations, net of income taxes	 28,549	43,919
Net Income	45,662	68,148
Basic earnings per share: Income from continuing operations Income from discontinued operations	0.23	0.59
Net Income	\$ 0.62	\$ 0.92
Diluted earnings per share: Income from continuing operations Income from discontinued operations	0.23	0.32 0.58
Net Income	\$ 0.62	\$ 0.90
Weighted average number of shares outstanding (000's): Basic		74,471
Diluted	74,010	75,475
Dividends per share	\$ 0.315	\$

See accompanying notes.

COOPER TIRE & RUBBER COMPANY CONSOLIDATED STATEMENTS OF CASH FLOWS NINE MONTHS ENDED SEPTEMBER 30, 2003 AND 2004 (UNAUDITED)

(Dollar amounts in thousands)

	2003	2004
Operating activities:		
Net income from continuing operations Adjustments to reconcile net income from continuing operations to net cash provided by	\$ 17,113	\$ 24,229
continuing operations:	04.455	04 504
Depreciation	81,166 2,256	
Amortization of intangibles Deferred income taxes	2,256	756
Adjustments to class action warranty	(3,900)	
Restructuring asset write-down	==	9,110
Changes in operating assets and liabilities of continuing operations:		
Accounts receivable Inventories	(112,424)	(16,392)
Prepaid expenses	324	(48,349) (10,108)
Accounts payable	32,098	
Accrued liabilities	33,919	
Other liabilities	10,930	14,349
Not good provided by continuing operations	84,771	
Net cash provided by continuing operations Net cash provided by discontinued operations	38,569	
Net cash provided by discontinued operations		
Net cash provided by operating activities Investing activities:	123,340	236,815
Property, plant and equipment	(67,606)	(96,289)
Acquisition of business, net of cash acquired	(13,111)	
Proceeds from the sale of assets	355	
Net cash used in continuing operations		(96,272)
Net cash used in discontinued operations	(36,397)	(33,456)
Net cash used in investing activities Financing activities:		(129,728)
Payments on long-term debt Net proceeds from (repayments of) borrowings	(2,525)	(92,525)
under credit facilities	26,052	(44,599)
Payment of dividends		(23,481)
Issuance of common shares	2,533	
Net cash provided by (used in) continuing operations	2,869	(145,528)
Net cash provided by (used in) discontinued operations	(34,487)	(145,528) 35,022
Net cash used in financing activities	(31,618)	
Effects of exchange rate changes on cash of		
continuing operations	4,450	10,120
Effects of exchange rate changes on cash of	,	,
discontinued operations	13,224	(17,667)
Changes in cash and cash equivalents	(7,363)	(10,966)
Cash and cash equivalents at beginning of period	44,748	66,426
Cash and cash equivalents at end of period	\$ 37,385 =======	
n and cash equivalents at September 30, 2004		
Continuing operations		\$ 24,932
Discontinued operations		30,528
Cash and cash equivalents at end of period		\$ 55,460
cash and cash equivarents at end of period		5 55,460

See accompanying notes.

Cash

COOPER TIRE & RUBBER COMPANY

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(Dollar amounts in thousands except per-share amounts)

1. The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by accounting principles generally accepted in the United States for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. There is a year-round demand for the Company's passenger and truck replacement tires, but passenger replacement tires are generally strongest during the third and fourth quarters of the year. Operating results for the three-month and nine-month periods ended September 30, 2004 are not necessarily indicative of the results that may be expected for the year ended December 31, 2004.

The balance sheet at December 31, 2003 has been derived from the audited financial statements at that date but does not include all of the information and footnotes required by accounting principles generally accepted in the United States for complete financial statements.

For further information, refer to the consolidated financial statements and footnotes thereto included in the Company's Annual Report on Form 10-K for the year ended December 31, 2003.

2. On September 17, 2004, the Company announced the signing of a definitive agreement to sell its automotive business, Cooper-Standard Automotive ("Cooper-Standard"). Also in September, the Tire Group announced its intent to cease its inner tube business. The Company is currently in discussions with potential buyers of the inner tube business or its assets. These operations are considered to be discontinued operations as defined under Statement of Financial Accounting Standard ("SFAS") No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets," and require specific accounting and reporting for this quarter which differs from the approach used to report the Company's results in prior quarters. The standard also requires restatement of comparable prior periods to conform to the required presentation.

The Company's consolidated financial statements reflect the accounting and disclosure requirements of SFAS No. 144, which mandate the segregation of operating results for the current year and comparable prior year periods and the current balance sheet related to the discontinued operations from those related to ongoing operations. Accordingly, the consolidated statements of income for the three and nine month periods ended September 30, 2003 and 2004 reflect this segregation as income from continuing operations and income from discontinued operations and the consolidated balance sheet at September 30, 2004 displays the segregation of the total assets of the operations to be sold as an aggregated current asset and the related total liabilities as an aggregated current liability.

In addition to the segregation of operating financial results, assets, and liabilities, Emerging Issues Task Force ("EITF") No. 87-24, "Allocation of Interest to Discontinued Operations," mandates the reallocation to continuing operations of general corporate overhead previously allocated to discontinued operations and permits the allocation of interest to discontinued operations in accordance with specific guidelines. Corporate overhead that previously would have been allocated to Cooper-Standard of \$3,223 for the quarter ended September 30, 2004 (\$3,149 for the quarter ended September 30, 2003) and \$9,670 for the nine-month period ended September 30, 2004 (\$9,449 for the nine-month period ended September 30, 2003) is charged against continuing operations in the Company's consolidated statements of income. The Company is using the permitted allocation method for interest expense on corporate debt, which is based on the ratio of net assets to be sold or discontinued to the sum of total net assets of the consolidated Company plus consolidated debt. Under this method, \$7,845 and \$23,656 of interest has been attributed to discontinued operations in the three-month and nine-month periods ended September 30, 2003 and 2004, respectively in addition to interest on debt held directly by Cooper-Standard. The actual amount of debt to be liquidated by application of the proceeds of the sale of Cooper-Standard has not yet been determined and may differ from the amount used in

the allocation process for preparing the consolidated financial statements. The Company has previously announced that proceeds from the sale of Cooper-Standard will be used for debt reduction, investment in the Company's tire operations, the repurchase of shares or a combination thereof and the amount of such uses will not be determined until an assessment is made of the Company's investment prospects and market conditions after the proceeds are available. In October of 2004 the Company commenced the repurchase of its common shares under a program approved by the Board of Directors in May of 2000.

Operating results for Cooper-Standard included in income from discontinued operations, net of income taxes on the Company's consolidated statements of income and are presented in the following table.

	Three months ended September 30					Nine months ended Septem				
		2003		2004		2003		2004		
Net sales	\$	392,159	\$	421,706	\$	1,209,518	\$	1,400,428		
Operating profit, including restructuring costs		14,649		11,060		70,471		101,475		
Interest expense Other - net		9,751 (666)		8,568 (882)		29,440 (3,437)				
Income from discontinued operations before income taxes		5,564		3,374		44,468		76,131		
Provision for income taxes		2,381		1,790		16,459		27,927		
<pre>Income from discontinued operations, net of income taxes</pre>	\$	3,183	\$	1,584	\$	28,009	\$	48,204		

The total assets and total liabilities of Cooper-Standard at December 31, 2003 and September 30, 2004 are detailed below. The Company's condensed consolidated balance sheet at September 30, 2004 presents these amounts as assets of discontinued operations and liabilities of discontinued operations.

	December 31, 2003			PTEMBER 30, 2004		
ASSETS						
Current assets	\$	440,287	\$	476,084		
Net property, plant, and equipment		506,225		481,196		
Goodwill		384,568		384,568		
Intangibles and other assets		55,749		66,481		
Total assets of discontinued operations	\$	1,386,829	\$	1,408,329		
	===	=======	===	=======		
LIABILITIES						
Current liabilities	\$	215,913	\$	238,895		
Long-term debt		8,056		6,426		
Postretirement benefits other than pensions		69,061		73,622		
Other long-term liabilities		51,445		49,387		
Deferred income taxes		(4,862)		841		
Total liabilities of discontinued operations	\$	339,613	\$	369,171		
	===	=======	===	=======		

Operating results for the inner tube business are included in income from discontinued operations, net of income taxes on the Company's consolidated statements of income and are presented in the following table.

	Three months ended September 30				Nine months ended September 30			
		2003		2004		2003	:	2004
Net sales	\$	6,676	\$	3,514	\$	17,389	\$	14,029
Operating profit (loss), including restructuring costs		(148)		(7,521)		831		(6,593)
Provision (benefit) for income taxes		(52)		(2,632)		291		(2,308)
<pre>Income (loss) from discontinued operations, net of income taxes</pre>	====:	(96)	====	(4,889)	====	540 ======	====:	(4,285)

The total assets of the inner tube facility at December 31, 2003 included in the Company's condensed consolidated balance sheets as assets of discontinued operations were \$10,299. These assets included land, buildings, machinery and equipment. During the third quarter of 2004, the buildings, machinery and equipment of the inner tube facility were written down to fair market value, as determined by the Company's expectations for proceeds upon its disposition, and liabilities were recorded for employee benefit, severance and environmental costs. At September 30, 2004, the total assets and total liabilities of the inner tube facility were \$5,729 and \$2,400, respectively.

3. In accordance with SFAS No. 131, "Disclosures about Segments of an Enterprise and Related Information," the Company has evaluated the determination of its reporting segments as a result of changes in its organizational reporting structure and the classification of Cooper-Standard as discontinued operations during the third quarter of 2004. The Company has determined it has two reportable segments for continuing operations - North American Tire Group and International Tire Group.

The following table details information on the Company's operating segments:

	Twelve months ended 12/31/03	NINE MONTHS ENDED 9/30/2004
Revenues		
North American Tire	\$ 1,682,593	\$ 1,383,665
International Tire		
Eliminations and other	209,631 (41,371)	
Consolidated	1,850,853	1,540,642
Segment profit		
North American Tire	77,370	61,788
International Tire	9,710	9,498
Unallocated corporate charges		
and eliminations	(22,061)	(15,422)
Operating profit	65,019	55,864
Other income - net	1,308	311
Interest expense	(28,884)	(20,959)
Income before income taxes	37,443	35,216
Depreciation and amortization expense		
North American Tire	99,027	73,268
International Tire	11,814	9,331
Corporate	1,840	1,335
Consolidated		83,934
Expenditures for long-lived assets		
North American Tire	86,257	90,420
International Tire	9,094	5,702
Corporate	730	167
Consolidated	96,081	96,289
Segment assets		
North American Tire	\$ 1,130,642	\$ 1,190,245
International Tire	181,802	197,338
Corporate and other	159,295	155,214
Consolidated	1,471,739	1,542,797

Geographic information for revenues, based on continent of origin, and long-lived assets follows:

	Twelve months ended 12/31/03	NINE MONTHS ENDED 9/30/2004
Revenues		
North America	\$ 1,650,743	\$ 1,351,179
Europe	200,110	189,463
Consolidated	1,850,853	1,540,642
Long-lived assets		
North America	\$ 612,032	\$ 620,184
Europe	79,330	76,268
Other	12	12
Consolidated	691,374	696,464

	Three months ended September 30			Nine months ended September 30				
		2003	DCI 3	2004		2003	C1 J	2004
Revenues from external customers:								
North American Tire	\$	477,819	\$	499,374	\$	1,211,165	\$	1,383,665
International Tire		54,759		65,985		158,917		194,403
Corporate/Eliminations		(13,407)		(13,913)		(33,967)		(37,426)
Net sales	\$	519,171	\$	551,446	\$	1,336,115	\$	1,540,642
	====	=======	===	=======	==	=======	==	=======
Segment profit:								
North American Tire	\$	31,041	\$	26,807	\$	53,136	\$	61,788
International Tire		2,650		2,802		10,122		9,498
Unallocated corporate charges and eliminations		(4,680)		(3,874)		(15,768)		(15,422)
Operating profit		29,011		25,735		47,490		55,864
Interest expense		7,001		6,580		22,419		20,959
Other - net		(318)		6		(976)		(311)
Income from continuing operations before income taxes	\$	22,328	\$	19,149	\$	26,047	\$	35,216
	====		===		==	========	==:	

4. On an annual basis, disclosure of comprehensive income is incorporated into the Statement of Shareholders' Equity. This statement is not presented on a quarterly basis. Comprehensive income includes net income and components of other comprehensive income, such as foreign currency translation adjustments, unrealized gains or losses on certain marketable securities and derivative instruments and minimum pension liability adjustments.

The Company's comprehensive income is as follows:

	Three	months end	ded Se	ptember 30	Nine	e months ende	ed Sep	tember 30
		2003		2004		2003		2004
Income from continuing operations Other comprehensive income (loss):	\$	14,670	\$	13,175	\$	17,113	\$	24,229
Currency translation adjustments Unrealized new gains (losses) on		1,352		(64)		(11,680)		17,346
derivative instruments		899		(1,983)		158		(93)
Comprehensive income from continuing operations	\$	16,921	\$	11,128	\$	5,591	\$	41,482

^{5.} The Company accounts for employee stock option plans in accordance with Accounting Principles Board Opinion ("APB") No. 25, "Accounting for Stock Issued to Employees." SFAS No. 123, "Accounting for Stock-Based Compensation," requires, if APB Opinion No. 25 is followed, disclosure of pro forma information regarding net income and earnings per share determined as if the Company accounted for its employee stock options under the fair value method. The fair value for these options was estimated at the date of grant using a Black-Scholes option pricing model with the following weighted-average assumptions:

	2003	2004
Risk-free interest rate	1.9%	2.4%
Dividend yield	2.8%	2.1%
Expected volatility of the		
Company's common stock	0.341	0.336
Expected life in years	6.6	6.7

The weighted-average fair value of options granted in 2003 and 2004 was \$3.74 and \$5.68, respectively. For purposes of pro forma disclosures, the estimated fair value of options is amortized to expense over the options' vesting period. The Company's reported and pro forma financial results are as follows:

		Three mo Septe		Nine months ended September 30				
	2	2003		2004		2003		2004
Income from continuing operations Deduct: Total stock-based employee compensation expense determined under the fair value based method for all awards, net of related tax	\$	14,670	\$	\$ 13,175		\$ 17,113		24,229
effects		(410)		(331)		(1,272)		(993)
Pro forma income from continuing operations	\$ ====	14,260	\$	12,844	\$ ===:	15,841 ======	\$ ====	23,236
Basic earnings per share from continuing operations: Reported Pro forma	\$	0.20 0.19	\$	0.18 0.17	\$	0.23 0.22	\$	0.33 0.31
Diluted earnings per share from continuing operations: Reported Pro forma	\$	0.20 0.19	\$	0.17 0.17	\$	0.23 0.21	\$	0.32 0.31

^{6.} The following table discloses the amount of net periodic benefit costs related to continuing operations for the nine months ended September 30, 2003 and 2004 for the Company's defined benefit plans and other postretirement benefits:

	Pension Benefits Three months ended September 30 Nine months ended September 3									
		2003		2004		2003		2004		
Components of net periodic benefit cost:										
Service cost	\$			5,178		13,987		15,578		
Interest cost		9,496		12,942		28,487		38,977		
Expected return on plan assets						(34,283)				
Amortization of transition obligation		(42)		(9)		(128)		(29)		
Amortization of prior service cost		608				, -		1,766		
Recognized actuarial loss		2,972		3,539		8,917		10,681		
Net periodic benefit cost		6,268	\$	7,656	\$	18,804	\$	23,067		
	===	=======	===	======	===	=======	===	=======		
	Other Postretirement Benefits									
	Three months ended September 30 Nine months ended September 30									
		2003	2004					2004		
Components of net periodic benefit cost:										
Service cost	\$	1,276	\$	1,313	\$	3,826	\$	3,941		
Interest cost		3,814		3,976		11,442		11,927		
Amortization of prior service cost		841		918		2,524	_	2,753		
Net periodic benefit cost	\$	5,931			\$	17,792	\$	18,621		

The Medicare Prescription Drug, Improvement and Modernization Act of 2003 (the "Act") was enacted in December 2003. The Act introduced a prescription drug benefit under Medicare Part D as well as a federal subsidy to sponsors of retiree health plans that provide a benefit that is at least actuarially equivalent to Medicare Part D. In May 2004, the Financial Accounting Standards Board ("FASB") issued FASB Staff Position ("FSP") 106-2, "Accounting and Disclosure Requirements Related to the Medicare Prescription Drug Improvement and Modernization Act of 2003." This FSP provided accounting and disclosure guidance for employers who sponsor postretirement health care plans that provide drug benefits. Preliminary Medicare reform regulations were issued in draft form late in July for comment before October 4, 2004. These regulations are complex and contain acknowledged open issues. The Company has prepared an estimate of their potential effect and recorded it during the third quarter of 2004, retroactive to January 1, 2004 as prescribed by FSP 106-2.

The Act reduced net periodic postretirement benefit cost by \$546 and \$1,637 for the three and nine months ended September 30, 2004, respectively, including service cost, interest cost and amortization of the actuarial gain. The total impact of the Act on the Company's actuarial liability under all U.S. plans was a reduction of \$15,300 and is being accounted for as an actuarial gain that will be amortized as a reduction of the Company's periodic expense and balance sheet liability over periods ranging from ten to twelve years.

7. The earnings per share previously reported for the first and second quarters and six months of 2004 are restated below for the segregation of operating results related to continuing operations from those related to discontinued operations, as well as for the impact of the Medicare Prescription Drug, Improvement and Modernization Act of 2003 which was recorded during the third quarter of 2004, retroactive to January 1, 2004.

		VIOUSLY CLOSED		TATED LOSURE
FIRST QUARTER 2004				
Basic earnings per share: Income from continuing operations Income from discontinued operations			·	0.03
Net income		0.32	\$	0.33
Diluted earnings per share: Income from continuing operations Income from discontinued operations				0.03
Net income		0.32	\$	0.32
SECOND QUARTER 2004				
Basic earnings per share: Income from continuing operations Income from discontinued operations			\$	0.12 0.34
Net income		0.45	•	0.46
Diluted earnings per share: Income from continuing operations Income from discontinued operations			\$	0.12 0.33
Net income	\$ =====	0.44	\$	0.45
SIX MONTHS 2004				
Basic earnings per share: Income from continuing operations Income from discontinued operations			\$	0.15 0.64
Net income		0.77		0.79
Diluted earnings per share: Income from continuing operations Income from discontinued operations			\$	0.15 0.62
Net income		0.76	\$ =====	0.77

8. During the third quarter of 2004, the North American Tire segment continued its consolidation of pre-cure retread operations and recorded \$1,100 of restructuring expense related to equipment disposal.

Also during the third quarter of 2004, the North American Tire segment announced a plan to cease production of radial medium truck tires at its Albany, GA facility by the end of the third quarter of 2005. These tires will be sourced from Asian manufacturers in the future. No employees will be affected by this initiative. The segment recorded an impairment charge of \$7,300 for equipment associated with radial medium truck tire production in order to write the equipment down to its fair market value, as determined by the Company's expectations for proceeds upon its disposition.

9. The Company provides a reserve for the estimated cost of product warranties at the time revenue is recognized. The reserve is based primarily on historical return rates and includes accruals for the Company's normal warranty programs and the enhanced warranty granted under terms of the settlement of class action litigation during 2001. During the quarter ended September 30, 2004, the Company received notice from the judge overseeing the Company's compliance with the terms of the settlement agreement that the court had reviewed the Company's compliance efforts and was satisfied with the Company's compliance. Such review included plant visits, reports by a third party of the Company's compliance with terms of the agreement, and the number of claims for enhanced warranty benefits. After considering this development and the number of enhanced warranty claims to date, the Company reevaluated the reserve required for the enhanced warranty claims and reduced it by \$11,867 during the quarter to reflect costs expected during the remaining term of the enhanced warranty. The following table summarizes the activity in the Company's product warranty liabilities since December 31, 2003:

Reserve at December 31, 2003	\$ 22,642
Additions	3,558
Reduction to enhanced warranty reserve	(11,867)
Payments	 (4,345)
Reserve at September 30, 2004	\$ 9,988

- 10. The Company has provided a guarantee of a portion of the bank loans made to its automotive business joint venture with Nishikawa Rubber Company. In 2003, the joint venture entered into an additional bank loan, maturing in 2008, with the joint venture partners each guaranteeing an equal portion of the amount borrowed. In accordance with FASB Interpretation No. 45, "Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness of Others," guarantees meeting the characteristics described in the Interpretation are required to be recorded at fair value. As of September 30, 2004, the Company has recorded a \$36 liability related to the guarantee of this debt with a corresponding increase to the carrying value of its investment in the joint venture. The Company's maximum exposure under the two guarantee arrangements at September 30, 2004 was approximately \$4,500.
- 11. The Company is a defendant in various judicial proceedings arising in the ordinary course of business. A significant portion of these proceedings are products liability cases in which individuals involved in vehicle accidents seek damages resulting from allegedly defective tires manufactured by the Company. Litigation of this type has increased significantly throughout the tire industry following the Firestone tire recall announced in 2000.

The Company accrues costs for products liability at the time a loss is probable and the amount of loss can be estimated. During the quarter ended June 30, 2004, the Company refined the specific criteria against which to evaluate claims. The Company believes the probability of loss can be established and the amount of loss can be estimated only after certain minimum information is available, including verification that Company-produced products were involved in the incident giving rise to the claim, the condition of the product purported to be involved in the claim, the nature of the incident giving rise to the claim, and the extent of the purported injury or damages. In cases where such information is known, each products liability claim is evaluated based on its specific facts and circumstances. A judgment is then made, taking into account the views of counsel and other relevant factors, to determine the requirement for establishment or revision of an accrual for any potential liability. In most cases, the liability cannot be determined with precision until the claim is resolved. Pursuant to applicable accounting rules, the Company accrues the minimum liability for each known claim when the estimated outcome is a range of possible loss and no one amount within that range is more likely than another. No specific accrual is made for individual unasserted claims or for asserted claims where the minimum information needed to evaluate the probability of a liability is not yet known. However, an accrual for such claims based, in part, on management's expectations for future

litigation activity is maintained. The total cost of resolution of such claims, or increase in reserves resulting from greater knowledge of specific facts and circumstances related to such claims, could have a greater impact on the consolidated results of operations and financial position of the Company in future periods and, in some periods, could be material.

The products liability expense reported by the Company includes amortization of insurance premium costs, adjustments to settlement reserves, and legal costs incurred in defending claims against the Company offset by recoveries of legal fees. The Company is entitled to reimbursement under certain insurance contracts in place for periods ending prior to April 1, 2003 of legal fees expensed in prior periods based on events occurring in the those periods. During the three month periods ended September 30, 2003 and 2004 products liability expense totaled \$12,800 and \$16,800, respectively. For the nine month periods ended September 30, 2003 and 2004 products liability expense totaled \$28,200 and \$44,500, respectively. The nine-month period of 2003 included only six months under the new program which was initiated on April 1, 2003. The new program includes occurrence-based insurance coverage with an increased per claim retention limit, increased policy limits, and the establishment of a captive insurance company. Premium costs for insurance coverage in excess of the self-insured amounts for the first policy year were \$10,300 higher than under the previous program, the per claim retention limit increased \$13,300 and the aggregate retention limit was eliminated, while excess liability coverage increased by \$35,000. The program was renewed effective April 1, 2004 with a two percent increase in insurance premiums cost. Recoveries of legal fees were \$1,400 and \$400, respectively, in the three months ended September 30, 2003 and 2004. Policies applicable to claims occurring on April 1, 2003 and thereafter do not provide for recovery of legal fees.

12. The Company's effective income tax rate for continuing operations for both the third quarter and first nine months of 2003 and 2004 was 34.3 percent and 31.2 percent, respectively. The rates reflect the impact of tax credits, global tax planning, the mix of earnings by entity across foreign and domestic jurisdictions and, for the 2004 periods, the effect of the benefit accruing to the Company under the Medicare Prescription Drug, Improvement and Modernization Act of 2003

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

This Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A") presents information related to the consolidated results of operations of the Company, including the impact of restructuring costs on the Company's results, a discussion of the past results and future outlook of each of the Company's segments and its discontinued operations, and information concerning both the liquidity and capital resources of the Company. An important qualification regarding the "forward-looking statements" made in this discussion is then presented.

On September 17, 2004 the Company announced the signing of a definitive agreement to sell its automotive business, Cooper-Standard Automotive. Also in September, the Tire Group announced its intent to cease its inner tube business and is currently in discussions with potential buyers for this business or its assets. These operations are considered to be discontinued operations as defined under Statement of Financial Accounting Standard ("SFAS") No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets," and require specific accounting and reporting for this quarter which differs from the approach used to report the Company's results in prior quarters. It also requires restatement of comparable prior periods to conform to the required presentation.

The Company's consolidated financial statements reflect the accounting and disclosure requirements of SFAS No. 144 which mandates the segregation of operating results for the current year and comparable prior year periods and the current balance sheet related to the operations to be sold from those related to ongoing operations. Accordingly, the consolidated statements of income for the three and nine-month periods ended September 30, 2003 and 2004 reflect this segregation as income from continuing operations and income from discontinued operations and the consolidated balance sheet at September 30, 2004 displays the segregation of the

total assets of the operations to be sold as an aggregated current asset and the related total liabilities as an aggregated current liability.

CONSOLIDATED RESULTS OF OPERATIONS

(Dollar amounts in millions except per share amounts)

	Three months ended September 30					Nine months ended September 30					
		003	% Change	2	004	2	 2003 	% Change	:	2004	
Revenues:											
North American Tire International Tire Eliminations	\$	477.8 54.8 (13.4)	4.5 20.4 4.5	\$	499.4 66.0 (14.0)		1,211.2 158.9 (34.0)	14.2 22.3 10.3		1,383.7 194.4 (37.5)	
Net sales	\$ ===	519.2	6.2	\$	551.4		1,336.1	15.3		1,540.6	
Segment profit:											
North American Tire International Tire Unallocated corporate charges	\$	31.0 2.7	-13.5 3.7	\$	26.8 2.8	\$	53.1 10.1	16.4 -5.9	\$	61.8 9.5	
and eliminations		(4.7)	-17.0		(3.9)		(15.7)	-1.9		(15.4)	
Operating profit Interest expense Other income - net		29.0 7.0 (0.3)	-11.4 -5.7 -100.0		25.7 6.6 		47.5 22.5 (1.0)	17.7 -6.7 -70.0		55.9 21.0 (0.3)	
Income from continuing operations before income taxes		22.3	-14.3		19.1		26.0	35.4		35.2	
Provision for income taxes		7.6	-22.4		5.9		8.9	23.6		11.0	
Income from continuing operations		14.7	-10.2		13.2		17.1	41.5		24.2	
<pre>Income (loss) from discontinued operations, net of income taxes</pre>		3.1	-206.5		(3.3)		28.6	53.5		43.9	
Net income	\$ ===	17.8	-44.4		9.9	\$	45.7	49.0	\$	68.1	
Basic earnings per share: Income from continuing operations Income (loss) from discontinued operations	\$	0.20		\$	0.18	\$	0.23		\$	0.33	
operacions					(0.03)					0.59	
Net income	\$ ===	0.24	-45.8	\$ ===	0.13	\$ ===	0.62	48.4	\$ ==:	0.92	
Diluted earnings per share: Income from continuing operations Income (loss) from discontinued	\$	0.20		\$	0.17	\$	0.23		\$	0.32	
operations		0.04			(0.04)		0.39			0.58	
Net income	\$	0.24	-45.8	\$	0.13	\$	0.62	45.2	\$	0.90	

Consolidated net sales for the three-month period ended September 30, 2004 were \$32 million higher than for the comparable period one year ago. Net sales for the North American Tire segment increased \$22 million. The International Tire segment increased net sales by \$11 million, with favorable foreign currency contributing \$7 million to this improvement. The impacts of more favorable mix and the price increases achieved since October of 2003 more than offset the impact of lower sales volumes. Operating profit in the third quarter of 2004 decreased by \$3 million from the operating profit reported for third quarter of 2003. This reduction in operating profit was the result of higher raw material costs, inefficiencies in plant operations, lower unit volume, restructuring charges, increased marketing programs costs, increased product liability costs and expanded advertising programs which were partially offset by the impacts of improvements in pricing and mix, Lean savings and the reduction to the class action liability reserve.

Consolidated net sales for the nine-month period ended September 30, 2004 were \$205 million higher than for the comparable period one year ago. Net sales for the North American Tire segment increased \$173 million and net sales for the International Tire segment increased by \$36 million, with favorable foreign currency contributing \$22 million to the latter increase. The increase in net sales was a result of increased sales volumes, price increases achieved since October of 2003 and more favorable mix. Operating profit in the first nine months of 2004 increased by \$8 million from the operating profit reported for the first nine-months of 2003. The improvement in operating profit resulted from the impacts of improved pricing, Lean savings, improved mix, higher sales volumes and the reduction to the class action liability reserve which exceeded the impacts of higher raw material costs, increased marketing costs, expanded advertising programs, product liability costs, inefficiencies in plant operations and restructuring charges.

The Company experienced significant increases in the costs of certain of its principal raw materials during the third quarter and the first nine months of 2004 compared with the levels experienced during the comparable periods of 2003. The principal raw materials for the North American Tire and International Tire segments include synthetic rubber, carbon black, natural rubber, chemicals and reinforcement components. The principal raw materials for the former Automotive segment, now classified as discontinued operations, include fabricated metal-based components, synthetic rubber, carbon black and natural rubber. The Company manages the procurement of its raw materials to assure supply and to obtain the most favorable pricing. For natural rubber, procurement is managed by buying forward of production requirements and by buying in the spot market. For metal-based components, procurement is managed through long-term supply contracts. For other principal materials, procurement arrangements include multi-year supply agreements that may contain formula-based pricing based on commodity indices. These arrangements provide quantities needed to satisfy normal manufacturing demands.

The increases in the costs of natural rubber and crude oil were the most significant drivers of higher raw material costs during the third quarter and the first nine months of 2004. The cost of natural rubber increased approximately 33 percent and 37 percent, respectively, from the quarter and nine-month levels of the comparable periods of 2003. The price of crude oil, the primary raw material used in the production of the synthetic rubber, carbon black and many chemicals used by the Company, has risen to historically high levels during 2004 The increasing price of crude oil and the growing global demand for its derivative products is contributing to the cost increases being experienced for raw materials used by the Company and adding to concerns regarding their availability.

The price of components fabricated from steel, including automotive fabricated metal-based components and tire cord and bead components, are being adversely impacted by scarcity of supply. Since March 2004, the Company has paid surcharges for its steel components in excess of the pricing contained in its steel-component supplier contracts to ensure supply. Through September 30, no interruption of the supply of components fabricated from steel has been experienced. Surcharges continued to be implemented during the third quarter and are anticipated to remain at current levels at least through the fourth quarter of 2004.

Selling, general, and administrative expenses were \$39 million in the third quarter of 2004 (7.1 percent of net sales) compared to \$36 million (7.0 percent of net sales) in the same period in 2003. Increased costs associated with an expanded advertising program and the timing of those costs were responsible for the increase. For the

first nine months of 2004, selling, general, and administrative costs were \$126 million (8.2 percent of net sales) compared to \$107 million (8.0 percent of net sales) in the comparable period of 2003 with the same factors contributing to the increase.

Interest expense decreased less than \$1 million in the third quarter of 2004 from the third quarter of 2003 and decreased slightly more than \$1 million during the first nine months of 2004 compared to the first nine months of 2003 reflecting lower interest rates and lower debt levels.

Other-net decreased during the third quarter as a result of foreign currency losses being recorded in 2004 compared to gains recorded in 2003 and higher interest income in 2004. For the nine months ended September 30, 2004, other-net decreased less than \$1 million from the first nine months of 2003 due to the same factors impacting the quarter.

The Company's effective income tax rate for continuing operations for the third quarter and first nine months of 2003 and 2004 was 34.3 percent and 31.2 percent, respectively. The rates reflect the impact of tax credits, global tax planning, the mix of earnings by entity across foreign and domestic jurisdictions and, for the 2004 periods, the effect of the benefit accruing to the Company under the Medicare Prescription Drug, Improvement and Modernization Act of 2003.

RESTRUCTURING

During the third quarter of 2004, the North American Tire segment continued its consolidation of pre-cure retread operations and recorded \$1.1 million of restructuring expense related to equipment disposal.

Also during the third quarter of 2004, the North American Tire segment announced a plan to cease production of radial medium truck tires at its Albany, GA facility by the end of the third quarter of 2005. These tires will be sourced from Asian manufacturers in the future. No employees will be affected by this initiative. The segment recorded an impairment charge of \$7.3 million for equipment associated with radial medium truck tire production, writing it down to its fair market value, as determined by the Company's expectations for proceeds upon its disposition.

North American Tire Segment

(Dollar amounts in millions)	THREE MON'	THS ENDED SEPTE	MBER 30	NINE MONTHS ENDED SEPTEMBER 30				
	2003	CHANGE %	2004	2003	CHANGE %	2004		
Sales	\$ 477.8	4.5%	\$ 499.4	\$ 1,211.2	14.2%	\$ 1,383.7		
Operating profit	\$ 31.0	-13.5%	\$ 26.8	\$ 53.1	16.4%	\$ 61.8		
United States unit sales changes: Passenger tires								
Company		-8.2%			0.9%			
RMA members		-4.5%			1.9%			
Light truck tires								
Company		-5.0%			12.0%			
RMA members		-7.3%			2.7%			
Total light vehicle tires								
Company		-7.7%			2.8%			
RMA members		-4.9%			2.0%			
Total segment unit sales changes		-6.9%			4.0%			

OVERVIEW

Shipments of passenger car and light truck tire replacement units in the United States market by members of the Rubber Manufacturers Association ("RMA"), a group comprised of the eleven largest tire companies in the world including the segment, and which accounted for over 90 percent of the total United States tire market in recent years, decreased approximately 4.9 percent in the third quarter of 2004 from shipment levels in the third quarter of 2003. Passenger tire unit shipments, which account for over 80 percent of the combined passenger and light truck tire markets, decreased by 4.5 percent while light truck tire unit shipments decreased by approximately 7.3 percent.

For the first nine months of 2004, RMA shipments of passenger and light truck tire replacement units increased 2.0 percent from the first nine months of 2003. Passenger tire shipments increased 1.9 percent while light truck tire shipments increased 2.7 percent. The replacement tire market in the United States was weak throughout the first five months of 2003 but started to strengthen in June, and remained strong for the rest of the year.

SALES

Sales of the North American Tire segment increased \$22 million in the third quarter of 2004 from 2003 levels. Tire unit sales were down 6.9 percent from the 2003 third quarter period. The segment's unit sales of passenger and light truck tires decreased by 8.2 percent and 5.0 percent, respectively, in the third quarter of 2004 compared to the third quarter of 2003.

Increased unit sales volumes were achieved in several product categories in which new product offerings are being introduced -- high performance and sport utility vehicle tires. However, declines in the broadline economy and light truck tire product categories during the quarter more than offset those improved volumes. The segment recorded increased sales in the distributor channel and increased sales of its proprietary, brand name tires. Declines in sales to mass merchandiser customers partially offset those increases. In addition to the impact of improvements to customer and product mix achieved during the third quarter, sales for the segment benefited from the price increases implemented in October of 2003 and in February and June of 2004.

Sales increased \$173 million during the first nine months of 2004 from levels in 2003 due to higher sales volumes, the price increases achieved since October 2003, and improvements in both customer and product mix. Increases in sales in the distributor and retail channels and increased sales of its proprietary, brand name tires were partially offset by a decline in sales to mass merchandiser customers.

The segment's tire unit sales were up 4.0 percent from the 2003 nine-month period. Unit sales of passenger and light truck tires increased by .9 percent and 12.0 percent, respectively, during the first nine months of 2004 compared to the first nine months of 2003.

OPERATING PROFIT

Segment operating profit in the third quarter of 2004 decreased \$4 million from the third quarter of 2003. The impact of price increases (\$34 million), improved customer and product mix (\$8 million), savings generated by Lean initiatives (\$8 million) and the reversal of a portion of the class action reserve (\$7 million) increased operating profit during the quarter. Higher raw material costs (\$27 million), lower unit sales volumes (\$9 million), increased marketing program costs (\$7 million), restructuring charges (\$8 million), increases in products liability costs (\$4 million) and increases in other operating costs lowered operating profit.

For the first nine months of 2004, operating profit increased \$9 million over the comparable period of 2003. The impact of price increases (\$64 million), savings generated from Lean initiatives (\$29 million), improved customer and product mix (\$21 million), higher sales (\$17 million) and the reversal of a portion of the class action reserve (\$7 million) more than offset higher raw material costs (\$60 million), increased marketing costs associated with the higher sales revenue (\$19 million), increased product liability costs (\$16 million), increases in other operating costs (\$14 million), increased costs associated with expanded advertising programs (\$12 million) and restructuring charges (\$8 million).

During the quarter ended September 30, 2004, the Company received notice from the judge overseeing the Company's compliance with the terms of the 2001 class action settlement agreement that the court had reviewed the Company's compliance efforts and was satisfied with the Company's compliance. Such review included plant visits, reports by a third party of the Company's compliance with terms of the agreement, and the number of claims for enhanced warranty benefits. After considering this development, the number of enhanced warranty claims to date, and the adequacy of other settlement-related serves, the Company reevaluated the class action settlement reserve and reduced it by \$11.3 million during the quarter to reflect costs expected during the remaining term of the enhanced warranty. During the quarter ended September 30, 2003, the Company had reviewed the adequacy of the enhanced warranty liability related to the class action settlement and reduced it by \$3.9 million. The reduction was attributed to a reduction in the eligible population of tires subject to the enhanced warranty due to the passage of time and to lower than expected claims.

OUTLOOK

The Company is optimistic that the net sales improvement experienced by the North American Tire segment during the third quarter will carry over into the fourth quarter of the year. Future comparisons into the first half of 2005 will be difficult given the strong industry volumes achieved in the first half of 2004, but the segment's continued market share gains by its house brands and increased volumes for its high performance products are expected to continue in 2005.

The Company believes the segment's operating profit levels will improve due not only to the implementation of recently announced price increases, but also due to the favorable impact of improved product and customer mix, improvements in operating efficiencies and manufacturing capacity, and the cost reductions generated through Lean initiatives. Raw material prices continue to be very difficult to predict accurately due to the volatility of prices for crude oil, energy and transportation. The market scarcity of steel for the segment's tire cord and bead components and crude oil-based raw materials is a concern and is being actively managed, although no interruption of supply has been experienced. The Company believes raw material costs will be approximately three percent higher on average in the fourth quarter of 2004 than in the third quarter of 2004.

The segment is near completion of expansions at two of its domestic tire manufacturing facilities and has expansions underway at two other domestic facilities. The segment currently has manufacturing supply agreements with two Asian manufacturers to provide passenger tires from China for distribution in the North American market. In addition, the segment is implementing its plans to transfer its radial medium truck tire production to China through contract manufacturing arrangements which will make domestic production capacity available for production of certain light truck tires and other higher-margin products. The timing of the transfer of radial medium truck tires has lagged expectations and, as a result, total deliveries for the year will be less than previously anticipated. The domestic plant expansions and inventory management initiatives will compensate for the passenger tire units originally expected to be sourced from Asia. The domestic plant expansions, Asian sourcing arrangements and inventory management initiatives are important to the segment's ability to profitably provide tire products to its customers.

In the wake of the Firestone recall announced in 2000, the tire industry and the Company have experienced a significantly higher level of product liability litigation. Effective April 1, 2003, the Company established a new excess liability insurance program. The new program covers the Company's product liability claims occurring on or after April 1, 2003 and is occurrence-based insurance coverage which includes an increased per claim retention limit, increased policy limits, and the establishment of a captive insurance company. Premium costs for insurance coverage in excess of the self-insured amounts for the first policy year were \$10.3 million higher than under the previous program, the per claim retention limit increased \$13.3 million and the aggregate retention limit was eliminated, while excess liability coverage increased by \$35 million. The program was renewed effective April 1, 2004 with a two percent increase in insurance premiums cost. It is possible product liability costs may fluctuate from period to period in the future and that such costs could have a greater impact on the consolidated results of operations and financial position of the Company in future periods than in the past and, in some periods, could be material. The Company is aggressively managing its product liability costs.

INTERNATIONAL TIRE SEGMENT

(Dollar amounts in millions)

	THREE MONTHS ENDED SEPTEMBER 30					NINE MONTHS ENDED SEPTEMBER 30					
	 2003 CHANGE % 2004		2003		CHANGE %		2004				
Sales	\$ 54.8	20.4%	\$	66.0	\$	158.9	22.3%	\$	194.4		
Operating profit	\$ 2.7	3.7%	\$	2.8	\$	10.1	-5.9%	\$	9.5		

OVERVIEW

The International Tire segment currently manufactures and markets passenger car, light truck and motorcycle tires for the replacement market, as well as racing tires and materials for the tire retread industry, in Europe and the United Kingdom.

Shipments of passenger car and light truck tires in the segment's markets, based on data published by the industry and other sources, increased approximately six percent in the third quarter of 2004 from the comparable period in 2003. For the first nine months of 2004, market shipments of passenger and light truck tires increased four percent from the first nine months of 2003.

SALES

Sales of the Company's International Tire operations increased \$11 million, or 20.4 percent, in the third quarter of 2004 from the comparable period of 2003. Approximately \$7 million of the increase was attributable to the foreign exchange impact of a weakened United States dollar in relation to the British pound. The remaining sales growth resulted from increased sales volumes in established distribution channels and sales growth of new product offerings in the performance lines of tires.

Sales increased \$35 million, or 22.3 percent, in the first nine months of 2004 from the comparable period of 2003. Approximately \$22 million of the increase was attributable to the foreign exchange impact of a weakened United States dollar in relation to the British pound. The remaining sales growth resulted from increased sales volumes in established distribution channels in the United Kingdom and the rest of Europe.

OPERATING PROFIT

Operating profit for the International Tire segment for the third quarter of 2004 was equal to the operating profit generated in the third quarter of 2003. The impact of price increases (\$2 million), savings generated from Lean initiatives (\$2 million) and the increased sales volume (\$1 million) were offset by increases in other operating costs (\$4 million) and increased raw material prices (\$1 million).

For the nine months ended September 30, 2004, operating profit for the segment decreased less than \$1 million due to the same factors cited for the third quarter.

OUTLOOK

The Company believes that the net sales improvements experienced in the third quarter by the International Tire segment will continue into the fourth quarter, and that demand is expected to remain strong for its products into the first half of 2005.

The Company also believes that the segment's operating profit levels will improve over the fourth quarter of 2003 due to price increases implemented earlier in the year, improved product and customer mix, improvements in operating efficiencies and cost reductions generated through its Lean initiatives. Raw material costs are expected to be three percent higher on average in the fourth quarter of 2004 than in the third quarter of 2004, and energy costs remain high.

The segment continues to pursue opportunities for expansion in Asia through joint ventures and other forms of alliance. The segment currently has a manufacturing supply agreement with an Asian manufacturer to provide passenger tires for distribution in its markets.

DISCONTINUED OPERATIONS

On September 17, 2004 the Company announced the signing of a definitive agreement to sell its automotive business, Cooper-Standard Automotive. Also in September, the Tire Group announced its intent to cease its inner tube business and is currently in discussions with potential buyers for this business. These operations are considered to be discontinued operations as defined under Statement of Financial Accounting Standard ("SFAS") No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets," and require specific accounting and reporting for this quarter which differs from the approach used to report the Company's results in prior quarters. It also requires restatement of comparable prior periods to conform to the required presentation.

COOPER-STANDARD AUTOMOTIVE

(Dollar amounts in millions)		THREE MON	THS ENDED SE		NINE MONTHS ENDED SEPTEMBER 30						
	CHANGE					CHANGE					
	2003		8	2004		2003		%	2004		
Sales	\$	392.2	7.5%	\$	421.7	\$	1,209.5	15.8%	\$	1,400.4	
Operating profit	\$	14.6	-24.0%	\$	11.1	\$	70.5	44.0%	\$	101.5	
Annualized vehicle build (millions) North America Europe							15.9 19.5	0.0% 3.1%		15.9 20.1	
Sales to U.Sbased OEMs		77%			75%		79%			78%	

OVERVIEW

Cooper-Standard Automotive serves automotive original equipment manufacturers ("OEMs") throughout the world. Light vehicle production in North America fell by more than one percent but production in Europe increased by almost three percent. For the nine months ended September 30, 2004, light vehicle production in North America was slightly lower than during the nine month period of 2003 while European production increased by more than four percent.

Market share pressure on the U.S.-based OEMs, as evidenced by the pricing environment of zero percent financing and record high rebates, increased pension and other retirement-related costs, and the impact of global overcapacity have reduced the overall profitability of the industry, and have resulted in continued pressure on suppliers for price concessions.

In spite of these industry conditions, Cooper-Standard Automotive has improved its profitability in the first nine months of 2004 by emphasizing continuous improvement, Lean manufacturing and cost reduction initiatives, execution of restructuring initiatives and implementing the sourcing of components and product from low-cost Asian manufacturers.

In May 2003, the Company increased its ownership position in Jin Young Standard of South Korea from 49 percent to 90 percent and changed the name of the operations to Cooper-Standard Automotive Korea, Inc. Cooper-Standard Automotive is working closely with its Korean subsidiary to expand its business with the Korean OEMs, who have increased their share of the global automobile market in recent years.

In July 2004, the segment entered into a joint-venture agreement with China-based Saiyang Sealing Products to manufacture and sell automotive sealing products in China under the name Cooper Saiyang Wuhu Automotive. The venture has secured business with two Chinese OEMs.

Cooper-Standard recently established two small manufacturing facilities in China for the production of NVH control systems and fluid handling systems products. The NVH control systems location has begun operations for the manufacture and sales of NVH control systems in Kunshan, China and will soon break ground for an expanded manufacturing, research and development facility in the same city.

SALES

Sales for Cooper-Standard Automotive ("Cooper-Standard") increased \$30 million in the third quarter of 2004 compared to the third quarter of 2003. Sales increases in North America of \$15 million for the quarter were the result of net new business and the impact of favorable foreign currency translation offset by lower production volumes and price concessions. In the segment's international operations, a sales increase of \$15 million is attributable to the favorable impact of foreign currency translation, net new business and higher production volumes.

Cooper-Standard's sales for the first nine months of 2004 increased \$191 million over the comparable 2003 period. Sales increases in North America of \$116 million for the nine-month period were the result of net new business, higher production levels and the impact of favorable foreign currency translation offset by price concessions. The sales increase in the international operations was \$75 million and is attributable to the favorable impact of foreign currency translation, higher production levels and the inclusion of the sales of Cooper-Standard Automotive Korea.

OPERATING PROFIT

Operating profit in the third quarter of 2004 for Cooper-Standard was more than \$3 million lower than the operating profit reported in the third quarter of 2003. Operating margins were 1.1 percentage points lower than in 2003. The positive impacts of net new business (\$26 million) and Lean savings (\$17 million) were more than offset by higher raw material costs (\$13 million), lower volumes (\$13 million), increased price concessions (\$8 million), higher restructuring costs (\$3 million), higher selling, general and administrative and benefit costs (\$3 million), and increases in other operating costs.

Cooper-Standard's operations in North America were less profitable in the third quarter of 2004 than in the comparable 2003 period due to the impact of raw material costs and price concessions which were only partially offset by net new business and Lean savings. Operations outside of North America improved due to the accomplishment of Lean savings and improved pricing.

Operating profit in the first nine months of 2004 for Cooper-Standard Automotive was \$31 million higher than the operating profit reported in the comparable period of 2003. Operating margins were 1.4 percentage points higher than in 2003. The positive impacts of net new business (\$59 million), Lean savings (\$54 million), favorable foreign currency translation (\$7 million) and higher production levels (\$2 million) were offset by higher raw material costs (\$28 million), increased price concessions (\$28 million), increased manufacturing costs not related to volume (\$13 million), higher restructuring costs (\$10 million), higher selling, general and administrative and benefit costs (\$10 million) and other cost increases.

OUTLOOK

On September 17, 2004, the Company announced the signing of a definitive agreement to sell its automotive business, Cooper-Standard Automotive. The segment is continuing its operations as usual by seeking new business, developing new products, implementing its Asian strategy, and filling customer orders as needed to maintain the expected level of customer service.

On September 27, 2004, the United Steel Workers of America ("USWA") filed a complaint asserting they have the right to require the buyer to negotiate new labor agreements affecting four Cooper-Standard facilities before a sale takes place. On November 2, 2004, the U. S. District Court for the Northern District of Indiana issued a preliminary injunction enjoining the sale of those four facilities and ordering expedited arbitration of the USWA grievance to determine whether the proposed sale violates successorship language of the collective bargaining agreements. On November 5, 2004, the Company appealed this decision to the U. S. Seventh Circuit Court of Appeals. The Company believes this ruling will be reversed on appeal and continues to pursue the closing of the transaction according to the original schedule.

INNER TUBE BUSINESS

On September 8, 2004 the Company announced its intent to cease its inner tube business and is currently in discussions with potential buyers for this business or its assets. During the third quarter of 2004, the Company recorded restructuring charges of \$7.1 million related to this decision. The charges included equipment write-downs of \$4.2 million, severance costs of \$1.6 million, and employee benefit costs of \$1.0 million and environmental cost of \$.3 million.

Sales for the Company's inner tube business for the third quarter and nine months of 2004 decreased \$3 million from the comparable 2003 periods. Without the restructuring charge, the operating loss in the third quarter was \$.4 million compared to \$.1 million in 2003. For the nine months ended September 30, 2004, the operating profit before the restructuring charge was \$.5 million, lower than the \$.8 million recorded in the comparable period of 2003.

LIQUIDITY AND CAPITAL RESOURCES

Generation and uses of cash -- Net cash provided by operating activities of continuing operations was \$151 million in the first nine months of 2004, an increase of \$66 million from the \$85 million generated in the first nine months of 2003. Net income after adjustments for non-cash items increased \$10 million reflecting higher net income (\$7 million), restructuring asset write-downs (\$9 million) and the reduction of the class action settlement reserve (\$7 million). Changes in operating assets and liabilities provided cash of \$44 million in 2004 versus the use of \$12 million of cash in 2003. These changes result primarily from the timing of payrolls and higher accruals for benefits, product liability reserves and enhanced advertising programs. Accounts receivable balances in 2003 increased partly due to changes in payment patterns of certain North American Tire segment customers. No additional impact of the customer changes occurred in 2004 and accounts receivable balances increased only modestly in 2004 in spite of increases in net sales. Inventory balances have increased in 2004 in the North American Tire finished goods component and in the Company's raw materials area.

Net cash used in investing activities during the first nine months of 2004 reflects capital expenditures of \$96 million, up \$29 million from the comparable period in 2003. During the first nine months of 2003, the Company acquired Max-Trac Tire Co., Inc., known as Mickey Thompson Performance Tires & Wheels, for \$13 million.

The Company's financing activities during the first nine months of 2004 reflect the early payment of \$90 million under a variable rate facility that was due in 2006. Borrowings under the Company's short-term credit facilities were reduced \$45 million during the nine months of 2004 while in the nine months of 2003 the Company borrowed \$26 million. The issuance of common shares from the exercise of stock options generated \$15 million during the first nine months of 2004. Dividends paid on the Company's common shares in the first nine months of 2004 and 2003 were \$23 million.

Available credit facilities - On June 30, 2004, the Company extended its revolving credit facility with a consortium of eleven banks ("the Agreement") by an additional one year that provides up to \$175 million in credit facilities until August 31, 2008 and an additional \$175 million in credit facilities until June 29, 2005. The Company has the option to convert any outstanding loans under the short-term commitment into a one-year term loan. The Company generally renegotiates the short-term portion of its credit facility each year. The credit facilities support the issuance of commercial paper.

Also on June 30, 2004, the credit facility was restated and amended. Pursuant to the amendment, the ratio of income before fixed charges and income taxes to fixed charges (the "fixed charge coverage ratio") was eliminated and replaced by an interest coverage ratio. This ratio (consolidated earnings before interest, taxes, depreciation and amortization divided by consolidated net interest expense) is required to be maintained at a minimum of 3.0 times by the Company. The amendment also changed the computation of the ratio of total debt to total capitalization to consolidated net indebtedness to consolidated capitalization. Consolidated net indebtedness measured in accordance with generally accepted accounting principles in the United States reduced by cash and eligible short term investments in excess of \$30 million. The Company is required to maintain this ratio below 55 percent. As of September 30, 2004 the Company was in compliance with the financial covenants contained in its credit agreements. At that date, the percentage of consolidated net indebtedness to consolidated capitalization was 41.8 percent and the interest coverage ratio was 6.6 times. The Company anticipates that it will remain in compliance with these covenants in 2004, based upon its business forecast for the year.

The Company's credit agreement also contains a covenant which prevents the disposition of a substantial portion of its assets. The pending disposition of Cooper-Standard Automotive will require a waiver of this covenant by its consortium of lenders to continue the credit agreements. The Company is in discussions with its agent bank and expects to obtain a waiver of this covenant by its lenders prior to the closing of the sale transaction.

There were no changes in the Company's long and short-term debt ratings during the quarter. However, Standard & Poor's placed its credit ratings for the Company on "credit watch with negative implications" in March 2004 following the announcement of the exploration of the possibility of a sale of Cooper-Standard Automotive. If a downgrade in its credit ratings were to occur, the Company believes it would continue to have access to the credit markets, although at higher borrowing costs than is presently the case.

Available cash and contractual commitments -- The Company anticipates cash flows from operations in 2004 will meet its projected capital expenditures and dividends goals. The Company will begin to make investments in China during the fourth quarter of 2004 and in October of 2004 commenced the repurchase of its common shares under a program approved by the Board of Directors in May of 2000 under which it has authority as of September 30, 2004 to acquire an additional 8.7 million shares. The Company may consider a self-tender for its shares in the future. At September 30, 2004 the Company had cash of \$25 million and could borrow, under its credit agreement with its bank group and other bank lines, up to an additional \$350 million without violating the financial covenants contained in its credit agreements.

CONTINGENCIES

The Company is a defendant in various judicial proceedings arising in the ordinary course of business. A significant portion of these proceedings are product liability cases, in which individuals involved in vehicle accidents allege damages resulting from allegedly defective tires manufactured by the Company. Litigation of this type has increased significantly throughout the tire industry following the Firestone tire recall announced in 2000. After reviewing all of such proceedings known at the time of this filing, and taking into account all relevant factors concerning them, the Company does not believe that any liabilities resulting from these proceedings, in excess of amounts currently reserved, are reasonably likely to have a material adverse effect on its liquidity, financial condition or results of operations. As a result of the changes in the Company's insurance program effective April 1, 2003, product liability costs could have a greater impact on the consolidated results of operations and financial position of the Company in future periods than in the past and, in some periods, could be material. The Company is aggressively managing its product liability costs.

FORWARD-LOOKING STATEMENTS

This report contains what the Company believes are "forward-looking statements," as that term is defined under the Private Securities Litigation Reform Act of 1995, regarding projections, expectations or matters that the Company anticipates may happen with respect to the future performance of the industries in which the Company operates, the economies of the United States and other countries, or the performance of the Company itself, which involve uncertainty and risk. Such "forward-looking statements" are generally, though not always,

preceded by words such as "anticipates," "expects," "believes," "projects," "intends," "plans," "estimates," and similar terms that connote a view to the future and are not merely recitations of historical fact. Such statements are made solely on the basis of the Company's current views and perceptions of future events, and there can be no assurance that such statements will prove to be true. It is possible that actual results may differ materially from those projections or expectations due to a variety of factors, including but not limited to:

- o changes in economic and business conditions in the world, especially the continuation of the global tensions and risks of further terrorist incidents that currently exist;
- o increased competitive activity, including the inability of the Tire segment to obtain and maintain price increases to offset higher production or material costs;
- o the failure to achieve expected sales levels;
- o consolidation among the Company's competitors and customers;
- o technology advancements;
- o unexpected costs and charges, including those associated with new vehicle launches;
- o fluctuations in raw material and energy prices, including those of steel, crude petroleum and natural gas and the unavailability of such raw materials or energy sources;
- o changes in interest and foreign exchange rates;
- o increased pension expense resulting from investment performance of the Company's pension plan assets and changes in discount rate, salary increase rate, and expected return on plan assets assumptions;
- o government regulatory initiatives, including the proposed and final regulations under the TREAD Act;
- o the cyclical nature and overall health of the global automotive industry, and the impact of the inability of the Company's customers to meet their sales and production goals;
- o changes in the Company's customer relationships, including loss of particular business for competitive or other reasons;
- o the impact of labor problems, including a strike brought against the Company or against one or more of its large customers;
- o litigation brought against the Company;
- o an adverse change in the Company's credit ratings, which could increase its borrowing costs and/or hamper its access to the credit markets;
- o the impact of the disposition of Cooper-Standard Automotive, if completed;
- o the inability of its segments to execute the cost reduction/Asian strategies outlined by each for the coming year; and
- o the impact of reductions in the insurance program covering the principal risks to the Company, and other unanticipated events and conditions.

It is not possible to foresee or identify all such factors. Any forward-looking statements in this report are based on certain assumptions and analyses made by the Company in light of its experience and perception of historical trends, current conditions, expected future developments and other factors it believes are appropriate in the circumstances. Prospective investors are cautioned that any such statements are not a guarantee of future performance and actual results or developments may differ materially from those projected.

The Company makes no commitment to update any forward-looking statement included herein or to disclose any facts, events or circumstances that may affect the accuracy of any forward-looking statement.

Further information covering issues that could materially affect financial performance is contained in the Company's periodic filings with the U. S. Securities and Exchange Commission.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

There have been no material changes in market risk at September 30, 2004 from those detailed in the Company's Annual Report on Form 10-K filed with the U. S. Securities and Exchange Commission ("SEC") for the year ended December 31, 2003.

ITEM 4. CONTROLS AND PROCEDURES

Pursuant to the requirements of the Sarbanes-Oxley Act, the Company's management, with the participation of the Chief Executive Officer and Chief Financial Officer of the Company, has evaluated, as of the end of the period covered by this quarterly report on Form 10-Q, the effectiveness of the Company's disclosure controls and procedures, as defined in Rules 13a-15(e) and 15d-15(e) promulgated under the Securities Exchange Act of 1934, including its internal controls and procedures. Based upon that evaluation, the Chief Executive Officer and the Chief Financial Officer have concluded that, as of the end of such period, the Company's disclosure controls and procedures are effective in identifying the information required to be disclosed in the Company's periodic reports filed with the SEC, including this quarterly report on Form 10-Q, and ensuring that such information is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms.

There have been no changes in the Company's internal control over financial reporting during the quarter ended September 30, 2004 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

The Company is a defendant in various judicial proceedings arising in the ordinary course of business. A significant portion of these proceedings are products liability cases, in which individuals involved in vehicle accidents seek damages resulting from allegedly defective tires manufactured by the Company. Litigation of this type has increased significantly throughout the tire industry following the Firestone tire recall announced in 2000. After reviewing all such proceedings known at the time of this filing, and taking into account all relevant factors concerning them, the Company does not believe that any liabilities resulting from these proceedings, in excess of amounts currently reserved, are reasonably likely to have a material adverse effect on its liquidity, financial condition or results of operations. As a result of the changes in the Company's insurance program effective April 1, 2003, product liability costs could have a greater impact on the consolidated results of operations and financial position of the Company in future periods than in the past and, in some periods, could be material.

On September 17, 2004, the Company announced the signing of a definitive agreement to sell its automotive business, Cooper-Standard Automotive. On September 27, 2004, the United Steel Workers of America ("USWA") filed a complaint asserting they have the right to require the buyer to negotiate new labor agreements affecting four Cooper-Standard facilities before a sale takes place. On November 2, 2004, the U. S. District Court for the Northern District of Indiana issued a preliminary injunction enjoining the sale of those four facilities and ordering expedited arbitration of the USWA grievance to determine whether the proposed sale violates successorship language of the collective bargaining agreements. On November 5, 2004, the Company appealed this decision to the U. S. Seventh Circuit Court of Appeals. The Company believes this ruling will be reversed on appeal and continues to pursue the closing of the transaction according to the original schedule.

ITEM 5. OTHER INFORMATION

(a) On September 11, 2004, management of the North American Tire segment determined that an impairment charge of \$7.3 million with respect to equipment used in the production of radial medium truck tires at the segment's Albany, Georgia facility was required under generally accepted accounting principles. The equipment was tested for impairment as a result of approval of a plan to cease production of radial medium truck tires at the Albany facility by the end of the third quarter of 2005. These tires will be sourced from Asian manufacturers in the future. As a result of the charge, the equipment has been written down to its fair market value as determined by the Company's expectations for proceeds upon its disposition.

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits

- (4) Amendment No. 1 to Amended and Restated Rights Agreement, dated as of May 7, 2004, by and among Cooper Tire & Rubber Company, Fifth Third and Computershare Investor Services, LLC
- (10) Stock Purchase Agreement, dated as of September 16, 2004, by and among Cooper Tire & Rubber Company, Cooper Tyre & Rubber Company UK Limited, and CSA Acquisition Corp.
- (31.1) Certification of Chief Executive Officer pursuant to Rule 13a 14(a)
- (31.2) Certification of Chief Financial Officer pursuant to Rule 13a 14(a)
- (32) Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

(b) Reports on Form 8-K

A Form 8-K (Item 8.01) was furnished on September 8, 2004 relating to the Company's announced cessation of inner tube production.

A Form 8-K (Items 1.01 and 9.01) was furnished on September 17, 2004 relating to the Company's announcement of the execution of a definitive stock purchase agreement to sell its automotive business, Cooper-Standard Automotive.

A Form 8-K (Items 2.02 and 9.01) was furnished on October 21, 2004 relating to the release of the Company's third quarter 2004 earnings.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

COOPER TIRE & RUBBER COMPANY

/S/ P. G. Weaver

P. G. Weaver Vice President and Chief Financial Officer (Principal Financial Officer)

/S/ E. B. White

E. B. White Corporate Controller (Principal Accounting Officer)

November 8, 2004 -----(Date)

Exhibit (4)

May 7, 2004

The Fifth Third Bank Corporate Trust Administration 38 Fountain Square Plaza MD-1090D2 Cincinnati, Ohio 45263 Attention: Trust Officer

Computershare Investor Services, LLC Two North LaSalle Street Chicago, IL 60602 Attention: Relationship Management 312-588-4993 - phone 312-601-4340 - fax

Re: Amendment No. 1 to Amended and Restated Rights Agreement, Removal of Rights Agent Under Cooper Tire & Rubber Company Rights Plan, and Appointment of Successor Rights Agent Under Cooper Tire & Rubber Company Rights Plan (the "Amendment, Removal and Appointment Agreement")

Ladies and Gentlemen:

- 1. Pursuant to Section 27 of the Amended and Restated Rights Agreement, dated as of May 11, 1998 (the "Rights Agreement"), between Cooper Tire & Rubber Company (the "Company"), and The Fifth Third Bank, as rights agent (the "Rights Agent" and, together with the Company, the "Rights Agreement Parties"), the Company, by resolution adopted by its Directors, and the Rights Agent hereby amend the Rights Agreement as follows:
- (a) Section 1(f) of the Rights Agreement is hereby amended and restated in its entirety as follows:
- "'Business Day' shall mean any day other than a Saturday, Sunday or a day on which the New York Stock Exchange is closed."
- (b) Section 18(a) of the Rights Agreement is hereby amended and restated in its entirety as follows:
- "(a) The Company agrees to pay to the Rights Agent reasonable compensation for all services rendered by it hereunder and, from time to time, on demand of the Rights Agent, its reasonable expenses, including reasonable fees and disbursements of its counsel, incurred in connection with the execution and administration of this Agreement and the exercise and performance of its duties hereunder. The Company shall indemnify the Rights Agent for, and hold it harmless against, any loss, liability, or expense, incurred without gross negligence, bad faith or willful misconduct on the part of the Rights Agent, for anything done or omitted by the Rights Agent in connection with the acceptance and administration of this Agreement, including the costs and expenses of defending against any claim of liability hereunder."

- (c) Section 20(c) of the Rights Agreement is hereby amended and restated in its entirety as follows:
- "(c) The Rights Agent shall be liable hereunder only for its own gross negligence, bad faith or willful misconduct."
- (d) Section 31 of the Rights Agreement is hereby amended and restated in its entirety as follows:

"SECTION 31. Governing Law. This Agreement, each Right and each Rights Certificate issued hereunder shall be governed by, and construed in accordance with, the laws of the State of Delaware; provided, however, that Sections 18, 19, 20 and 21 shall be governed by, and construed in accordance with, the laws of the State of Illinois (or the state of incorporation of any successor Rights Agent)."

(e) Section 21 of the Rights Agreement is hereby amended and restated in its entirety as follows:

"SECTION 21. Change of Rights Agent. The Rights Agent or any successor Rights Agent may resign and be discharged from its duties under this Agreement upon thirty days' prior notice in writing mailed to the Company, and to each transfer agent of the Preferred Stock and the Company Common Stock, by registered or certified mail, and to the holders of Rights Certificates by first class mail. The Company may remove the Rights Agent or any successor Rights Agent upon thirty days' prior notice in writing, mailed to the Rights Agent or successor Rights Agent, as the case may be, and to each transfer agent of the Preferred Stock and the Company Common Stock, by registered or certified mail, and to the holders of Rights Certificates by first class mail. If the Rights Agent shall resign or be removed or shall otherwise become incapable of acting, the Company shall appoint a successor to the Rights Agent. If the Company shall fail to make such appointment within a period of thirty days after giving notice of such removal or after it has been notified in writing of such resignation or incapacity by the resigning or incapacitated Rights Agent or by the holder of a Rights Certificate (who shall, with such notice, submit such holder's Rights Certificate for inspection by the Company), then any registered holder of any Rights Certificate may apply to any court of competent jurisdiction for the appointment of a new Rights Agent. Any successor Rights Agent, whether appointed by the Company or by such a court, shall be (a) a corporation, limited liability company or trust company (or similar form of entity under the laws of the United States or any state of the United States), in good standing and authorized to conduct business under the laws of the United States or any state of the United States, which is authorized under such laws to exercise corporate trust or stock transfer powers and is subject to supervision or examination by federal or state authority and which has at the time of its appointment as Rights Agent a combined capital and surplus of at least \$10,000,000 or (b) an Affiliate controlled by an entity described in clause (a) of this sentence. After appointment, the successor Rights Agent shall be vested with the same powers, rights, duties and responsibilities as if it had been originally named as Rights Agent without further act or deed; but the predecessor Rights Agent shall deliver and transfer to the successor Rights Agent any property at the time held by it hereunder, and execute and deliver any further assurance, conveyance, act or deed necessary for the purpose. Not later than the effective date of any such appointment, the Company shall file notice thereof in writing with the predecessor Rights Agent and each transfer agent of the Preferred Stock and the Company Common Stock, and mail a notice thereof in

writing to the registered holders of the Rights Certificates. Failure to give any notice provided for in this Section 21, however, or any defect therein, shall not affect the legality or validity of the resignation or removal of the Rights Agent or the appointment of the successor Rights Agent."

- (f) The Company and ComputerShare Investor Services, LLC ("ComputerShare") agree that the address and contact information set forth above for ComputerShare will be the information for ComputerShare for purposes of Section 25 of the Rights Agreement.
- (g) The Rights Agreement shall not otherwise be supplemented or amended by virtue of this Amendment, Removal and Appointment Agreement, but shall remain in full force and effect.
- (h) Capitalized terms used without other definition in Section 1 of this Amendment, Removal and Appointment Agreement shall be used as defined in the Rights Agreement.
- (i) This Amendment, Removal and Appointment Agreement shall be governed by, and construed in accordance with the laws of the State of Ohio.
- (j) This Amendment, Removal and Appointment Agreement may be executed (including by telecopier) in one or more counterparts, including by the Rights Agreement Parties in separate counterparts, each of which when executed shall be deemed to be an original, but all of which taken together shall constitute one and the same instrument.
- (k) Exhibits B and C to the Rights Agreement shall be deemed amended in a manner consistent with this Amendment, Removal and Appointment Agreement.
- 2. Pursuant to Section 21 of the Rights Agreement, the Company hereby provides 30 days' notice of the Rights Agent's removal as rights agent pursuant to the Rights Agreement, effective as of May 7, 2004, and the Rights Agent hereby accepts and agrees to such removal, and waives the time period for notice of removal under the Rights Agreement, by its countersignature to this Amendment, Removal and Appointment Agreement in the space provided below.
- 3. Pursuant to Section 21 of the Rights Agreement, the Company hereby appoints Computershare as successor to the Rights Agent, as rights agent to act as agent for the Company and its stockholders in accordance with the terms and conditions of the Rights Agreement, effective as of May 7, 2004, and Computershare hereby accepts and agrees to such appointment, also effective as of May 7, 2004, by its countersignature to this Amendment, Removal and Appointment Agreement in the space provided below.

Very truly yours,

COOPER TIRE & RUBBER COMPANY

By: /s/ James E. Kline

James E. Kline, Vice President
General Counsel & Secretary

Accepted and agreed to as of the date first written above:

THE FIFTH THIRD BANK

By: /s/ Randolph J. Stierer Name: Randolph J. Stierer Name: Michael J. Lang Title: Vice President

JEK/rlg

Accepted and agreed to as of the date first written above:

COMPUTERSHARE INVESTOR SERVICES, LLC

By: /s/ Michael J. Lang

Title: Relationship Manager

4

EXHIBIT (10)

STOCK PURCHASE AGREEMENT

AMONG

COOPER TIRE & RUBBER COMPANY,

COOPER TYRE & RUBBER COMPANY UK LIMITED

AND

CSA ACQUISITION CORP.

DATED AS OF

SEPTEMBER 16, 2004

TABLE OF CONTENTS

(continued)

ARTICLE I	DEFINITIONS	PAGE 1
1.1	Certain Defined Terms	1
1.2	Other Interpretive Provisions	12
ARTICLE II	PURCHASE AND SALE	12
2.1	Purchase and Sale of the Shares	12
2.2	Consideration	13
2.3	Adjustment to Cash and Cash Equivalents and Debt Obligations	13
2.4	The Closing.	14
2.5	Deliveries at the Closing.	15
2.6	Structuring Cooperation	16
ARTICLE III	REPRESENTATIONS AND WARRANTIES OF THE SELLERS	16
3.1	Organization	16
3.2	Authorization; Enforceability	17
3.3	Capital Stock of the Sold Companies	17
3.4	Subsidiaries and Joint Ventures	17
3.5	Financial Statements	19
3.6	No Conflicts or Approvals	19
3.7	Compliance with Law; Governmental Authorizations	20
3.8	Proceedings	20
3.9	Absence of Certain Changes	21
3.10	Tax Matters	21
3.11	Employee Benefits	23
3.12	Labor Relations	27
3.13	Intellectual Property	28
3.14	Contracts	28
3.15	Environmental Matters	30
3.16	Insurance	31
3.17	Personal Property Assets	32
3.18	Real Property	32
3.19	Customers and Suppliers	32
3.20	Product Liability	33
3.21	Affiliate Transactions.	33
3.22	Accounts	33
3.23	Inventory	34
3.24	Bank Accounts	34
3.25	No Brokers' or Other Fees	34
3.26	Directors and Officers	34
3.27	No Other Representations or Warranties	34
ARTICLE IV	REPRESENTATIONS AND WARRANTIES OF THE BUYER	34
4.1	Organization	34
4.2	Authorization; Enforceability	34
4.3	No Conflicts or Approvals	35
4.4	Proceedings	35
4.5	No Brokers' or Other Fees.	35

TABLE OF CONTENTS

(continued)

4.6	Investment Intent	PAGE 35
4.7	Financing	36
4.8	No Other Representations or Warranties	36
ARTICLE V	COVENANTS AND AGREEMENTS	36
5.1	Conduct of the Business Prior to the Closing	36
5.2	Access to Books and Records; Confidentiality	39
5.3	Section 338 Election	40
5.4	Tax Returns, Contests and Cooperation	40
5.5	Tax Matters	42
5.6	Refunds and Tax Benefits	44
5.7	Employees; Benefit Plans	45
5.8	Labor Matters	57
5.9	Contact with Customers and Suppliers	58
5.10	Intercompany Debt	58
5.11	Minimum Cash Requirement	59
5.12	Corporate Names	59
5.13 5.14	Restructuring	60
5.14	Consents and Competition	60
5.15	Intentionally omitted	61 61
5.10	Insurance	62
5.17	Noncompetition	62
5.19	Non-Solicitation and Non-Hire of Employees	63
5.20	Acquisition Financing	63
5.21	Letters of Credit on Account of Sold Companies or Subsidiaries	64
5.22	Guarantees.	64
5.23	Buyer Financing	64
ARTICLE VI	CONDITIONS TO CLOSING	64
		- 4
6.1	Conditions to the Obligations of the Parties	64
6.2	Conditions to the Obligation of the Sellers	65
6.3	Conditions to the Obligations of the Buyer	65
ARTICLE VII	TERMINATION	66
7.1	Termination.	66
7.2	Effect of Termination.	66
, . =	222000 02 2023334002031111111111111111111111111111	0.0
ARTICLE VIII	INDEMNIFICATION	66
8.1	Indemnification by the Sellers	66
8.2	Indemnification by the Buyer	68
8.3	Indemnification as Exclusive Remedy	68
8.4	Survival	69
8.5	Limitations on Indemnification	69
8.6	Procedures	72
ARTICLE IX	MISCELLANEOUS	73
9.1	Fees and Expenses	73

TABLE OF CONTENTS

(continued)

		PAGE
9.2	Governing Law	73
9.3	Projections	73
9.4	Amendment	73
9.5	Assignment	73
9.6	Waiver	73
9.7	Notices	74
9.8	Complete Agreement	75
9.9	Counterparts	
9.10	Publicity	75
9.11	Headings	
9.12	Severability	75
9.13	Third Parties	
9.14	Further Assurances	
9.15	Consent to Jurisdiction; Waiver of Jury Trial	76
9.16	Specific Performance	76

SCHEDULES

```
Schedule A
                                      Sold Companies
Schedule B
                                      Restructuring
Schedule C
                                      Capital Expenditure Budget
Schedule 1.1(a) Debt Obligations
Schedule 1.1(b) Knowledge of the Buyer
Schedule 1.1(c)(i) Knowledge of the Sel
                                   Knowledge of the Sellers; Sold Companies or
                                           Subsidiaries
                                    Knowledge of the Sellers; Venture Entities
Schedule 1.1(c)(ii)
Schedule 1.1(d)
Schedule 2.5(a)(vii)
Schedule 2.5(a)(viii)
Resignations of Directors and Officers
Schedule 3.3
                                     Capital Stock of the Sold Companies
Schedule 3.4(a)
                                 Subsidiaries
Schedule 3.4(b)
                           Venture Entities
Financial Statements
Undisclosed Liabilities
Venture Entity Liabilities
No Conflicts
Consents
                               Venture Entities
Schedule 3.5(a)
Schedule 3.5(b)
Schedule 3.5(d)
Schedule 3.6(a)
Schedule 3.6(b)
Schedule 3.7
                                     Compliance with Law; Governmental Authorizations
Schedule 3.8
                                     Proceedings
Schedule 3.9
                                    Absence of Certain Changes
Schedule 3.10
                                      Tax Matters
Schedule 3.11(a)(i)-1
                                      Employee Benefits; Plans
Schedule 3.11(a)(i)-1
Schedule 3.11(a)(i)-2
                                      Employee Benefits; Company Employees
Schedule 3.11(b)
                                      Employee Benefits; IRS Exceptions
Schedule 3.11(c)
                                      Employee Benefits; Compliance
                                     Employee Benefits; Proceedings
Schedule 3.11(f)
                              Employee Benefits; Proceedings
Employee Benefits; Parachute Payments
Employee Benefits; 280G Payments
Employee Benefits; US Employees
Employee Benefits; Foreign Plan Exceptions
Labor Relations
Schedule 3.11(g)
Schedule 3.11(h)
Schedule 3.11(i)
Schedule 3.11(j)
Schedule 3.12
                              Intellectual Property
Intellectual Property Exceptions
Material Contracts
Schedule 3.13(a)
Schedule 3.13(b)
Schedule 3.14(a)
                                    Environmental Matters
Insurance Policies
Schedule 3.15
Schedule 3.16(a)
                             Insurance Folicial
Insurance Exceptions
Personal Property Assets
Toased Real Property
Schedule 3.16(c)
Schedule 3.17
Schedule 3.18(a)

Schedule 3.18(b)

Schedule 3.19

Customers and Suppliers
Schedule 3.20
                                    Product Liability
Affiliate Transactions
Schedule 3.21
Schedule 3.24
                                    Bank Accounts
Schedule 3.26
                                     Directors and Officers
                               Buyer Consents
Schedule 4.3(b)
Schedule 4.4
                                     Buyer Proceedings
Schedule 4.7
                                      Financing
```

Schedule 5.1	Conduct of the Business Prior to the Closing
Schedule 5.1(h)	Executives
Schedule 5.3	Section 338 Elections
Schedule 5.7(a)	Active Company Employees
Schedule 5.7(e)	Former Employee Pension Participants
Schedule 5.7(f)	Former Employee Savings Participants
Schedule 5.7(g)	Assumed Plans
Schedule 5.7(i)(i)	Cooper Severance Obligations
Schedule 5.7(j)	Former Employee Retiree Welfare Benefit Participants
Schedule 5.7(m)	Cooper's SERP
Schedule 5.7(t)	Secunded Employees
Schedule 5.12	Trademarks
Schedule 8.1(a)(vi)(D)	Environmental Matters

EXHIBITS

Exhibit A	Confidentiality Agreement
Exhibit B	Transition Services Agreement
Exhibit C	Severance Procedures

v

STOCK PURCHASE AGREEMENT

This STOCK PURCHASE AGREEMENT, dated as of September 16, 2004, is by and among Cooper Tire & Rubber Company, a Delaware corporation ("COOPER"), Cooper Tyre & Rubber Company UK Limited, a company organized under the laws of England and Wales ("COOPER UK", and, together with Cooper, the "SELLERS") and CSA Acquisition Corp., a Delaware corporation (the "BUYER").

- A. Cooper's automotive business segment, indirectly through the Sold Companies (as defined below), the Subsidiaries (as defined below) and the Venture Entities (as defined below), is engaged in the design, manufacture and sale of fluid handling systems, body sealing systems and active and passive vibration control systems (any of said activities as so conducted anywhere in the world by all, any or each of them, as the context may require, the "BUSINESS");
- B. Cooper and Cooper UK own all of the issued and outstanding shares of capital stock (or equivalent) designated on SCHEDULE 3.3 (such shares, the "SHARES") of the companies listed on SCHEDULE A (such companies, the "SOLD COMPANIES");
- C. The Sellers desire to sell, and the Buyer desires to purchase, the Shares, upon the terms and subject to the conditions set forth in this Agreement; and

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements herein contained and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE I DEFINITIONS

- 1.1 CERTAIN DEFINED TERMS. As used in this Agreement, the following terms have the following meanings:
- "ACQUISITION FINANCING" has the meaning set forth in SECTION 4.7.
- "ACTIVE COMPANY EMPLOYEES" has the meaning set forth in SECTION 5.7(a).
- "ADJUSTED PURCHASE PRICE ALLOCATION" has the meaning set forth in SECTION 2.2(b)(ii).
- "AFFILIATE" means, with respect to any Person, (i) any Person directly or indirectly controlling, controlled by or under common control with such Person, (ii) any Person directly or indirectly owning or controlling ten percent (10%) or more of any class of outstanding equity securities of such Person or (iii) any officer, director, general partner or trustee of any such Person described in clause (i) or (ii).
- "AFFILIATE CONTRACTS" has the meaning set forth in SECTION 3.21.
- "AGREEMENT" means this Stock Purchase Agreement (including the Schedules and Exhibits), as amended, modified or supplemented from time to time.
- "ANNUAL FINANCIAL STATEMENTS" has the meaning set forth in SECTION 3.5(a).
- "APPLICABLE RATE" means a rate per annum equal to the "prime rate" as set forth from time to time in The Wall Street Journal, Eastern Edition, "Money Rates" column.

"APPLICABLE PORTION" means, as to any non-wholly owned Subsidiary or Venture Entity, the percentage ownership interest (direct or indirect) of the Sellers in such entities as of the Closing.

"ARBITRATION FIRM" means the firm of PricewaterhouseCoopers LLP or, if such firm is unable or unwilling to serve in such capacity on terms mutually acceptable to Buyer and Sellers, another independent accounting firm mutually acceptable to Buyer and Sellers.

"ASSUMED PLANS" has the meaning set forth in SECTION 5.7(g).

"BALANCE SHEET" has the meaning set forth in SECTION 3.5(a).

"BUSINESS" has the meaning set forth in the recitals.

"BUSINESS DAY" means any day that is not a Saturday, a Sunday or other day on which banks in New York, New York are required or authorized by Law to be closed.

"BUYER" has the meaning set forth in the preamble.

"BUYER FILED TAX RETURNS" has the meaning set forth in SECTION 5.4(b).

"BUYER INDEMNIFIED PERSONS" has the meaning set forth in SECTION 8.1(a).

"BUYER ACTUARY" has the meaning set forth in SECTION 5.7(s)(i).

"BUYER'S PENSION PLAN" has the meaning set forth in SECTION 5.7(e).

"BUYER'S PRB PLAN" has the meaning set forth in SECTION 5.7(j).

"BUYER'S SAVINGS PLAN" has the meaning set forth in SECTION 5.7(f).

"BUYER'S WELFARE PLANS" has the meaning set forth in SECTION 5.7(d).

"CANADIAN SALARIED PLAN" means the Pension Plan for Salaried Employees of Cooper-Standard Automotive Canada Limited (FSCO Reg. No. 0250548).

"CASH AND CASH EQUIVALENTS" means, as of any date, the fair market value (expressed in United States dollars) of (i) all cash and cash equivalents (including marketable securities and short term investments) of any Sold Company or wholly owned Subsidiary plus (ii) the Applicable Portion of cash and cash equivalents (including marketable securities and short term investments) of any non-wholly owned Subsidiary or Venture Entity. Any monetary conversion from the currency of a foreign country to United States dollars shall be calculated using the applicable exchange rates set forth in The Wall Street Journal, Eastern Edition.

"CAPITAL EXPENDITURE BUDGET" means the capital expenditure with respect to the Business for the calendar year 2004, as set forth on SCHEDULE C.

"CELLECT" means Cellect LLC.

"CHANGE OF CONTROL OF COOPER" means the occurrence of any of the following events after the Closing Date: (a) any Person engaged in a Competitive Business is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Securities Exchange Act of 1934), directly or indirectly, of securities of Cooper representing 50% or more of the combined voting power of Cooper's then-

outstanding voting securities; (b) Cooper merges or consolidates into itself, or is merged or consolidated with, another Person engaged in a Competitive Business and as a result of such merger or consolidation less than 50% of the voting power of the then-outstanding voting securities of the surviving or resulting entity immediately after such transaction are directly or indirectly beneficially owned in the aggregate by the former stockholders of Cooper immediately prior to such transaction; or (c) all or substantially all the assets accounted for on the consolidated balance sheet of Cooper are sold or transferred to one or more Persons engaged in a Competitive Business.

"CLAIM" has the meaning set forth in SECTION 8.6(a).

"CLAIM NOTICE" has the meaning set forth in SECTION 8.6(a).

"CLAIM RESPONSE" has the meaning set forth in SECTION 8.6(a).

"CLOSING" has the meaning set forth in SECTION 2.4.

"CLOSING DATE" has the meaning set forth in SECTION 2.4.

"CODE" has the meaning set forth in SECTION 2.2(c).

"COMPANY BENEFIT PLAN" has the meaning set forth in SECTION 3.11(a)(i).

"COMPANY EMPLOYEE" has the meaning set forth in SECTION 3.11(a)(i).

"COMPANY FOREIGN BENEFIT PLAN" has the meaning set forth in SECTION 3.11(a)(ii).

"COMPANY U.S. BENEFIT PLAN" has the meaning set forth in SECTION 3.11(a)(ii).

"COMPETITION/INVESTMENT LAW" means any Law that is designed or intended to prohibit, restrict or regulate (a) foreign investment or (b) antitrust, monopolization, restraint of trade or competition.

"COMPETITIVE BUSINESS" has the meaning set forth in SECTION 5.18(a).

"CONFIDENTIALITY AGREEMENT" means the confidentiality agreement between the Buyer and Cooper, a copy of which is attached as EXHIBIT A.

"CONSENT" means any consent, approval, order, Permit, authorization, waiver, report or notification required to be obtained from, filed with or delivered to any Governmental Authority or other third party in connection with the execution, delivery and performance of this Agreement and the Transaction Agreements, the consummation of the transactions contemplated hereby and thereby or compliance by the Sellers, the Subsidiaries and the Sold Companies with any of the provisions hereof and thereof.

"CONTRACT" means any contract, agreement, obligation, plan, indenture, note, bond, loan, instrument, lease (including real property leases), conditional sale contract, mortgage, license, Permit, franchise, insurance policy, undertaking, commitment or other enforceable arrangement or agreement, whether written or oral.

"CONTROLLED GROUP" has the meaning set forth in SECTION 3.11(c).

"COOPER" has the meaning set forth in the preamble.

"COOPER'S ACTUARY" has the meaning set forth in SECTION 5.7(e)(v).

"COOPER'S FLEXIBLE ACCOUNT PLAN" has the meaning set forth in SECTION 5.7(k).

"COOPER'S PENSION PLAN" has the meaning set forth in SECTION 5.7(e).

"COOPER'S SAVINGS PLAN" has the meaning set forth in SECTION 5.7(f).

"COOPER'S SERP" has the meaning set forth in SECTION 5.7(m).

"COOPER UK" has the meaning set forth in the preamble.

"CRAIG ASSEMBLY" means Craig Assembly Inc., a Michigan corporation.

"CSA" has the meaning set forth in SECTION 5.7(t).

"CSA UK FLUID" has the meaning set forth in SECTION 5.7(s)(i).

"CURRENT TAX MATTER" means any claim, suit, action, audit, litigation or proceeding relating to Taxes solely in respect of a Pre-Closing Tax Period (other than any Transfer Pricing Tax Matter) that is in progress as of the Closing Date.

"DEBT FINANCING" has the meaning set forth in SECTION 4.7.

"DEBT FINANCING COMMITMENT" has the meaning set forth in SECTION 4.7.

"DEBT OBLIGATIONS" means, with respect to any Person as of any date without duplication, all (a) indebtedness for borrowed money of such Person (including principal and accrued interest), including under those agreements set forth on SCHEDULE 1.1(a), (b) obligations of such Person in respect of capitalized leases required to be recorded as such on a balance sheet prepared in accordance with GAAP and obligations of such Person for the deferred purchase price of goods or services (other than trade payables incurred in the ordinary course of business), (c) obligations of such Person in respect of banker's acceptances or letters of credit issued or created for the account of such Person (excluding any letters of credit issued or created by or for the benefit of such Person relating to workers' compensation, workers entitlement guarantees or employers' liability obligations of such Person), (d) all indebtedness or obligations of such Person of the types referred to in the preceding clauses (a) and (b) secured by any Lien on any assets of such Person, (e) guarantees of obligations of any other Person of the type described in clauses (a) and (b) above by such Person, and (f) any payment obligation in respect of interest under any existing interest rate swap, hedge or similar agreement entered into by any Person with respect to any indebtedness described in clause (a) above, including under those agreements set forth on SCHEDULE 1.1(a).

"DEDUCTIBLE" has the meaning set forth in SECTION 8.1(b).

"DISCONTINUED BUSINESS EMPLOYEE" has the meaning set forth in SECTION 3.11(a)(i).

"EC MERGER REGULATION" means Council Regulation (EEC) 4064/89 of the European Community, as amended.

"EMPLOYMENT AGREEMENT" means the Employment Agreement, dated June 6, 2000, by and between Cooper and James S. McElya, as amended by that certain First Amendment to Employment Agreement dated as of February 4, 2004.

"ENVIRONMENT" means the outdoor or indoor environment, including soil, surface waters, groundwater, land, stream sediments, surface or subsurface strata or ambient air.

"ENVIRONMENTAL CLAIM" means any written notice, or, to the Knowledge of the Sellers, any oral notice, or Proceeding by any Person alleging liability or potential liability relating to any Environmental Losses or in respect of any Environmental Laws.

"ENVIRONMENTAL LAW" means any Law, including common law, relating to the protection of the Environment, natural resources or, to the extent relating to exposure to Hazardous Materials, human health or safety.

"ENVIRONMENTAL LOSSES" means Losses arising from a Release or threatened Release of Hazardous Materials or noncompliance with or liability or Loss under any Environmental Law.

"EQUITY FINANCING" has the meaning set forth in SECTION 4.7.

"EQUITY FINANCING COMMITMENT" has the meaning set forth in SECTION 4.7.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder.

"ESTIMATED CASH AND CASH EQUIVALENTS" has the meaning set forth in SECTION 2.3(a).

"ESTIMATED CASH AND DEBT STATEMENT" has the meaning set forth in SECTION 2.3(a).

"ESTIMATED DEBT OBLIGATIONS" has the meaning set forth in SECTION 2.3(a).

"FACTORING AGREEMENTS" means collectively the (a) Professional Receivables Assignment Agreement, dated as of April 30, 1997, by and between Standard Products Industriel (n.k.a. Cooper-Standard Automotive France S.A.S.) and Banque Française du Commerce Exterior (n.k.a. Natexis Banques Populaires);

(b) Discount With Limited Recourse Agreement, dated as of June 18, 2003, by and between Cooper-Standard Automotive France S.A.S. and Credit Lyonnais; and (c) Discount Without Any Recourse Agreement, dated as of December 16, 2002, by and between Cooper-Standard Automotive France S.A.S. and Credit Commercial de France.

"FIFO" has the meaning set forth in SECTION 3.23.

"FINAL CASH AND CASH EQUIVALENTS" means the amount of Cash and Cash Equivalents as determined pursuant to SECTION 2.3(c).

"FINAL CASH AND DEBT STATEMENT" has the meaning set forth in SECTION 2.3(b).

"FINAL DEBT OBLIGATIONS" means the amount of Debt Obligations of the Sold Companies and wholly-owned Subsidiaries and Venture Entity and Non-Wholly Owned Subsidiary Debt Obligations as determined pursuant to SECTION 2.3(c).

"FINAL PURCHASE PRICE" means an amount equal to the Preliminary Purchase Price as adjusted to reflect the differences, if any, between (a) Estimated Cash and Cash Equivalents and Final Cash and Cash Equivalents and (b) Estimated Debt Obligations and Final Debt Obligations.

"FINAL TRANSFER DATE" has the meaning set forth in SECTION 5.7(e)(iv).

"FINANCIAL STATEMENTS" has the meaning set forth in SECTION 3.5(a).

"FINANCING COMMITMENTS" has the meaning set forth in SECTION 4.7.

"FORMER EMPLOYEES" means former employees of the Sold Companies, the Subsidiaries and the Sellers, who were employed primarily in the Business.

"GAAP" means United States generally accepted accounting principles and practices as in effect from time to time.

"GENERAL ENFORCEABILITY EXCEPTIONS" has the meaning set forth in SECTION 3.2.

"GOVERNMENTAL AUTHORITY" means any government or other political subdivision (whether federal, state, provincial, local or foreign), or any agency or instrumentality of any such government or political subdivision, or any federal, state, provincial, local or foreign court or arbitrator.

"GOVERNMENTAL ORDER" means any judgment, order, writ, assessment, injunction, decree or ruling of any Governmental Authority.

"GOVERNMENT SCHEMES" means any mandatory government-sponsored or maintained agreements, arrangements, customs, practices or obligations under any Law to which any company contributes in compliance with applicable Law in existence as of the date of this Agreement for the payment of, provision for, or contribution towards, any pensions, allowances, lump sums or other like benefits on retirement, death, termination of employment (whether voluntary or not), or during periods of sickness or disablement, which are for the benefit of an employee or the benefit of persons dependent on any employee.

"HAZARDOUS MATERIALS" means any pollutant, toxic substance, asbestos and asbestos-containing materials, hazardous waste, solid waste, hazardous material, hazardous substance, contaminant, petroleum, petroleum-containing materials, radiation and radioactive materials and polychlorinated biphenyls as defined in, or regulated by, any Environmental Law and any other material that could result in liability under any Environmental Law.

"HSR ACT" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"INCOME TAXES" means any Tax imposed on or measured by income.

"INDEMNIFIED PARTY" means a party entitled to indemnification under this Agreement.

"INDEMNIFYING PARTY" means a party obligated to provide indemnification under this Agreement.

"INSURANCE POLICIES" has the meaning set forth in SECTION 3.16(a).

"INTELLECTUAL PROPERTY" means all (a) patents, (b) inventions, discoveries, processes, formulae, designs, models, industrial designs, know-how, confidential information, proprietary information and trade secrets, whether or not patented or patentable, (c) trademarks, service marks, trade names, brand names, trade dress, slogans, logos, internet domain names and all goodwill associated therewith, (d) copyrights and other copyrightable works and works in progress, databases and software, (e) all other intellectual property rights and foreign equivalent or counterpart rights and forms of

protection of a similar or analogous nature or having similar effect in any jurisdiction throughout the world, (f) any renewals, extensions, continuations, divisionals, reexaminations or reissues or equivalent or counterpart of any of the foregoing in any jurisdiction throughout the world, and (g) all registrations and applications for registration of any of the foregoing.

"INTERIM FINANCIAL STATEMENTS" has the meaning set forth in SECTION 3.5(a).

"IRS" has the meaning set forth in SECTION 3.11(b).

"KNOWLEDGE OF THE BUYER" means the actual knowledge of the individuals listed on SCHEDULE 1.1(b).

"KNOWLEDGE OF THE SELLERS" means (a) when used with reference to the Sold Companies or Subsidiaries, the actual knowledge of the individuals listed on SCHEDULE 1.1(c)(i) and (b) when used with reference to the Venture Entities, the actual knowledge of the individuals listed on SCHEDULE 1.1(c)(ii).

"LAW" means any law, statute, code, ordinance, treaty, Governmental Order, rule or regulation of any Governmental Authority.

"LEASED REAL PROPERTY" has the meaning set forth in SECTION 3.18(a).

"LIABILITIES" means any indebtedness, obligations or liabilities of any kind (whether accrued, absolute, contingent or otherwise, and whether or not due or to become due or asserted or unasserted).

"LIEN" means any voting trust, shareholder agreement, proxy or other similar restriction, lien, mortgage, pledge, security interest, or other encumbrance.

"LIFO" has the meaning set forth in SECTION 3.23.

"LOSSES" means any and all claims, liabilities, losses, damages, fines, penalties, judgments and costs (in each case including reasonable out-of-pocket expenses).

"MATERIAL CONTRACTS" has the meaning set forth in SECTION 3.14(a).

"MAXIMUM AMOUNT" has the meaning set forth in SECTION 8.1(b).

"NET INTERCOMPANY AMOUNTS" has the meaning set forth in SECTION 5.10.

"NISHIKAWA" means Nishikawa Standard Company, a Delaware general partnership.

"NLRA" has the meaning set forth in SECTION 5.8(c).

"NLRB" has the meaning set forth in SECTION 3.12(a).

"OLIVER RUBBER EMPLOYEE" means the individual identified on SCHEDULE 5.7(v).

"ORDINARY COURSE OF BUSINESS" means, in all material respects, the usual, regular and ordinary course of a business consistent with the past practice thereof.

"ORGANIZATIONAL DOCUMENT" means, as to any Person, its certificate or articles of incorporation, its regulations or by-laws or any equivalent documents under the Law of such Person's jurisdiction of incorporation or organization.

"OUTSIDE DATE" has the meaning set forth in SECTION 7.1(b).

"OWNED REAL PROPERTY" has the meaning set forth in SECTION 3.18(b).

"PBGC" has the meaning set forth in SECTION 3.11(c).

"PENSION PARTICIPANTS" has the meaning set forth in SECTION 5.7(e).

"PERIODIC COSTS" has the meaning set forth in SECTION 5.7(s)(i).

"PERMITS" has the meaning set forth in SECTION 3.7.

"PERMITTED LIENS" means (a) Liens arising under the Debt Obligations as set forth on SCHEDULE 1.1(d), (b) Liens for Taxes, assessments and other charges of Governmental Authorities not yet due and payable or being contested in good faith by appropriate proceedings, (c) mechanics', workmen's, repairmen's, warehousemen's, carriers' or other like Liens arising or incurred in the ordinary course of business or by operation of Law if the underlying obligations are not delinquent, (d) with respect to the Real Property (i) any conditions that may be shown by a current, accurate survey, (ii) easements, encroachments, restrictions, rights of way and any other non-monetary title defects; and (iii) zoning, building and other similar restrictions; provided none of the foregoing described in clause (d) will individually or in the aggregate materially impair the value or continued use and operation of the property to which they relate in the Business as presently conducted.

"PERSON" means any individual, sole proprietorship, partnership, firm, corporation, association, trust, unincorporated organization, joint venture, limited liability company, Governmental Authority or other legal entity.

"PLAN" has the meaning set forth in SECTION 3.11(a)(i).

"POST-CLOSING LOSS" has the meaning set forth in SECTION 5.6(a).

"PRE-CLOSING TAX MATTER" has the meaning set forth in SECTION 5.4(e)(ii).

"POST-CLOSING TAX PERIOD" means a taxable period (or in the case of a Straddle Period, the portion thereof) ending after the Closing Date.

"PRE-CLOSING EMPLOYER'S LIABILITY INSURANCE CLAIMS" has the meaning

set forth in SECTION 5.7(n)(ii).

"PRE-CLOSING WORKERS' COMPENSATION CLAIMS" has the meaning set forth in SECTION 5.7(n)(i).

"PRE-CLOSING SELF INSURED WORKERS' COMPENSATION ARRANGEMENTS" has

the meaning set forth in SECTION 5.7(n)(i).

"PRE-CLOSING TAX PERIOD" means a taxable period (or in the case of a Straddle Period, the portion thereof) ending on or before the close of business on the Closing Date.

"PRELIMINARY PURCHASE PRICE" has the meaning set forth in SECTION 2.2(a).

"PROCEEDING" means any claim, action, suit, proceeding, administrative enforcement proceeding or arbitration proceeding before any Governmental Authority.

"PURCHASE PRICE ALLOCATION" has the meaning set forth in SECTION 2.2(b).

"RABBI TRUST" has the meaning set forth in SECTION 5.7(i)(ii).

"REAL PROPERTY" means the Owned Real Property and the Leased Real Property.

"RELEASE" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, or disposing of a Hazardous Material into the Environment.

"RESPONSE PERIOD" has the meaning set forth in SECTION 8.6(a).

"RESTRAINT" has the meaning set forth in SECTION 6.1(a).

"RESPONSIBLE PARTY" has the meaning set forth in SECTION 8.6(b)(ii).

"RESTRUCTURING" has the meaning set forth in SECTION 5.13.

"RETAINED NAMES" has the meaning set forth in SECTION 5.12(a).

"SAVINGS PARTICIPANTS" has the meaning set forth in SECTION 5.7(f).

"SAVINGS TRANSFER DATE" has the meaning set forth in SECTION 5.7(f)(iii).

"SCHENECTADY FACILITY MATTER" means the Settlement Agreement and Mutual Release, dated as of October 8, 2003, by and among Standard, Sentinel and Cellect, the Environmental Indemnification Agreement, dated as of March 1, 1998, by and between Schenectady County Industrial Development Agency ("SCIDA") and Standard Products Company, the Environmental Indemnification Agreement, dated as of October 8, 2003, by and between Standard and the SCIDA and the Agreement Concerning Campbell Plastics Building, dated as of October 8, 2003, by and among the SCIDA, Standard, Sentinel and Cellect, pursuant to which Standard has agreed to share the costs of demolition and disposal of the building and agreed to indemnify SCIDA, Sentinel and Cellect for certain matters relating to the property at or about 2900 Campbell Avenue in the Town of Rotterdam and City of Schenectady, New York.

"SCIDA" means the Schenectady County Industrial Development Agency.

"SECUNDED EMPLOYEES" has the meaning set forth in SECTION 5.7(t).

"SECUNDMENT PERIOD" has the meaning set forth in SECTION 5.7(t).

"SECURITIES ACT" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"SELLER BENEFIT PLAN" has the meaning set forth in SECTION 3.11(a)(iii).

"SELLER INDEMNIFIED PERSONS" has the meaning set forth in SECTION 8.2(a).

"SELLER TAXES" shall mean (a) all Taxes imposed on or payable with respect to the Sold Companies or the Subsidiaries relating to a Pre-Closing Tax Period, (b) all Taxes of any member of an affiliated, consolidated, combined or unitary group of which the Sold Companies or the Subsidiaries (or any predecessor of any of the foregoing) are or were a member on or prior to the Closing Date, including pursuant to Treasury Regulation Section 1.1502-6 or any analogous or similar state, local, or foreign law or regulation, as a transferee or successor, by contract or otherwise, (c) any Taxes (including Transfer Taxes) arising from or in connection with the Restructuring, or (d) any Taxes arising from or in connection with any breach by Sellers of any representation, warranty or covenant contained in this Agreement, in each case, to the extent such Taxes are the liability of Sellers pursuant to SECTION 5.5(a) hereof.

"SELLER WELFARE PLANS" has the meaning set forth in SECTION 5.7(d).

"SELLERS' ACTUARY" has the meaning set forth in SECTION 5.7(s)(i).

"SELLER'S LONG-TERM DISABILITY PLAN" has the meaning set forth in SECTION 5.7(q).

"SELLERS" has the meaning set forth in the preamble.

"SENTINEL" means Sentinel Products Corp.

"SEVERANCE PLAN" means the Cooper Tire & Rubber Company Change in Control Severance Pay Plan.

"SHARES" has the meaning set forth in the recitals.

"SIEBE ACTUARY" has the meaning set forth in SECTION 5.7(s)(i).

"SIEBE PLAN" has the meaning set forth in SECTION 5.7(s)(i).

"SIEBE PURCHASE AGREEMENT" has the meaning set forth in SECTION 8.1(a).

"SIEBE RULES" has the meaning set forth in SECTION 5.7(s)(i).

"SOLD COMPANIES" has the meaning set forth in the recitals.

"SOLD COMPANY BENEFIT PLAN" has the meaning set forth in SECTION 3.11(a)(iii).

"SOLD COMPANY MATERIAL ADVERSE EFFECT" means any state of facts, change, occurrence or development that (i) directly or indirectly prevents or materially impairs or delays the ability of any Seller to perform its obligations hereunder or (ii) has a material adverse effect on the condition (financial or otherwise), results of operations, business, properties, assets or Liabilities of the Business or the Sold Companies, Subsidiaries and Venture Entities taken as a whole, but excludes any effect (a) resulting from general economic conditions (whether as a result of acts of terrorism, war (whether or not declared), armed conflicts or otherwise) and (b) impacting companies in the industry in which the Business is conducted generally, in each such case except to the extent such effects, facts, change, occurrence or development has a disproportionately adverse impact on the Business or the Sold Companies, Subsidiaries and Venture Entities.

"STANDARD" means Cooper-Standard Automotive Inc.

"STRADDLE PERIOD" means any taxable period which begins before and ends after the Closing Date.

"STRADDLE PERIOD TAX MATTER" has the meaning set forth in SECTION 5.4(e)(iii).

"SUBSIDIARIES" means the Persons listed on SCHEDULE 3.4(a), constituting all of the direct and indirect majority or wholly owned subsidiaries of the Sold Companies.

"TAX" or "TAXES" means any income, alternative or add-on minimum, gross receipts, sales, use, ad valorem, franchise, profits, license, transfer, withholding, payroll, employment, excise, severance, stamp, occupation, premium, value added, property, environmental or windfall profits taxes, customs, duties or similar fees, assessments or charges of any kind whatsoever, together with any interest and any penalties, additions to tax or additional amounts imposed by any Taxing Authority.

"TAX CLAIM" has the meaning set forth in SECTION 5.4(e)(i).

"TAX RESERVE" means the amount of any liability, accrual or reserve for Taxes other than Income Taxes reflected on the Balance Sheet (excluding reserves for deferred Taxes), as adjusted solely for operations and transactions through the Closing Date consistent with past custom and practice.

"TAX RETURN" means any and all returns (including amended returns), filings, statements or similar reports relating to Taxes.

"TAXING AUTHORITY" means any Governmental Authority responsible for the administration or imposition of any Tax.

"THIRD PARTY CLAIM" has the meaning set forth in 8.6(b)(i).

"TRADEMARKS" has the meaning set forth in SECTION 5.12(b).

"TRANSACTION" has the meaning set forth in SECTION 5.8(c).

"TRANSACTION AGREEMENTS" means (a) with respect to the Shares, such instruments of sale, conveyance, transfer and assignment, and such other agreements or documents, if any, necessary under Law or contemplated by this Agreement in order to transfer all right, title and interest of the applicable Sellers in such Shares in accordance with the terms hereof and (b) the Transition Services Agreements.

"TRANSFER AMOUNT" has the meaning set forth in SECTION 5.7(e)(v).

"TRANSFER PRICING TAX MATTER" means any audit or other administrative or judicial Proceeding that relates to transfer pricing or other similar issues for any of the Sold Companies or the Subsidiaries for a Pre-Closing Tax Period or Straddle Period.

"TRANSFER TAXES" has the meaning set forth in SECTION 5.4(c).

"TRANSFERRED BENEFIT LIABILITY" has the meaning set forth in SECTION 5.7(e)(v).

"TRANSITION SERVICES AGREEMENTS" means (a) the Transition Services Agreement to be entered into between Cooper and Buyer and (b) the two Supply Agreements to be entered into between Cooper and Standard, substantially in the forms attached hereto as EXHIBIT B.

"TREASURY REGULATIONS" means the Treasury Regulations promulgated under the Code.

"UNION EMPLOYEES" has the meaning set forth in SECTION 5.7(c).

"VENTURE ENTITIES" means Craig Assembly and Nishikawa.

"VENTURE ENTITY AND NON-WHOLLY OWNED SUBSIDIARY DEBT OBLIGATIONS" means the Applicable Portion of Debt Obligations of the Venture Entities and the non-wholly owned Subsidiaries.

"VENTURE FINANCIAL STATEMENTS" has the meaning set forth in SECTION 3.5(c).

"VENTURE ENTITY PLAN" means each Plan which is sponsored, entered into or maintained by any Venture Entity.

- "WAIVER" means a fully executed, binding, and irrevocable waiver entered into prior to the Closing Date (i) with respect to James S. McElya, of any claims or right he may have under the Employment Agreement to termination or severance benefits, other than as a result of the termination of his employment during the Term (as defined in the Employment Agreement) (x) following the Closing Date by the Buyer without Cause (as defined in the Employment Agreement) or (y) following the Closing Date by Mr. McElya due to the failure of the Buyer to provide him with the employment terms and compensation set forth in definitive employment documentation to be entered into with Buyer prior to the Closing Date, which failure is not remedied by the Buyer within ten (10) calendar days after receipt by the Buyer of written notice of such failure; and
- (ii) with respect to Paul C. Gilbert, Edward A. Hasler, James W. Pifer, Gary T. Phillips, Allen J. Campbell, Michael C. Verwilst, Helen T. Yantz and Larry J. Beard, of any claims or right he or she may have under the Severance Plan to termination or severance benefits, other than as a result of the termination of their employment during the Severance Period (as defined in the Severance Plan)
- (x) following the Closing Date by the Buyer without Cause (as defined in the Severance Plan) or (y) following the Closing date by such individual due to the failure of the Buyer to provide him or her with the employment and compensation set forth in definitive employment documentation to be entered into with Buyer prior to the Closing Date, which failure is not remedied by the Buyer within ten (10) calendar days after receipt by the Buyer of written notice of such failure.

"WARN ACT" has the meaning set forth in SECTION 3.12(a).

"WICKES AGREEMENT" means the Settlement Agreement and Mutual Release entered into as of the 1st day of June, 1998, by and between Cooper and Wickes Manufacturing Company.

1.2 OTHER INTERPRETIVE PROVISIONS. The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole (including any Schedules and Exhibits hereto) and not to any particular provision of this Agreement, and all Article, Section, Schedule and Exhibit references are to this Agreement unless otherwise specified. The words "include," "includes" and "including" will be deemed to be followed by the phrase "without limitation." The meanings given to terms defined herein will be equally applicable to both the singular and plural forms of such terms. Whenever the context may require, any pronoun includes the corresponding masculine, feminine and neuter forms. Except as otherwise expressly provided herein, all references to "dollars" or "\$" will be deemed references to the lawful money of the United States of America.

ARTICLE II PURCHASE AND SALE

2.1 PURCHASE AND SALE OF THE SHARES. On the Closing Date and subject to the terms and conditions set forth in this Agreement, Cooper and Cooper UK shall sell, convey, assign and transfer to the Buyer and the Buyer will purchase and acquire, all of Cooper's and Cooper UK's right, title and interest in and to the Shares, in each case, free and clear of all Liens, other than such Liens as may be created by or on behalf of the Buyer.

- 2.2 CONSIDERATION. (a) On the Closing Date and subject to the terms and conditions set forth in this Agreement, the Buyer will pay to Cooper and Cooper UK, in consideration of the sale, assignment and transfer of the Shares, the aggregate sum of \$1,157,500,000 in cash (in U.S. dollars) plus all Estimated Cash and Cash Equivalents minus all Estimated Debt Obligations (as adjusted pursuant to SECTION 2.3, the "PRELIMINARY PURCHASE PRICE"). Such amount shall be paid to the applicable Seller on the Closing Date by means of one or more wire transfers of immediately available funds to an account or accounts designated in writing by Cooper at least one Business Day prior the Closing Date.
- (b) (i) On or prior to the Closing Date, Buyer and Sellers shall jointly prepare and agree in good faith to a statement allocating the Preliminary Purchase Price among the Shares sold by each Seller and the non-competition and non-solicitation covenants contained herein (the "PURCHASE PRICE ALLOCATION") in accordance with applicable Law. If the parties cannot agree in good faith within the time period set forth above, then Cooper and Buyer jointly shall engage the Arbitration Firm to resolve such dispute and whose fees shall be borne equally by Cooper and Buyer. The determination by the Arbitration Firm shall be binding on the parties. None of the Sellers, the Buyer or their respective Affiliates shall take any position inconsistent with the Purchase Price Allocation on any Tax Return or in any audit or other proceeding relating to Taxes unless otherwise required by applicable Law.
- (ii) If an adjustment is made to the (A) Preliminary Purchase Price pursuant to SECTION 2.3 hereof or otherwise or (B) Final Purchase Price, the parties shall jointly agree to allocate such adjustment among the Shares sold by each Seller and the non-competition and non-solicitation covenants contained herein (the "ADJUSTED PURCHASE PRICE ALLOCATION"). If after good faith negotiations, the parties cannot agree upon the allocation of such adjustment among the Shares sold by each Seller and such non-competition and non-solicitation covenants within thirty (30) days after such adjustment was made, then Cooper and Buyer jointly shall engage the Arbitration Firm to resolve such dispute and whose fees shall be borne equally by Cooper and Buyer. The determination by the Arbitration Firm shall be binding on the parties. None of the Sellers, the Buyer or their respective Affiliates shall take any position inconsistent with the Adjusted Purchase Price Allocation on any Tax Return or in any audit or other proceeding relating to Taxes unless otherwise required by Law.
- (c) Notwithstanding anything in this Agreement to the contrary, if any of the Sellers fails to provide the Buyer with the certification provided in

SECTION 2.5(a)(v) in whole or in part, the Buyer shall be entitled to withhold the requisite amount from the Preliminary Purchase Price in accordance with

Section 1445 of the Internal Revenue Code of 1986, as amended (the "CODE") and the Treasury Regulations promulgated thereunder or other applicable Law.

2.3 ADJUSTMENT TO CASH AND CASH EQUIVALENTS AND DEBT OBLIGATIONS. Five (5) Business Days prior to the Closing Date, Cooper shall prepare and deliver to Buyer (i) a statement (the "ESTIMATED CASH AND DEBT STATEMENT") of (A) a good faith estimate of the amount of Cash and Cash Equivalents anticipated to exist immediately prior to the Closing (the "ESTIMATED CASH AND CASH EQUIVALENTS"), and (B) a good faith estimate of the Debt Obligations of the Sold Companies and the wholly-owned Subsidiaries and the Venture Entity and Non-Wholly Owned Subsidiary Debt Obligations anticipated to be outstanding immediately prior to the Closing ("ESTIMATED DEBT OBLIGATIONS") and (ii) a statement reflecting a good faith estimate of the combined balance sheet of the Business as of the Closing. The Estimated Cash and Debt Statement shall be subject to the review of Buyer, and Cooper and the Buyer will cooperate and negotiate in good faith to resolve any dispute regarding the Estimated Cash and Debt Statement prior to the Closing; provided that if any item of dispute regarding the Estimated Cash and Debt Statement is not resolved by agreement in writing between Cooper and the Buyer two (2) Business Days prior to the Closing Date, then Cooper's estimate of such disputed item together with the resolved disputed items and the undisputed items contained in the Estimated Cash and Debt Statement

shall be deemed final and shall be deemed the Estimated Cash and Cash Equivalents and the Estimated Debt Obligations for purposes of SECTION 2.2(a).

- (b) Within 45 days following the Closing Date, Buyer shall prepare and deliver to Cooper a statement setting forth the Final Cash and Cash Equivalents, the Final Debt Obligations and the Final Purchase Price (the "FINAL CASH AND DEBT STATEMENT").
- (c) Within fifteen (15) days following receipt by Cooper of the Final Cash and Debt Statement, Cooper shall deliver written notice to Buyer of any dispute it has with respect to the preparation or content of the Final Cash and Debt Statement. Such notice must describe in reasonable detail the items contained in the Final Cash and Debt Statement that Cooper disputes and the basis for any such disputes. If Cooper does not notify Buyer of a dispute with respect to the Final Cash and Debt Statement within such 15-day period, such Final Cash and Debt Statement will be final, conclusive and binding on the parties. In the event of such notification of a dispute, Cooper and Buyer shall negotiate in good faith to resolve such dispute. If Cooper and Buyer, notwithstanding such good faith effort, fail to resolve such dispute within ten (10) days after Cooper advises Buyer of its objections, then Cooper and Buyer jointly shall engage the Arbitration Firm to resolve such dispute. As promptly as practicable thereafter, Cooper and Buyer shall each prepare and submit a written presentation to the Arbitration Firm. As soon as practicable thereafter, Cooper and Buyer shall cause the Arbitration Firm to choose one of the party's positions based solely upon the presentation by Cooper and Buyer. All fees and expenses relating to this work of the Arbitration Firm shall be borne by the party whose position was not selected by the Arbitration Firm. All determinations made by the Arbitration Firm will be final, conclusive and binding on the parties.
- (d) For purposes of complying with the terms set forth in this

SECTION 2.3, each party shall cooperate with and make available to the other parties and their respective representatives all information, records, data and working papers, and shall permit reasonable access to its facilities and personnel, as may be reasonably required in connection with the preparation and analysis of the Final Cash and Debt Statement and the resolution of any disputes thereunder.

- (e) If Final Purchase Price (as finally determined pursuant to
- SECTION 2.3(c)) is less than the Preliminary Purchase Price, Cooper shall pay Buyer by means of one or more wire transfers of immediately available funds to an account or accounts designated in writing by Buyer an amount in cash (in U.S. dollars) equal to such shortfall (together with interest thereon at the Applicable Rate from and including the Closing Date to, but excluding, the date of such payment). Such payment is to be made within five (5) Business Days of the date on which Final Purchase Price is finally determined pursuant to SECTION 2.3(c).
- (f) If Final Purchase Price (as finally determined pursuant to

SECTION 2.3(c)) is greater than the Preliminary Purchase Price, Buyer shall pay Cooper by means of one or more wire transfers of immediately available funds to an account or accounts designated in writing by Cooper an amount in cash (in U.S. dollars) equal to such excess (together with interest thereon at the Applicable Rate from and including the Closing Date to, but excluding, the date of such payment). Such payment is to be made within five (5) Business Days from the date on which the Final Purchase Price is finally determined pursuant to SECTION 2.3(c).

2.4 THE CLOSING. Unless this Agreement is terminated pursuant to ARTICLE VII, the closing of the transactions contemplated by this Agreement (the "CLOSING") shall take place at the offices of Simpson Thacher & Bartlett LLP, 425 Lexington Avenue, New York, New York 10017, not later than the third Business Day following the satisfaction or waiver of the conditions set forth in ARTICLE VI hereof

(other than those conditions that are to be satisfied at the Closing), or at such other place and on such other date or time as may be agreed upon by the Sellers and the Buyer (the "CLOSING DATE").

- 2.5 DELIVERIES AT THE CLOSING. (a) At or prior to the Closing, the Sellers shall deliver or cause to be delivered to the Buyer:
- (i) stock certificates (or local legal equivalent) evidencing the Shares to be sold by each Seller, accompanied by stock powers duly executed in blank and requisite transfer tax stamps, if any, as may be necessary or desirable to effect the transactions described in SECTION 2.1;
- (ii) a receipt from each Seller for the portion of the Preliminary Purchase Price paid to such Seller;
- (iii) the Transaction Agreements to which each Seller is a party, duly executed by each relevant Seller;
- (iv) copies of the resolutions (or local equivalent) of the boards of directors of the Sellers, authorizing and approving this Agreement and the Transaction Agreements and the transactions contemplated hereby and thereby, certified by the respective corporate secretaries (or local equivalent) of the applicable Sellers to be true and complete and in full force and effect and unmodified as of the Closing Date;
- (v) certificates in form and substance satisfactory to the Buyer, duly executed and acknowledged by each Seller to the extent required under applicable Law, certifying that the transactions contemplated hereby are exempt from withholding under Section 1445 of the Code and other applicable Law;
- (vi) the Consents listed on SCHEDULE 2.5(a)(vi);
- (vii) the certificates required by SECTION 6.3(c);
- (viii) the resignations of the officers, as corporate officers, and directors of the Sold Companies and Subsidiaries set forth on SCHEDULE 2.5 (a)(viii);
- (ix) assignment of the Wickes Agreement;
- (x) a receipt duly executed by each Seller and each Sold Company and Subsidiary acknowledging, on behalf of Sellers and each of their subsidiaries, settlement of all intercompany receivables, payables, loans and investments then existing between any Seller or any of its subsidiaries that is not a Sold Company or Subsidiary, on the one hand, and the Sold Companies or Subsidiaries, on the other hand, pursuant to SECTION 5.10;
- (xi) all minute books, stock record books (or similar registries) and corporate (or similar) records and seals of each Sold Company and Subsidiary not already in the possession of the Sold Companies or the Subsidiaries; and
- (xii) a list of the signatories on each of the bank accounts set forth on SCHEDULE 3.24.

- (b) At or prior to the Closing, the Buyer shall deliver or cause to be delivered to the respective Sellers the following:
- (i) the Preliminary Purchase Price by wire transfer of immediately available funds to an account or accounts designated by Cooper as provided in SECTION 2.2(a);
- (ii) a receipt evidencing the Buyer's receipt of the Shares;
- (iii) copies of the resolutions of the board of directors (or equivalent) of the Buyer authorizing and approving this Agreement and all other transactions and agreements contemplated hereby, certified by the corporate secretary (or equivalent) of the Buyer to be true and complete and in full force and effect and unmodified as of the Closing Date;
- (iv) the Transaction Agreements to which the Buyer is a party, duly executed by the Buyer;
- (v) the certificate required by SECTION 6.2(c); and
- (vi) assumption of Cooper's obligations under the Wickes Agreement.
- 2.6 STRUCTURING COOPERATION. Notwithstanding any other provision of this Agreement to the contrary, prior to the Closing (A) the Sellers shall, and shall cause the Sold Companies and Subsidiaries to, cooperate with Buyer in connection with structuring the acquisition of the Shares (including without limitation allowing for the purchases contemplated hereunder to be effected by one or more Affiliates of the Buyer, and providing for the separate purchase of all or any portion of the equity of any Subsidiary or Venture Entity by Buyer or an Affiliate of Buyer, any such equity to be included in the definition of "Shares" for purposes of this Agreement) and shall take any and all actions necessary to effectuate such structure (including without limitation entering into separate agreements with respect to such transactions), provided that such cooperation and other actions do not have any adverse financial or Tax consequences to the Sellers or any of their Affiliates that will not be reimbursed pursuant to a mutually agreed to indemnity given by Buyer and (B) Buyer shall, and shall cause any of its Affiliates (1) to cooperate with Sellers in connection with filing Form 8832, Entity Classification Election, with respect to Cooper-Standard Automotive (Deutschland) GmbH, Cooper-Standard Automotive Ceska republika s.r.o. and Cooper-Standard Automotive Espana, S.A. (or its successor), and (2) to otherwise cooperate with Sellers in connection with structuring the acquisition of the Shares, provided that such cooperation referred to in clause (2) (A) does not have any adverse financial or Tax consequences to Buyer or any of its Affiliates as of the Closing Date that will not be reimbursed pursuant to a mutually agreed to indemnity given by Sellers and (B) such cooperation will not be expected to have any adverse structural, operational or similar consequences to Buyer, the Sold Companies or the Subsidiaries after the Closing Date, as determined in good faith by Buyer. Sellers and Buyer shall and shall cause of each of their Affiliates to file all relevant Tax Returns consistently with the structure agreed to by the parties hereto pursuant to this SECTION 2.6, unless and until otherwise required by a Taxing Authority or a court of competent jurisdiction.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE SELLERS

The Sellers, jointly and severally, hereby represent and warrant to the Buyer as follows:

3.1 ORGANIZATION. Each of the Sellers and the Sold Companies is a corporation duly incorporated, validly existing and in good standing under the Laws of its jurisdiction of incorporation.

Each of the Sellers and the Sold Companies has the requisite corporate power and authority to own, lease and operate its assets and to carry on its business as now being conducted and is duly qualified, authorized or licensed to do business and is in good standing in the jurisdictions in which the ownership, lease or operation of its assets or the conduct of its business requires such qualification, authorization or license, except where the failure to be so qualified, authorized or licensed (a) would not reasonably be expected, individually or in the aggregate, to have a material adverse effect on the ability of the Sellers to consummate the transactions contemplated by this Agreement or (b) with respect to the Sold Companies, would not reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect.

- 3.2 AUTHORIZATION; ENFORCEABILITY. Each of the Sellers has the requisite corporate power and authority to execute and deliver this Agreement and the Transaction Agreements to which each is a party and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the Transaction Agreements by each of the Sellers, as applicable, and the performance by each of them of their respective obligations hereunder and thereunder have been duly authorized by all necessary corporate action on the part of such party and no other corporate or shareholder proceedings or actions are necessary to authorize and consummate this Agreement, the Transaction Agreements or the transactions contemplated hereby or thereby. This Agreement has been, and each of the Transaction Agreements will be, when delivered to Buyer, duly executed and delivered by each of the Sellers, as applicable, and, assuming due authorization, execution and delivery by the Buyer, this Agreement constitutes, and each of the Transaction Agreements will constitute, a valid and binding agreement of each of the Sellers, as applicable, enforceable against each of them in accordance with its terms, except as may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law) (the "GENERAL ENFORCEABILITY EXCEPTIONS").
- 3.3 CAPITAL STOCK OF THE SOLD COMPANIES. SCHEDULE 3.3 sets forth for each of the Sold Companies (a) its jurisdiction of incorporation, formation or organization, as applicable, and (b) the number of authorized, issued and outstanding shares of capital stock, the names of the holders thereof, and the number of shares of capital stock held by each such holder and, except as set forth on SCHEDULE 3.3, there are no other authorized, issued or outstanding shares of capital stock or other equity interests of any of the Sold Companies. All of the issued and outstanding shares of capital stock are owned, beneficially and of record, free and clear of any Liens. All of the shares of capital stock are duly authorized and validly issued and are fully paid and nonassessable and have been issued in compliance with (and since such issuance, have not been transferred except in compliance with) all applicable federal, state and foreign securities Laws and any preemptive rights, rights of first refusal or similar rights of any Person. Except for this Agreement, there are no outstanding subscriptions, options, warrants, calls, preemptive rights, conversion or other rights, agreements, commitments, arrangements, trusts, proxies or understandings relating to the sale, issuance or voting of any shares of the capital stock of any of the Sold Companies, or of any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any shares of capital stock of any of the Sold Companies. There are no outstanding agreements or commitments obligating any Seller or Sold Company to repurchase, redeem or otherwise acquire any outstanding shares or other equity interests of any Sold Company. At the Closing, the applicable Sellers will convey good and valid title to the Shares to the Buyer, free and clear of any Liens, other than Liens created by or on behalf of the Buyer.

3.4 SUBSIDIARIES AND JOINT VENTURES.

(a) SUBSIDIARIES. SCHEDULE 3.4(a) sets forth for each of the Subsidiaries (i) its jurisdiction of incorporation, formation or organization, as applicable and (ii) the number of authorized, issued and outstanding shares of each class of its capital stock or other authorized, issued and outstanding

equity interests, as applicable, the names of the holders thereof, and the number of shares or percentage interests, as applicable, held by each such holder. Each Subsidiary is duly incorporated, formed or organized, as applicable, validly existing and, where applicable, in good standing under the Laws of its jurisdiction of incorporation, formation or organization, as applicable, has the requisite corporate or similar power and authority to own, lease and operate its assets and to carry on its business as now being conducted, and is duly qualified, authorized or licensed to do business, and, where applicable, is in good standing in the jurisdictions in which the ownership, lease or operation of its assets or the conduct of its business requires such qualification or license, except where the failure to be so qualified, authorized or licensed would not reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect. Except as set forth on SCHEDULE 3.4(a), all the issued and outstanding shares of capital stock or other equity interests of the Subsidiaries are owned, beneficially and of record, by the Sold Companies or another Subsidiary, free and clear of any Liens, and such Sold Company or Subsidiary has good and valid title to such shares of capital stock or other equity interests. All of such issued and outstanding shares were duly authorized and validly issued and are fully paid and nonassessable and have been issued in compliance with (and since such issuance, have not been transferred except in compliance with) all applicable federal, state and foreign securities Laws and any preemptive rights, rights of first refusal or similar rights of any Person. Except as set forth on SCHEDULE 3.4(a), there are no outstanding subscriptions, options, warrants, calls, preemptive rights, conversion or other rights, agreements, commitments, arrangements, trusts, proxies or understandings relating to the sale, issuance or voting of any shares of the capital stock or other equity interest of any of the Subsidiaries, or of any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any shares of capital stock or other equity interests of any of the Subsidiaries. There are no outstanding agreements or commitments obligating any Seller, Sold Company or Subsidiary to repurchase, redeem or otherwise acquire any outstanding shares or other equity interests of any Subsidiary. Except (i) for the Venture Entities and (ii) for equity interests set forth on SCHEDULE 3.4(a) that will no longer be owned by the Sold Companies and Subsidiaries after giving effect to the Restructuring, neither the Sold Companies nor the Subsidiaries own any equity interest in any other Person.

(b) JOINT VENTURES. SCHEDULE 3.4(b) sets forth for each of the Venture Entities (i) its jurisdiction of incorporation, formation or organization, as applicable and (ii) the number of authorized, issued and outstanding shares of each class of its capital stock or other authorized, issued and outstanding equity interests, as applicable, the names of the holders thereof, and the number of shares or percentage interests, as applicable, held by each such holder. Each Venture Entity is duly incorporated, formed or organized, as applicable, validly existing and, where applicable, in good standing under the Laws of its jurisdiction of incorporation, formation or organization, as applicable, has the requisite corporate, partnership or similar power and authority to own, lease and operate its assets and to carry on its business as now being conducted, and, to the Knowledge of the Sellers, is duly qualified, authorized or licensed to do business, and, where applicable, is in good standing in the jurisdictions in which the ownership, lease or operation of its assets or the conduct of its business requires such qualification or license, except where the failure to be so qualified, authorized or licensed would not reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect. The issued and outstanding shares of capital stock or other equity interests of the Venture Entities that are owned, beneficially and of record, by the Sold Companies or Subsidiaries, as applicable, are owned free and clear of any Liens, and such Sold Company or Subsidiary, as applicable, has good and valid title to such shares of capital stock or other equity interests. All of such issued and outstanding shares or other equity interests owned by the Sellers, Sold Companies or Subsidiaries were duly authorized and validly issued and are fully paid and nonassessable and have been issued in compliance with (and since such issuance, have not been transferred except in compliance with) all applicable federal, state and foreign securities Laws and any preemptive rights, rights of first refusal or similar rights of any Person. Except as set forth on SCHEDULE 3.4(b), there are no outstanding subscriptions, options, warrants, calls, preemptive rights, conversion or other rights, agreements, commitments, arrangements, trusts, proxies or understandings

relating to the sale, issuance or voting of any shares of the capital stock or other equity interest of any of the Venture Entities that are owned by the Sold Companies or Subsidiaries. There are no outstanding agreements or commitments obligating any Venture Entity to repurchase, redeem or otherwise acquire any outstanding shares or other equity interests of any Venture Entity that are owned by the Sold Companies or Subsidiaries.

- 3.5 FINANCIAL STATEMENTS. (a) SCHEDULE 3.5(a) sets forth the audited combined summary balance sheet of the Business as of December 31, 2003 and December 31, 2002, and the audited combined statements of income and retained earnings and the audited combined statements of cash flows of the Business for each of the years in the period then ended (the "ANNUAL FINANCIAL STATEMENTS"). SCHEDULE 3.5(a) also sets forth the unaudited combined summary pro forma balance sheet of the Business as of June 30, 2004 (the "BALANCE SHEET"), the related unaudited combined pro forma statement of income and retained earnings and the combined pro forma statement of cash flows for the six-month period then ended (the "INTERIM FINANCIAL STATEMENTS" and, together with the Annual Financial Statements, the "FINANCIAL STATEMENTS"). Except as set forth on SCHEDULE 3.5(a), the Financial Statements have been prepared from the books and records of the Business, and fairly present in all material respects the financial position and results of operations of the Business at the dates and for the respective periods covered, in each case in accordance with GAAP applied on a consistent basis, subject in the case of the Interim Financial Statements, to the absence of note disclosures and normal year end adjustments.
- (b) The Business has no Liabilities of any kind, whether absolute, accrued, contingent or otherwise, except for Liabilities (i) incurred in the ordinary course of business since the date of the Balance Sheet, (ii) reflected on, accrued or reserved against, in the Balance Sheet, (iii) that, if known, would not be required by GAAP to be reflected or reserved against on a balance sheet (or disclosed in the notes thereto) of the Business and the Sold Companies and the Subsidiaries, (iv) as would not reasonably be expected to, individually or in the aggregate, constitute a Sold Company Material Adverse Effect or (v) set forth on SCHEDULE 3.5(b).
- (c) Sellers have previously delivered to Buyer (i) the unaudited balance sheet of Craig Assembly as of June 30, 2004 and the unaudited statement of income of Craig Assembly for the three-month period then ended and (ii) the reviewed and unaudited balance sheet of Nishikawa for the six-month periods ended June 27, 2004 and June 29, 2003 and the related unaudited statements of income, changes in partners' equity and cash flows of Nishikawa for each of the periods then ended (the "VENTURE FINANCIAL STATEMENTS"). To the Knowledge of the Sellers, the Venture Financial Statements have been prepared from the books and records of the applicable Venture Entity, and fairly present in all material respects the financial position and results of operations of each Venture Entity at the date and for the period covered, in each case in accordance with GAAP applied on a consistent basis, subject to the absence of note disclosures and to normal year-end adjustments.
- (d) To the Knowledge of the Sellers, the Venture Entities have no Liabilities of any kind, whether absolute, accrued, contingent or otherwise, except for Liabilities (i) incurred in the ordinary course of business since the date of the applicable balance sheet included in the Venture Financial Statements, (ii) reflected on, accrued or reserve against, in such applicable balance sheet, (iii) that, if known, would not be required by GAAP to be reflected or reserved against on a balance sheet (or disclosed in the notes thereto) of the applicable Venture Entity, (iv) as would not reasonably be expected to, individually or in the aggregate, constitute a Sold Company Material Adverse Effect or (v) set forth on SCHEDULE 3.5(d).
- 3.6 NO CONFLICTS OR APPROVALS. (a) Except as set forth on SCHEDULE 3.6(a), the execution, delivery and performance by the Sellers of this Agreement and the Transaction Agreements

and the consummation by the Sellers of the transactions contemplated hereby and thereby do not and will not (i) violate, conflict with or result in a breach of the organizational documents of any of the Sellers, the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities, (ii) violate, conflict with or result in a breach of, or constitute a default by any of the Sellers, the Sold Companies, the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities (or create an event which, with notice or lapse of time or both, would constitute a default) or give rise to any right of termination, consent, cancellation, acceleration, increased Liabilities or fees, or right to increase the obligations or otherwise modify the terms under, or result in the creation of any Lien, other than Permitted Liens, upon any of the properties, rights or assets of any of the Sellers, its subsidiaries, the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities or on the Shares or to which any of such properties, assets or Shares are subject, under any Contract to which any of the foregoing is a party or otherwise bound, except as would not, individually or in the aggregate, have a Sold Company Material Adverse Effect or a material adverse effect on the ability of any of the Sellers to consummate the transactions contemplated by this Agreement, or (iii) subject to the receipt of the requisite approvals referred to on SCHEDULE 3.6(b), conflict with or violate any Governmental Order or Law applicable to any of the Sellers, the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities or any of their respective properties, rights or assets.

(b) Except as set forth on SCHEDULE 3.6(b), no Consent is required to be obtained, filed or delivered by the Sellers for the consummation by the Sellers of the transactions contemplated by this Agreement and the Transaction Agreements that if not obtained, filed or delivered, as the case may be, would reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect.

3.7 COMPLIANCE WITH LAW; GOVERNMENTAL AUTHORIZATIONS. Except as set

forth on SCHEDULE 3.7, the Sold Companies, the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities, and with respect to the Business, the Sellers and their subsidiaries, are currently in compliance with all Laws and Governmental Orders, except where noncompliance would not reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect. Except as set forth in SCHEDULE 3.7, each of the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, each Venture Entity possess all licenses, consents, approvals, permits, registrations, certificates and other governmental authorizations ("PERMITS") necessary to own, lease and operate its assets and conduct the Business as currently conducted, except, in each case, where the failure to have such Permits would not reasonably be expected to have a Sold Company Material Adverse Effect. Except as set forth in SCHEDULE 3.7, as of the date of this Agreement, the Permits are in full force and effect and the Sellers, their subsidiaries, the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities have not received any written notice from any Governmental Authority (a) asserting that any Sold Company, Subsidiary or Venture Entity is not in material compliance with any Law or Permit or (b) threatening to suspend, revoke, revise, limit, restrict or terminate any Permit held by any Sold Company, Subsidiary or Venture Entity or declare any such Permit invalid. Except as set forth in SCHEDULE 3.7, (x) the statutory records of each of the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities required to be maintained by the Laws of its jurisdiction of incorporation have been properly kept and contain an accurate and materially complete record of the applicable matters required to be contained therein and no notice or allegation that any of them is incorrect or should be rectified has been received and (y) all documents required to be filed with any relevant authority in any relevant jurisdiction in respect of the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities have been filed and were correct in all material respects.

3.8 PROCEEDINGS. Except as set forth on SCHEDULE 3.8, there are no Proceedings or, to the Knowledge of the Sellers, governmental investigations pending or, to the Knowledge of the Sellers,

threatened against any Seller, its subsidiaries, Sold Company or Subsidiary that

- (a) challenge, or question the validity of, this Agreement, any Transaction Agreement or any action taken or to be taken by any Seller, its subsidiaries, Sold Company or Subsidiary in connection with, or which seeks to enjoin or obtain monetary damages in respect of, the consummation of the transactions contemplated hereby or thereby or (b) involve more than \$1,000,000 in claims or damages individually or that seek injunctive or other equitable relief that would be materially adverse to the Business and the Sold Companies and Subsidiaries, taken as a whole. Except as set forth in SCHEDULE 3.8, to the Knowledge of the Sellers, there are no Proceedings or governmental investigations pending or threatened in writing against either of the Venture Entities.
- 3.9 ABSENCE OF CERTAIN CHANGES. Except as (a) set forth in SCHEDULE 3.9, (b) contemplated by the Restructuring or (c) otherwise expressly permitted or required by this Agreement, since December 31, 2003, (i) the Business has been conducted only in the ordinary course of business, (ii) there shall not exist any state of facts, change or event or effect that has had or would reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect and (iii) as of the date of this Agreement, the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities, and with respect to the Business, the Sellers and their subsidiaries, have not taken any action and there has not occurred any event or failure to act that if taken or had it occurred after the date hereof would constitute a violation of SECTION 5.1.
- 3.10 TAX MATTERS. Except as set forth in SCHEDULE 3.10;
- (a) All material Tax Returns required to be filed by or on behalf of the Sold Companies and Subsidiaries, and to the Knowledge of the Sellers, the Venture Entities have been timely filed (subject to permitted extensions applicable to such filing), and all such Tax Returns of the Sold Companies and Subsidiaries, and to the Knowledge of the Sellers, the Venture Entities are true, correct and complete in all material respects. All material Taxes of the Sold Companies and Subsidiaries, and to the Knowledge of the Sellers, the Venture Entities due or payable (whether or not shown on such Tax Returns) have been paid within the prescribed period or any extension thereof, other than Taxes that are being contested in good faith or reflected in the Tax Reserve. All material elections made on behalf of the Sold Companies or the Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities have been timely and properly made.
- (b) There are no material Liens relating to Taxes encumbering any of the Shares or any assets or properties of the Sold Companies or the Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities, in all cases, except for Permitted Liens.
- (c) As of the date of this Agreement, there are no Proceedings currently pending against the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, against the Venture Entities or, to the Knowledge of the Sellers, threatened, against the Sold Companies, the Subsidiaries or the Venture Entities in respect of any Tax that would result in material Taxes of the Sold Companies, Subsidiaries or the Venture Entities for any taxable period ending on or before the Closing Date, other than Taxes that are being contested in good faith or which are reflected in the Tax Reserve. To the Knowledge of the Sellers, no claims for Taxes have been made by any Taxing Authority in a jurisdiction in which any of the Sold Companies, Subsidiaries or Venture Entities does not file a Tax Return.
- (d) None of the Sold Companies or Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities has granted any extension or waiver of the statute of limitations period applicable to any material Tax or material Tax Return, or agreed to any extension of time with respect to a material Tax assessment or deficiency, which period (after giving effect to such extension or waiver) has not yet expired.

- (e) None of the Sold Companies or Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities (i) has been a member of an affiliated, consolidated, combined or unitary group as set forth in Section 1504 of the Code or any other similar provision of state, local or foreign Law other than a group the common parent of which is Cooper or (ii) has any material liability for Taxes of any Person (other than Cooper or any of its subsidiaries) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign law), as transferee or successor, by contract or otherwise.
- (f) All material amounts required to be withheld or collected for payment by the Sold Companies or Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities, including from employee salaries, wages and other compensation, have been collected or withheld and paid to the appropriate Taxing Authorities.
- (g) None of the Sold Companies or Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date
- (i) as a result of any change in method of accounting made between October 28, 1999 and on or prior to the Closing Date or to the Knowledge of Sellers, prior to October 28, 1999, (ii) pursuant to a "closing agreement" as described in
- Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax law) executed between October 28, 1999 and on or prior to the Closing Date or to the Knowledge of Sellers, prior to October 28, 1999, (iii) pursuant to a material installment sale or open transaction disposition made between October 28, 1999 and prior to the Closing or to the Knowledge of Sellers, prior to October 28, 1999, or (iv) as a result of any material prepaid amount received between October 28, 1999 and prior to the Closing or to the Knowledge of Sellers, prior to October 28, 1999. Any income, gain, deduction or loss attributable to "intercompany transactions" (as defined in Treasury Regulations under Section 1502 of the Code or any corresponding provision of state or local income Tax Law) involving the Sold Companies or any Subsidiary consummated prior to the Closing or "excess loss accounts" (as defined in Treasury Regulations under Section 1502 of the Code or any corresponding provision of state or local income Tax Law) with respect to the stock of the Sold Companies or any Subsidiary existing at the Closing will be taken into account for federal, state or local Income Tax purposes as a result of the transactions contemplated by this Agreement.
- (h) Within the last two (2) years, none of the Sold Companies or Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities has been a "distributing corporation" or a "controlled corporation" in a distribution that was purported or intended to be governed by Section 355(a) of the Code.
- (i) (A) Since 1999, no Tax Return filed with respect to the Sold Companies or Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities reflects any material adjustment pursuant to Section 482 of the Code or any analogous provision of state, local or foreign Tax law, and (B) other than with respect to the Code Section 482 type adjustments which have been questioned in connection with intercompany transactions between Standard and CSA Canada for years June 30, 1995 December 31, 2001, prior to 1999, to the Knowledge of Sellers, no Tax Return with respect to the Sold Companies or Subsidiaries or the Venture Entities reflects any material adjustment pursuant to Section 482 of the Code or any analogous provision of state, local or foreign Tax law.
- (j) None of the Sold Companies or Subsidiaries, or to the Knowledge of the Sellers, the Venture Entities has taken any deduction or received any Tax benefit arising from participation in a "tax shelter" as defined in Section 6111(c) of the Code or participated in a "reportable transaction" as defined in Treasury Regulation Sections 1.6011-4(a), (c)(3), 1.6011-4T(a), (b), or any analogous provision of state, local or foreign Tax law.

- 3.11 EMPLOYEE BENEFITS. SCHEDULE 3.11(a)(i) 1 sets forth a true and complete list of all "employee benefit plans" (within the meaning of Section 3(3) of ERISA, including, without limitation, multiemployer plans within the meaning of Section 3(37) of ERISA), all severance, termination, salary continuation, change in control, retention, parachute, employment, incentive, bonus, stock option, stock purchase, restricted stock, retirement, pension, redundancy, profit sharing, deferred compensation, employee loan, retiree welfare, fringe benefit and all other employee benefit plans, programs, agreements, policies or arrangements, whether or not subject to ERISA, whether formal or informal (each a "PLAN"), (i) under which any Company Employee has any present or future right to benefits and which are contributed to, entered into, sponsored by or maintained by the Sellers, Sold Companies or Subsidiaries, or any member of their respective Controlled Group (as defined below), (ii) under which the Sold Companies or Subsidiaries have had or have any present or future liability or (iii) which is a Venture Entity Plan (each, a "COMPANY BENEFIT PLAN"). "COMPANY EMPLOYEE" means any current or former employee, director or officer of the Sold Companies or Subsidiaries or any current employee of the Sellers who is or was employed primarily in the Business and is listed on SCHEDULE 3.11(a)(i)-2; provided, that, except as set forth on SCHEDULE 3.11(a)(i) 2, Company Employee shall not include any Discontinued Business Employee. "DISCONTINUED BUSINESS EMPLOYEE" means any present or former employee of the Seller or its Affiliates who was employed primarily in the Business, or any present or former employee of the Sold Companies or Subsidiaries, in each case who performs or performed services for (A) a plant or facility that has been divested, shut down or closed, (B) a plant or facility whose divestiture, shut-down or closure has been announced at any time prior to the Closing Date or
- (C) any other plant or facility not identified on SCHEDULES 3.18(a) or 3.18(b). Notwithstanding the foregoing, any Company Benefit Plans under which the Sold Companies or Subsidiaries have had liability, but do not have present or future liability, shall not be included on SCHEDULE 3.11(a)(i)-1.
- (ii) SCHEDULE 3.11(a)(i)-1 separately identifies each Company Benefit Plan that is maintained in the United States (each, a "COMPANY U.S. BENEFIT PLAN"), and each Company Benefit Plan that is not a Company U.S. Benefit Plan (each, a "COMPANY FOREIGN BENEFIT PLAN").
- (iii) SCHEDULE 3.11(a)(i)-1 separately identifies each Company Benefit Plan that is sponsored, maintained or entered into solely by the Sold Companies or Subsidiaries (and not by Sellers or its other Affiliates) (each, a "SOLD COMPANY BENEFIT PLAN"), and each Company Benefit Plan that is not a Sold Company Benefit Plan (each, a "SELLER BENEFIT PLAN").

Notwithstanding any provision of this Agreement, all Seller representations with respect to Venture Entity Plans shall be deemed to be limited to the Knowledge of the Sellers, and SECTION 3.11(a) shall apply only to Venture Entity Plans that are defined benefit plans (as defined in Section 3(35) of ERISA), multiemployer plans (as defined in Section 4001(a)(3) of ERISA and welfare benefit plans (as defined in Section 3 (1) of ERISA) that provide post-retirement, medical or life insurance benefits.

(b) Current, accurate and complete copies (or, as to clause (i) of this sentence, to the extent no copy exists, an accurate description) of the following materials have been delivered to the Buyer with respect to each Company U.S. Benefit Plan, to the extent applicable: (i) current plan documents, any amendments and any related trust agreement, group annuity contract or other funding instrument, (ii) except as set forth in SCHEDULE 3.11(b), the most recent determination letter from the Internal Revenue Service ("IRS"), (iii) the most recent summary plan description and summary of material modifications to the extent not included in the summary plan description in each case distributed to employees, and (iv) for the three (3) most recent years (A) the Form 5500 and attached schedules, (B) audited financial statements and (C) actuarial valuation reports.

- (c) Except as set forth in SCHEDULE 3.11(c) or as would not have a Sold Company Material Adverse Effect: (i) the Company U.S. Benefit Plans have been established and administered in compliance with their terms and the applicable requirements of ERISA, the Code, and other applicable Laws; (ii) each Company U.S. Benefit Plan and related trust that is intended to be qualified within the meaning of Section 401 or 501, as applicable, of the Code is so qualified and has received or has timely applied for a favorable determination letter as to its qualification, and nothing has occurred, whether by action or failure to act, that could reasonably be expected to cause the loss of such qualification or other losses arising from such action or failure to act; (iii) no event has occurred and no condition exists with respect to any Company U.S. Benefit Plan that would subject the Sold Companies or Subsidiaries, either directly or by reason of their affiliation with any member of their "CONTROLLED GROUP" (defined as any organization which is a member of a controlled group of organizations within the meaning of Sections 414(b), (c) (m) or (o) of the Code, or which would be considered to be a single employer with that entity pursuant to Section 4001(b) of ERISA), to any material tax, fine, lien, penalty or other liability imposed by ERISA, the Code or other applicable Laws; (iv) no "reportable event" (as such term is defined in Section 4043 of the Code), non-exempt "prohibited transaction" (as such term is defined in Section 406 of ERISA and Section 4975 of the Code), or "accumulated funding deficiency" (as such term is defined in Section 302 of ERISA and Section 412 of the Code (whether or not waived)) has occurred with respect to any Company U.S. Benefit Plan; (v) all premiums due to the Pension Benefit Guaranty Corporation ("PBGC") with respect to any Company U.S. Benefit Plan have been timely paid in full; (vi) the PBGC has not instituted proceedings to terminate any Company U.S. Benefit Plan; and (vii) no Liability (other
- (d) No insurance policy funding any Company U.S. Benefit Plan provides for a retroactive rate adjustment or loss sharing arrangement that would result in material liability to the Buyer, the Sold Companies or the Subsidiaries.

the PBGC) under Title IV of ERISA has been or could reasonably be expected to be incurred by any of the Sold Companies or Subsidiaries.

- (e) No Company U.S. Benefit Plan is a "multiemployer plan" (as defined in Section 4001(a)(3) of ERISA), and neither the Sold Companies nor Subsidiaries, nor any member of their Controlled Group, has any unpaid liability or obligation in respect of any multiemployer plan.
- (f) With respect to any Company U.S. Benefit Plan, except as set forth in SCHEDULE 3.11(f) or, with respect to clauses (i), (ii) and (iv), as would not have a Sold Company Material Adverse Effect: (i) there are no pending or, to the Knowledge of the Sellers, threatened Proceedings, other than routine claims for benefits in the ordinary course by participants and beneficiaries;
- (ii) to the Knowledge of the Sellers, no facts or circumstances exist that could give rise to any such Proceedings; (iii) no written or oral communication has been received from the PBGC in respect of any Company U.S. Benefit Plan subject to Title IV of ERISA concerning the funded status of any such plan or any transfer of assets and liabilities from any such plan in connection with the transactions contemplated herein; and (iv) no administrative investigation, audit or other Proceeding by the Department of Labor, the PBGC, the IRS or other Governmental Authorities is pending or in progress or, to the Knowledge of the Sellers, threatened.
- (g) Except as set forth in SCHEDULE 3.11(g), no Company U.S. Benefit Plan exists that, as a result of the execution of this Agreement, shareholder approval of this Agreement or the transactions contemplated by this Agreement (whether alone or in connection with any other events), would (i) entitle any Company Employee to severance pay or any increase in severance pay upon any termination of employment after the date of this Agreement; (ii) with respect to any Company Employee, accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any of the Company Benefit Plans; (iii) limit or restrict the right of the Sold Companies or Subsidiaries to merge, amend or terminate any of the Company U.S. Benefit Plans; or (iv)

cause the Sold Companies or Subsidiaries to record additional compensation expense on their income statements with respect to any outstanding stock option or other equity-based award.

- (h) Except as set forth in SCHEDULE 3.11(h), there is no Contract, plan or arrangement (written or otherwise) covering any Company Employee that, individually or collectively, could give rise to the payment of any amount that would be nondeductible pursuant to Section 280G of the Code or could give rise to the payment of any amount in respect of Code Section 4999, nor will any of the transactions contemplated by this Agreement result in any amounts that are not deductible pursuant to Section 280G of the Code or the payment of any amount in respect of Code Section 4999.
- (i) Except as would not have a Sold Company Material Adverse Effect, with respect to individuals described in this SECTION 3.11(i) who are located in the United States, (i) all persons classified as independent contractors of the Sold Companies or Subsidiaries satisfy and have at all times satisfied in all material respects the requirements of applicable Law to be so classified; (ii) the Sellers, Sold Companies and Subsidiaries have fully and accurately reported such persons' compensation on IRS Form 1099 when required to do so; (iii) none of the Sellers, the Sold Companies nor any Subsidiary has or has had any obligations to provide benefits with respect to such persons under any Company Benefit Plan or otherwise and (iv) none of the Sold Companies or Subsidiaries has employed or employs any "leased employees" as defined in Section 414 (n) of the Code.
- (j) (i) With respect to each Company Foreign Benefit Plan which is not a Government Scheme, to the extent applicable, current, accurate and complete copies of (A) current plan documents, any amendments and any related trust agreement, group annuity contract or other funding instrument (or, to the extent no copy exists, an accurate description), and (B) the most recent audited financial statements and/or actuarial valuation reports, have been delivered to the Buyer, except with respect to such Plans which, individually or in the aggregate, do not create Liabilities which are material to the Sold Companies as a whole; provided that any such documentation which has not been delivered to the Buyer as of the date hereof pursuant to the immediately preceding exception shall be delivered to the Buyer prior to the Closing Date;
- (ii) with respect to each Company Foreign Benefit Plan, except as set forth in SCHEDULE 3.11(j), all data and information required, in the case of such Plans that are not Government Schemes, to administer or , in the case of such Plans that are Government Schemes, to comply with applicable rules and regulations, is in the possession or control of the Sold Companies or the Subsidiaries and is complete and correct in all material respects;
- (iii) with respect to each Company Foreign Benefit Plan, except as set forth in SCHEDULE 3.11(j) or, with respect to clauses (A), (B), and (D), as would not have, individually or in the aggregate, a Sold Company Material Adverse Effect: (A) there are no pending or, to the Knowledge of the Sellers, threatened Proceedings, other than routine claims for benefits in the ordinary course by participants and beneficiaries; (B) to the Knowledge of the Sellers, no facts or circumstances exist that could give rise to any such Proceedings; (C) no written or oral communication has been received from any Governmental Authority in respect of any Company Foreign Benefit Plan concerning the funded status of any such plan or any transfer of assets and liabilities from any such plan in connection with the transactions contemplated herein; and (D) no administrative investigation, audit or other Proceeding by any Governmental Authority is pending or in progress or, to the Knowledge of the Sellers, threatened;
- (iv) with respect to each Company Foreign Benefit Plan, except as set forth in SCHEDULE 3.11(j) or as would not have, individually or in the aggregate, a Sold Company Material Adverse Effect: (A) each such plan that is intended to be tax qualified or tax registered (or in the

case of such plans established in the UK, exempt approved under Chapter 1, Part XIV of the UK Income and Corporation Taxes Act 1988) is so qualified or registered or exempt approved, and nothing has occurred, whether by action or failure to act, that could reasonably be expected to cause the loss of such qualification or registration or exemption or approval (and in the case of such plans established in Canada, the assets of such plans have been invested so as to avoid exposure to penalty taxes under the Income Tax Act (Canada)); (B) all Company Foreign Benefit Plans that are required to be funded are funded in accordance with applicable Law and other requirements (and in the case of such plans established in the UK in accordance with the advice of the actuary appointed to the plan) (and in the case of such plans established in Canada were fully funded on a going concern and on a solvency basis determined as of the date of the most recently filed actuarial valuation reports for such plans and in accordance with the methods and assumptions set forth therein); (C) for all Company Foreign Benefit Plans that are not required to be funded, reserves in amounts required by Law or applicable accounting standards exist in the accounts of the relevant obliged entities, calculated in compliance with actuarial methods and the Law applicable in the relevant jurisdictions; (D) all current obligations under any Company Foreign Benefit Plan have been met, and all contributions to and, with respect to Company Foreign Benefit Plans that are not Government Schemes, payments from any Company Foreign Benefit Plan in respect of any Company Employee that are required in accordance with the terms of such Company Foreign Benefit Plan or applicable Law have been timely made, or if not yet due, have been properly reflected in the Financial Statements; (E) where applicable, all contributions or premiums to any governmental or non governmental reinsurance scheme backing up any Company Foreign Benefit Plan in case of insolvency of the relevant obligor have been paid, and nothing has occurred, whether by action or failure to act, that could reasonably be expected to cause the loss of the coverage under such reinsurance scheme; (F) no Company Foreign Benefit Plans (other than Government Schemes which have been described generically) or similar schemes or promises, whether with respect to groups of beneficiaries or individual beneficiaries, exist with respect to or obliging any of the Sold Companies or Subsidiaries that have not been disclosed on SCHEDULE 3.11(j) or expressly or reasonably identifiable and adequately provided for on the financial statements of such companies; (G) all Company Foreign Benefit Plans (and their corresponding assets or funds) have been implemented, administered and invested in compliance with their terms, the Law and all applicable collective bargaining agreements, and all reports, returns or similar documents required to be filed with a Governmental Authority in respect of the Foreign Company Benefit Plans have been duly and timely filed; (H) any and all one-time lump sum payments to be paid under French law, applicable bargaining agreements or individual agreements to employees of the Sold Companies or Subsidiaries upon reaching retirement age (indemnites de depart) are either appropriately reflected in the so-called annexe to the relevant Financial Statements in compliance with Art. L. 123-13 of the French Code de Commerce or have been reserved for in the relevant Financial Statements; (I) with respect to all Foreign Company Benefit Plans in Canada, no Governmental Authority has the power to order a wind up (in whole or in part) of the plan in respect of events prior to the Closing Date, no such plans have received a transfer of assets from or been merged with the assets of another plan, and no conditions have been imposed by a Governmental Authority and no undertakings or commitments have been given in writing to any Governmental Authority, union or Company Employee concerning the use or recognition of actual or actuarial surplus assets relating to any Company Foreign Benefit Plan or any related funding medium; and (J) the Sold Companies' and Subsidiaries' essential obligation to or in respect of any Foreign Company Benefit Plan that is a multiple employer plan is to make contributions to such plan in the amounts and in the manner described in the relevant collective bargaining agreement;

(v) Except as set forth in SCHEDULE 3.11(j), no Company Foreign Benefit Plan exists that, as a result of the execution of this Agreement, shareholder approval of this Agreement or the

transactions contemplated by this Agreement (whether alone or in connection with any other events), would (A) entitle any Company Employee to severance pay or any increase in severance pay upon any termination of employment after the date of this Agreement; (B) with respect to any Company Employee, accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any of the Company Foreign Benefit Plans; (C) limit or restrict the right of the Sold Companies or Subsidiaries to merge, amend or terminate any of the Company Foreign Benefit Plans; or (D) cause the Sold Companies or Subsidiaries to record additional compensation expense on their income statements with respect to any outstanding stock option or other equity-based award.

- (k) (i) Except as would not have a Sold Company Material Adverse Effect, all contributions (including, without limitation, all employer matching or other contributions and employee salary reduction contributions) to and payments from any Company U.S. Benefit Plan in respect of any Company Employees that are required in accordance with the terms of such Company U.S. Benefit Plan, any related document, the Code or ERISA have been timely made, or, if not yet due, have been properly reflected in the Financial Statements; and (ii) all such contributions to, and payments from, any Company U.S. Benefit Plan, except those to be made from a trust qualified under Section 401(a) of the Code, that are required to be made as of the Closing Date will be made on or prior to the Closing Date.
- (1) None of the Company U.S. Benefit Plans is a multiple employer plan described in Section 413(c) of the Code.
- 3.12 LABOR RELATIONS. (a) Except as set forth in SCHEDULE 3.12 or as otherwise permitted pursuant to this Agreement, (i) none of the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities is a party to any collective bargaining agreement nor is any such contract or agreement presently being negotiated, (ii) none of the Sold Companies or Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities is in material breach of any collective bargaining agreement, (iii) within the past three (3) years, there has been no labor strike, work stoppage, slowdown, lockout or other labor controversy in effect with respect to the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities, or, to the Knowledge of the Sellers, threatened against the Sold Companies or the Subsidiaries or the Venture Entities, (iv) there are no material grievances or other material labor disputes or proceedings pending against the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities or, to the Knowledge of the Sellers, threatened against the Sold Companies or the Subsidiaries or involving any Company Employees or the Venture Entities, (v) there are no material unfair labor practice charges, grievances or complaints, actions, inquiries, proceedings or, to the Knowledge of the Sellers, investigations pending against the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities or, to the Knowledge of the Sellers, threatened against the Sold Companies or the Subsidiaries or the Venture Entities by or on behalf of any Company Employees and (vi) the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities are in compliance with their obligations pursuant to the Worker Adjustment and Retraining Notification Act of 1988 ("WARN ACT"), and all other notification and bargaining obligations arising under any collective bargaining agreement, statute or otherwise. As of the date of this Agreement, to the Knowledge of the Sellers, there is no union campaign being conducted to solicit cards from the Company Employees to authorize a union to request a National Labor Relations Board ("NLRB") certifications election or to seek a certification pursuant to the Ontario Labour Relations Act with respect to the Company Employees or any employee of either Venture Entity.
- (b) Except as set forth on SCHEDULE 3.12, the Sold Companies and Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities and, with respect to the Business, the Sellers and their subsidiaries are in material compliance with all applicable Laws, Governmental Orders, agreements,

contracts and policies relating to the employment of their respective employees, including all such Laws and Governmental Orders relating to wages, hours, collective bargaining, compensation, overtime (including but not limited to compensation for such overtime), benefits, terms and conditions of employment, termination or employment discrimination, immigration, disability, civil rights, occupational safety and health, workers' compensation, pay equity and the collection and payment of withholding and/or social contribution taxes and similar Taxes, except where noncompliance would not reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect.

- (c) Except as set forth on SCHEDULE 3.12. no Sold Company or Subsidiary or, to the Knowledge of the Sellers, the Venture Entities has, and, with respect to the Business, no Seller or any of its subsidiaries has, closed any plant or facility, effectuated any layoffs of employees or implemented any early retirement, separation or window program within the past three (3) years, nor has any Seller or any of its subsidiaries with respect to the Business or any Sold Company or Subsidiary or, to the Knowledge of the Sellers, the Venture Entities planned or announced any such action or program for the future.
- (d) Except as set forth on SCHEDULE 3.12, no Sold Company or Subsidiary or, to the Knowledge of the Sellers, the Venture Entities is, and, with respect to the Business, no Seller or any of its subsidiaries is, a party to or otherwise bound by, any consent decree with, or citation by, any Governmental Authority related to employees or employment practices.
- 3.13 INTELLECTUAL PROPERTY. SCHEDULE 3.13(a) sets forth a complete list of material Intellectual Property registrations and applications owned by the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities and any actions or fees due on or before January 31, 2005 to maintain same. Except (a) as set forth in SCHEDULE 3.13(b), (b) for the agreements listed on SCHEDULE 3.14(a) or (c) with respect to the use of corporate names (including the Trademarks) in accordance with SECTION 5.12, (i) one or more of the Sold Companies or the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities owns or has the right to use all material Intellectual Property necessary for the conduct of the Business as it is currently conducted (including the Trademarks), free from (A) any Liens other than Permitted Liens, and (B) any present or future requirement of royalty payments or other license fees or payments, (ii)(A) the operation of the Business does not infringe or otherwise materially violate any Intellectual Property of any other Person, (B) none of the Sold Companies or any of the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities, and with respect to the Business, the Sellers and their subsidiaries, have received written notice or, to the Knowledge of the Sellers, oral notice, from, or exchanged any correspondence with, any Person that challenges (or relates to the challenge of) the use or ownership of the Intellectual Property or the validity or enforceability thereof, except as would not reasonably be expected to have a Sold Company Material Adverse Effect, and (C) none of the Intellectual Property owned by the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities is being materially infringed by any other Person,
- (iii) after the Restructuring has been completed, the material Intellectual Property purportedly owned by the Sold Companies or the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities, will be owned solely by same, valid and enforceable and free of adverse ownership claims or interests of the Sellers or third parties, including, without limitation, current or former employees or contractors, subject to payment obligations to inventors required by applicable Law and (iv) there is no material Intellectual Property that is shared by one or more of the Sold Companies, the Subsidiaries or the Venture Entities on the one hand, and the Sellers or any of their Affiliates (excluding the Sold Companies, Subsidiaries and Venture Entities) on the other hand.
- 3.14 CONTRACTS. (a) SCHEDULE 3.14(a) sets forth a true, complete and correct list of each of the following contracts to which any of the Sold Companies or the Subsidiaries is a party or by

which any of them is bound as of the date of this Agreement, other than Company Benefit Plans (collectively, the "MATERIAL CONTRACTS"):

- (i) Contracts involving the expenditure by the Sold Companies or the Subsidiaries of more than \$1,000,000 in any instance for the purchase of materials, supplies, equipment or services, excluding any such contracts that are terminable by the Sold Companies or the Subsidiaries without penalty on not more than ninety (90) days notice;
- (ii) (A) indentures, mortgages, loan agreements, capital leases, security agreements, or other Contracts relating to Debt Obligations or (B) any Contract or other currently outstanding instrument under which any of the Sold Companies or the Subsidiaries has, directly or indirectly, made any advance, loan, extension of credit (other than an account receivable) or capital contribution to, or other investment in, any Person;
- (iii) Contracts that restrict the Sold Companies or the Subsidiaries or any of their Affiliates after the Closing Date from engaging in any line of business in any geographic area or competing with any Person;
- (iv) Contracts that restrict the declaration, set aside or payment of any dividends or distributions on, or in respect of, any capital stock or equity interest of any Sold Company or Subsidiary;
- (v) Contracts to acquire or sell goods with respect to the customers set forth on SCHEDULE 3.19;
- (vi) any option, other agreement or right to purchase or otherwise acquire or sell or otherwise dispose of any interest in real property involving payments by any of the Sold Companies or the Subsidiaries in excess of \$1,000,000;
- (vii) any commitment to make any capital expenditure or to purchase a capital asset not contemplated by the Capital Expenditure Budget;
- (viii) except as to the extent contemplated by the Capital Expenditure Budget, any commitment for the purchase or sale of any of its assets, other than in the ordinary course of business, or any capital stock of the Sold Companies or the Subsidiaries;
- (ix) any lease or similar agreement under which (A) any of the Sold Companies or the Subsidiaries is the lessee of, or holds or uses, any facility, machinery, equipment, vehicle or other tangible personal property owned by any third Person for an annual rent in excess of \$500,000 or (B) any of the Sold Companies or the Subsidiaries is the lessor of, or makes available for use by any third Person, any tangible personal property owned by any of the Sold Companies or the Subsidiaries for an annual rent in excess of \$500,000;
- (x) Contracts (i) entered into or assumed by any of the Sold Companies or Subsidiaries in which it has an obligation in respect of providing for indemnification or purchase price adjustment, in connection any disposition, sale or other transfer or any present or former business or commercial activity and (ii) which was either (A) entered into after January 1, 1999 or (B) pursuant to which there are any outstanding, unresolved or potential indemnification claims in excess of \$500,000 against any of the Sold Companies or Subsidiaries;

- (xi) Contracts under which any of the Sold Companies or the Subsidiaries has licensed material Intellectual Property to or from any other Person or that otherwise relate primarily to material Intellectual Property, excluding non-exclusive, commercially available software licenses entered into in the ordinary course of business for annual payments of less than \$250,000; provided that, for purposes of the Sellers' disclosure obligations under SECTIONS 3.14(a) and (b), Sellers shall provide only a representative sample of any of the foregoing Contracts that were entered into in the ordinary course of business with consultants and employees;
- (xii) partnership, limited liability company, joint venture agreements or other Contracts involving a sharing of profits or expenses by the Sold Companies or Subsidiaries; and
- (xiii) any other Contracts not otherwise described in clauses (i) through (xii) above to which any of the Sold Companies or the Subsidiaries is a party or is otherwise bound which is reasonably expected to result in an annual expenditure by any of the Sold Companies or the Subsidiaries after the Closing Date of more than \$1,000,000 for any such individual Contract.
- (b) True, correct and complete copies of each of the Material Contracts that are in writing have been made available to the Buyer and, if such Material Contract is not in writing, written summaries thereof have been made available to the Buyer, including in each case amendments, waivers or other changes thereto.
- (c) Each Material Contract is in full force and effect, and is a valid and binding agreement of the applicable Sold Company or Subsidiary enforceable by or against such applicable Sold Company or Subsidiary or, with respect to the Business, the applicable Seller or its subsidiary, in accordance with its terms subject to the General Enforceability Exceptions. The Sold Companies or the Subsidiaries have performed in all material respects its obligations required to be performed by them to date under the Material Contracts. There is no default or breach (or allegation of same) by the Sold Companies or the Subsidiaries or, to the Knowledge of the Sellers, any other party, in the timely performance of any material obligation to be performed or paid under any Material Contract.

3.15 ENVIRONMENTAL MATTERS. Except as set forth on SCHEDULE 3.15:

- (a) each of the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities is in compliance with all applicable Environmental Laws (including all Permits required under such Environmental Laws), except for such noncompliance that would not have, individually or in the aggregate, a Sold Company Material Adverse Effect;
- (b) none of the Sellers, the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities, has received any unresolved Environmental Claim or, to the Knowledge of the Sellers, notice of any threatened Environmental Claim regarding any Sold Company, Subsidiary, Venture Entity, the Business or the Real Property;
- (c) none of the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities has Released any Hazardous Materials at or in the vicinity of the Real Property that requires reporting, investigation, assessment, cleanup, remediation or other type of response action by any Sold Company, Subsidiary or, to the Knowledge of the Sellers, the Venture Entities pursuant to any Environmental Law or that could otherwise reasonably be expected to result in material liability under or relating to any Environmental Law, and Hazardous Materials are not otherwise present at or about any of the Real Property, or any other facility currently or formerly owned, leased or operated

by any of the Sold Companies, the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities, in amount or condition that could reasonably be expected to result in material liability to any Sold Company, Subsidiary or, to the Knowledge of the Sellers, the Venture Entities under or relating to any Environmental Law;

- (d) to the Knowledge of the Sellers, no material costs, including capital expenditures that are inconsistent with prior expenditures made in relation to the Sold Companies, the Subsidiaries and the Business and that are not reflected in the Capital Expenditure Budget will be required in order for the Sold Companies, the Subsidiaries and the Business to comply during the one-year period following the Closing Date with all existing applicable Environmental Laws existing as of the Closing Date (including Permits required pursuant to existing Environmental Law);
- (e) except as would not reasonably be expected to result in material liability under any Environmental Law, to the Knowledge of the Sellers in the case of the Venture Entities, there are no (i) underground or aboveground storage tanks, (ii) asbestos, (iii) polychlorinated biphenyls, or (iv) urea-formaldehyde insulation present at or about any of the Real Property in a condition that currently requires reporting, investigation, assessment, cleanup, remediation or other type of response action by any Sold Company, Subsidiary or, to the Knowledge of the Sellers, the Venture Entities pursuant to any Environmental Law, or that would otherwise result in liability under any Environmental Laws;
- (f) Hazardous Materials have not been Released, transported, disposed of or arranged to be disposed of by, or on behalf of, any of the Sold Companies, the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities to any location, in material violation of, or in a manner that would reasonably be expected to result in material liability under or relating to, any Environmental Law; and
- (g) none of the Sold Companies and the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities has expressly assumed or retained by contract any material liabilities or material obligations of other Persons under any Environmental Laws.
- 3.16 INSURANCE. (a) SCHEDULE 3.16(a) contains a complete and correct list of all insurance policies held in the names of the Sold Companies or the Subsidiaries and all other insurance arrangements or contracts for the transfer or sharing of insurance risks covering the assets, businesses, operations, Company Employees, officers or directors of the Sold Companies and the Subsidiaries (the "INSURANCE POLICIES") as of the date hereof, specifying the insurer, amount of coverage and type of insurance. All such policies are in full force and effect and were in full force and effect during the periods of time that such insurance policies purported to be in effect, are valid, enforceable, existing and binding and all premiums due thereon have been timely paid (subject to changes made in the ordinary course of business that will not materially reduce the coverage thereunder) and will remain in full force and effect until the Closing (subject to any renewals thereof), at which time, coverage thereunder will be discontinued with respect to the Sold Companies, the Subsidiaries and the Business.
- (b) The Insurance Policies are sufficient for compliance with applicable Law and all contracts to which any of the Sold Companies or the Subsidiaries or, with respect to the Business, any of the Sellers or any of their subsidiaries, is a party or by which they or any of their respective assets or properties are bound. All appropriate insurers under the Insurance Policies have been notified of all potentially insurable losses and pending litigation and legal matters, and no such insurer has informed any Seller or any of its subsidiaries or any Sold Company or any of the Subsidiaries of any denial of coverage or reservation of rights thereto.

- (c) Except as set forth in SCHEDULE 3.16(c), since the date of the Balance Sheet no event has occurred which, with notice or lapse of time or both, would constitute a material breach or default or give rise to any right of termination or cancellation or otherwise modify the terms under any Insurance Policy.
- 3.17 PERSONAL PROPERTY ASSETS. (a) Except as set forth on SCHEDULE 3.17, the Sold Companies, the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities have good, valid and marketable title to, or hold by valid and existing lease or license, all the tangible personal property assets reflected as assets on the Balance Sheet or in the balance sheets included in the Venture Financial Statements, as applicable, or acquired after the date of the Balance Sheet or such balance sheets, except with respect to assets disposed of in the ordinary course of business since such date, free and clear of all Liens, except for Permitted Liens.
- (b) After giving effect to the consummation of the Restructuring, the Sold Companies, the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities, will own, or have valid leasehold interests in, all the assets necessary for the conduct of the Business, as currently conducted.
- 3.18 REAL PROPERTY. (a) Leased Properties. SCHEDULE 3.18(a) lists all real property leased or subleased by any of the Sold Companies, the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities (the "LEASED REAL PROPERTY"). Except as set forth on SCHEDULE 3.18(a), the Sellers have delivered to the Buyer true and complete copies of the leases and subleases covering the Leased Real Property leased by any Sold Company or Subsidiary. With respect to each such lease and sublease and except as otherwise specified on SCHEDULE 3.18(a):
- (i) such lease or sublease is, to the Knowledge of the Sellers, in full force and effect in all respects and enforceable in accordance with its terms, subject to the General Enforceability Exceptions; and
- (ii) none of the Sold Companies or the Subsidiaries is in material default under any such lease or sublease and, to the Knowledge of the Sellers, no other party to any such lease or sublease is in material default thereunder.
- (b) OWNED PROPERTIES. SCHEDULE 3.18(b) lists all real property owned by any of the Sold Companies, the Subsidiaries or, to the Knowledge of the Sellers, the Venture Entities (the "OWNED REAL PROPERTY"). With respect to each such parcel of the Owned Real Property and except as otherwise specified on SCHEDULE 3.18(b):
- (i) the identified owner has, to the Knowledge of the Sellers in the case of the Venture Entities, fee simple title to the parcel of Owned Real Property, free and clear of any Liens, except for Permitted Liens; and
- (ii) there are no pending or, to the Knowledge of the Sellers, threatened condemnation Proceedings with respect to the Owned Real Property of the Sold Companies or Subsidiaries and, to the Knowledge of the Sellers, there are no pending or threatened condemnation Proceedings with respect to the Owned Real Property of the Venture Entities.
- 3.19 CUSTOMERS AND SUPPLIERS. SCHEDULE 3.19 sets forth a true, correct and complete list of (a) the ten (10) largest customers of the Business in terms of sales during the twelve-month period ended June 30, 2004 and (b) the ten (10) largest suppliers of the Business in terms of purchases during the

twelve-month period ended June 30, 2004. None of the Sellers or their subsidiaries or any Sold Company or Subsidiary has received written notice from any customer or supplier listed on SCHEDULE 3.19 that such customer or supplier intends to terminate or materially adversely modify its relationship or materially reduce the volume of business that it does with any Sold Company or Subsidiary. Except as set forth on SCHEDULE 3.19, the Sold Companies and Subsidiaries, and with respect to the Business, the Sellers and their subsidiaries have not purchased, from any single supplier, goods or services for which the aggregate purchase price exceeds five percent (5%) of the total amount of goods and services purchased for the Business during its most recent full fiscal year.

- 3.20 PRODUCT LIABILITY. Except for matters which would not reasonably be expected, individually or in the aggregate, to have a Sold Company Material Adverse Effect and except as set forth on SCHEDULE 3.20, no Sold Company, any Subsidiary or, to the Knowledge of the Sellers, any Venture Entity has, and, with respect to the Business, none of the Sellers or their subsidiaries have, received any written notice, or, to the Knowledge of Sellers, any oral notice, relating to any claim involving use of or exposure to any of the products (or any part or component) designed, manufactured, serviced or sold, or services performed, by the Business or Sold Company or any Subsidiary or any Venture Entity, including for negligence, strict liability, design or manufacturing defect, conspiracy, failure to warn, or breach of express or implied warranties or merchantability or fitness for any purpose or use, or from any alleged breach of implied warranties or representations, or any alleged noncompliance with any applicable Laws pertaining to products liability matters, including, without limitation, relating to the presence or alleged presence of asbestos or asbestos containing materials in such products.
- 3.21 AFFILIATE TRANSACTIONS. SCHEDULE 3.21 lists all existing contracts or other arrangements or transactions between any of the Sold Companies, the Subsidiaries and the Venture Entities, on the one hand, and any of (a) the Sellers and its Affiliates (other than the Sold Companies, the Subsidiaries and the Venture Entities) or (b) the directors, officers or employees (or any immediate family member thereof) of any Seller and its Affiliates (except those of a type available to employees generally), on the other hand (collectively, the "AFFILIATE CONTRACTS"). All Affiliate Contracts are at least as favorable to the Seller and its Affiliates as would be available with independent third parties dealing at arm's length. Except as set forth on SCHEDULE 3.21, all Affiliate Contracts shall be terminated and Buyer, the Sold Companies, the Subsidiaries and the Venture Entities shall not have any obligations thereunder.
- 3.22 ACCOUNTS. All accounts and notes receivable of the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities are valid and genuine and have arisen solely out of bona fide sales and deliveries of goods, performances of services and other business transactions in the ordinary course of business, and the accounts receivable reserve reflected in the Balance Sheet and, to the Knowledge of the Sellers, the balance sheets included in the Venture Financial Statements are, as of the date of the dates thereof, adequate and established in accordance with GAAP, subject to year-end adjustments and accruals in the ordinary course of business and not material in amount. Since the date of the Balance Sheet and, to the Knowledge of the Sellers, the dates of the balance sheets included in the Venture Financial Statements, there has been no event or occurrence that, when considered, individually or together with all such other events or occurrences, would cause such accounts receivable reserves to be inadequate. Since the date of the Balance Sheet and, to the Knowledge of the Sellers, the dates of the balance sheets included in the Venture Financial Statements, none of the Sold Companies, the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities or, with respect to the Business, the Sellers or their subsidiaries has, with respect to any non-de minimis portion of its trade accounts payable, (a) failed to pay its trade accounts payable in the ordinary course or (b) extended the terms of payment, whether by contract, amendment, act, deed, or course of dealing, of any trade account payable.

- 3.23 INVENTORY. The inventories set forth in the Balance Sheet and, to the Knowledge of the Sellers, the balance sheets included in the Venture Financial Statements were properly stated therein at cost, which is not in excess of market value and determined in accordance with GAAP consistently maintained and applied by the Sold Companies and the Subsidiaries and the Venture Entities. Inventory costs have been determined by the last-in, first-out ("LIFO") method for substantially all domestic inventories. Costs of other inventories have been determined principally by the first-in, first-out ("FIFO") method. Since the date of the Balance Sheet and, to the Knowledge of the Sellers, the dates of the balance sheets included in the Venture Financial Statements, such inventories have been maintained in the ordinary course of business. All such inventories are, to the Knowledge of the Sellers in the case of the Venture Entity inventories, owned free and clear of all Liens other than Permitted Liens. All of the inventories recorded on the Balance Sheet and, to the Knowledge of the Sellers, the balance sheets included in the Venture Financial Statements consist of, and such inventories on the Closing Date will consist of, items of a quality usable or saleable in the ordinary course of business subject to appropriate and adequate allowances, if any, reflected on the books and records of the Sellers with respect to the Business or the Sold Companies and the Subsidiaries and, to the Knowledge of the Sellers, the Venture Entities for obsolete, excess, slow-moving and other irregular items.
- 3.24 BANK ACCOUNTS. SCHEDULE 3.24 sets forth a true and complete list and description of the bank accounts, lock box accounts and other accounts maintained by or for the benefit of each of the Sold Companies and Subsidiaries.
- 3.25 NO BROKERS' OR OTHER FEES. Except for Lazard Freres & Co., LLC, whose fees and expenses will be paid by Cooper, no Person has been employed by or on behalf of the Sellers as a broker, finder, investment banker or financial advisor in connection with the transactions contemplated hereby, and no Person with which the Sellers have had any dealings or communications of any kind is entitled or will become entitled to any fee or commission, brokerage, finder's fee or like payment based in any way on any agreement, arrangement or understanding made by or on behalf of the Sellers to which the Buyer or its Affiliates (including, from and after the Closing Date, the Sold Companies, Subsidiaries and the Venture Entities) will have any obligation or responsibility in connection with the transactions contemplated hereby.
- 3.26 DIRECTORS AND OFFICERS. SCHEDULE 3.26 sets forth a true and complete list of the directors and officers of the Sold Companies and the Subsidiaries.
- 3.27 NO OTHER REPRESENTATIONS OR WARRANTIES. Except for the representations and warranties contained in this ARTICLE III and the Transaction Agreements, none of the Sellers or any other Person makes any other express or implied representation or warranty to the Buyer.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE BUYER

The Buyer hereby represents and warrants to the Sellers as follows:

- 4.1 ORGANIZATION. The Buyer is a corporation duly incorporated, validly existing and in good standing under the Laws of its jurisdiction of incorporation. The Buyer has the requisite corporate power and authority to own, lease and operate its assets and to carry on its business as now being conducted.
- 4.2 AUTHORIZATION; ENFORCEABILITY. The Buyer has the requisite corporate power and authority to execute and deliver this Agreement and the Transaction Agreements to which the Buyer is a party and to perform its obligations hereunder and thereunder. The execution and delivery of this

Agreement by the Buyer and the Transaction Agreements to which the Buyer is party and the performance by the Buyer of its obligations hereunder and thereunder have been duly authorized by all necessary corporate action on the part of the Buyer and no other corporate or shareholder proceedings or actions are necessary to authorize and consummate this Agreement, the Transaction Agreements or the transactions contemplated hereby or thereby. This Agreement has been, and each of the Transaction Agreements will be, when delivered to Sellers, duly executed and delivered by the Buyer and, assuming due authorization, execution and delivery by the Sellers, as applicable, this Agreement constitutes, and each of the Transaction Agreements will constitute, a valid and binding agreement of the Buyer, enforceable against it in accordance with their terms, subject to the General Enforceability Exceptions.

- 4.3 NO CONFLICTS OR APPROVALS. (a) The execution, delivery and performance by the Buyer of this Agreement and the Transaction Agreements to which the Buyer is a party and the consummation by the Buyer of the transactions contemplated hereby and thereby do not and will not (i) violate, conflict with or result in a breach by the Buyer of the organizational documents of the Buyer, (ii) violate, conflict with or result in a breach of, or constitute a default by the Buyer (or create an event which, with notice or lapse of time or both, would constitute a default) or give rise to any right of termination, consent, cancellation, acceleration, increased Liabilities or fees, or
- both, would constitute a default) or give rise to any right of termination, consent, cancellation, acceleration, increased Liabilities or fees, or right to increase the obligations or otherwise modify the terms under, or result in the creation of any Lien, other than Permitted Liens, upon any of the properties, rights or assets of the Buyer or to which any of such properties, rights or assets are subject under, any Contract to which the Buyer or any of its properties, rights or assets may be bound, except as would not, individually or in the aggregate, have a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement, or (iii) subject to the receipt of the requisite approvals referred to on SCHEDULE 4.3(b), conflict with or violate any Governmental Order or Law applicable to the Buyer or any of its properties, rights or assets.
- (b) Except as set forth on SCHEDULE 4.3(b), no Consent is required to be obtained, filed or delivered by the Buyer for the consummation by the Buyer of the transactions contemplated by this Agreement and the Transaction Agreements that if not obtained, filed or delivered, as the case may be, would reasonably be expected to have a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement and the Transaction Agreements.
- 4.4 PROCEEDINGS. Except as set forth in SCHEDULE 4.4, there are no Proceedings or, to the Knowledge of the Buyer, governmental investigations pending or, to the Knowledge of the Buyer, threatened against the Buyer that challenge, or question the validity of, this Agreement, any Transaction Agreement or any action taken or to be taken by the Buyer in connection with, or which seeks to enjoin or obtain monetary damages in respect of, the consummation of the transactions contemplated hereby or thereby.
- 4.5 NO BROKERS' OR OTHER FEES. No Person has been employed by or on behalf of the Buyer as a broker, finder, investment banker or financial advisor in connection with the transactions contemplated hereby, and no Person with which the Buyer has had any dealings or communications of any kind, is entitled or will become entitled to any fee or commission, brokerage, finder's fee or like payment based in any way on any agreement, arrangement or understanding made by or on behalf of the Buyer to which Seller or their Affiliates will have any obligation or responsibility in connection with the transactions contemplated hereby.
- 4.6 INVESTMENT INTENT. The Buyer is acquiring the Shares for the Buyer's own account for investment and not with a view to or for sale in connection with any distribution thereof other than in compliance with the Securities Act. The Buyer agrees that it will not transfer any of the Shares, except in compliance with the Securities Act. The Buyer further understands and acknowledges that the

Shares have not been registered under the Securities Act and agrees that the Shares may not be transferred unless (a) such transfer is pursuant to an effective registration statement under the Securities Act or (b) such transfer is exempt from the provisions of Section 5 of the Securities Act.

4.7 FINANCING. SCHEDULE 4.7 sets forth true and complete fully executed copies of the following commitment letters and related term sheets to be used in connection with the transactions contemplated hereby: (a) a commitment letter, dated September 15, 2004 (the "EQUITY FINANCING COMMITMENT"), with respect to a capital contribution to Buyer by Cypress Merchant Banking Partners II L.P. and GS Capital Partners 2000, L.P. or one or more of their respective Affiliates in the aggregate amount of \$318,000,000 ("EQUITY Financing") and (b) a commitment letter, dated September 15, 2004 (the "DEBT FINANCING COMMITMENT" and, together with the Equity Financing Commitment, the "FINANCING COMMITMENTS"), from Deutsche Bank Trust Company Americas, Deutsche Bank Securities Inc., Lehman Commercial Paper Inc., Lehman Brothers Inc., Goldman Sachs Credit Partners L.P., UBS Securities LLC and UBS Loan Finance LLC to the Buyer with respect to a \$625,000,000 senior secured credit facility and \$400,000,000 principal amount of senior subordinated notes (the "DEBT FINANCING" and together with the Equity Financing, the "ACQUISITION FINANCING"). As of the date hereof, the Financing Commitments are in full force and effect and have not been withdrawn or terminated or otherwise amended or modified in any respect. As of the date hereof, the Buyer has no reason to believe that any condition to such Financing Commitments within its control will not be satisfied by the Outside Date. The proceeds from such Acquisition Financing constitute all of the financing required to be provided by Buyer for the consummation of the transactions contemplated hereby.

4.8 NO OTHER REPRESENTATIONS OR WARRANTIES. Except for the representations and warranties contained in this ARTICLE IV and the Transaction Agreements, neither the Buyer nor any other Person makes any other express or implied representation or warranty to the Sellers.

ARTICLE V COVENANTS AND AGREEMENTS

- 5.1 CONDUCT OF THE BUSINESS PRIOR TO THE CLOSING. Except as (a) contemplated by the Restructuring, (b) otherwise expressly contemplated by this Agreement or (c) disclosed on SCHEDULE 5.1, from and after the date of this Agreement and until the Closing or the earlier termination of this Agreement in accordance with its terms, the Sellers shall, and shall cause the Sold Companies and the Subsidiaries to, and only to the extent the Sellers control the following activities with respect to the Venture Entities, shall cause the Venture Entities to: (i) conduct the operations of the Business in the ordinary course of business, (ii) use their commercially reasonable efforts to maintain and preserve intact the Business and to maintain satisfactory relationships with suppliers, customers, key employees and other Persons having business relationships with the Business and (iii) make capital expenditures in accordance with past practice. Except as (a) contemplated by the Restructuring,
- (b) otherwise contemplated by this Agreement or (c) as set forth on SCHEDULE 5.1, the Sellers shall not, and shall cause the Sold Companies and each of the Subsidiaries not to, and only to the extent the Sellers control the following activities with respect to the Venture Entities, shall cause the Venture Entities not to, do any of the following without the prior written consent of the Buyer:
- (a) purchase or sell any capital stock or other equity interests of any Sold Company or Subsidiary or take or approve any action to purchase or sell any capital stock or other equity interests of any Venture Entity or grant or make (or with respect to the Venture Entities, take or approve any action to grant or make) any option, subscription, warrant, call, commitment or agreement of any character in respect of any such capital stock or other equity interests;

- (b) acquire, lease, license, assign, sell, transfer or otherwise dispose of property, rights, businesses or assets (including by merger, consolidation or acquisition of stock or assets) related to the Business or, with respect to any property, rights, businesses or assets of any Venture Entity, take or approve any action to do any of the foregoing, excluding in all cases sales of inventory and obsolete equipment or non-exclusive licenses in the ordinary course of business;
- (c) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization with respect to any Sold Company or Subsidiary or take or approve any action to do so with respect to any Venture Entity;
- (d) in the case of any Sold Company or Subsidiary, incur, assume or guarantee any obligations for borrowed money or in the case of any Venture Entity, take or approve any action to do any of the foregoing, in all cases other than (i) intercompany loans from any of the Sellers or their subsidiaries in the ordinary course of business and (ii) under lines of credit existing on the date of this Agreement;
- (e) incur any Lien on any properties, rights or assets of any Sold Company or Subsidiary or related to the Business or, with respect to any properties, rights or assets of any Venture Entity, take or approve any action to do any of the foregoing, in each case, other than Permitted Liens;
- (f) enter into any new lease or modify, renew, extend or terminate any existing lease or purchase or acquire or enter into any agreement to acquire, assign, sell, transfer or dispose of any Real Property of any Subsidiary or Sold Company, or with respect to any Real Property of any Venture Entity, take or approve any action to do any of the foregoing;
- (g) (i) declare, set aside or pay any non-cash dividends or distributions on, or make any other non-cash distributions (whether in securities or other property) in respect of, any capital stock or equity interest of any Sold Company or Subsidiary (other than dividends and distributions to a wholly owned Sold Company or Subsidiary), or with respect to any capital stock or equity interest of any Venture Entity, take or approve any action to do any of the foregoing, (ii) split, combine or reclassify any of its outstanding capital stock or equity interest of any Sold Company or Subsidiary or issue or authorize the issuance of any capital stock or equity interest of any Sold Company or Subsidiary, or with respect to any capital stock or equity interest of any Venture Entity, take or approve any action to do any of the foregoing, (iii) purchase, redeem or otherwise acquire or dispose of any securities or equity interests of the Sold Companies or Subsidiaries, or with respect to any securities or equity interests of the Venture Entities, take or approve any action to do any of the foregoing or (iv) issue, sell, transfer, grant, pledge, dispose of or otherwise encumber any capital stock or equity interest of any Sold Company or Subsidiary, or with respect to any capital stock or equity interest of any Venture Entity, take or approve any action to do any of the foregoing;
- (h) (i) establish, adopt, enter into, amend or terminate any Company Benefit Plan or any plan, agreement, program, policy, trust, fund or other arrangement that would be a Company Benefit Plan if it were in existence as of the date of this Agreement in a manner that would affect the benefits provided to or with respect to any Company Employee or otherwise increase the Liabilities of the Sold Companies and any of its Subsidiaries under the Company Benefit Plans, or take any action to accelerate vesting under, or release any restrictions applicable under, any of the foregoing for any Company Employee,
- (ii) increase the compensation or fringe benefits of any Company Employee (except for increases in salary or hourly wage rates, in the ordinary course of business or as may be required by the collective bargaining agreements), (iii) grant any severance or termination pay to any Company Employee, except in the ordinary course of business or as may be required by the collective bargaining agreements, (iv) loan or advance any money or other property to any Company Employee, (v) increase the funding obligation to or contribution rate under any Company Benefit Plan subject to Title IV of

ERISA, except as required by Law, or (vi) terminate the employment of any of the executives listed on SCHEDULE 5.1(h);

- (i) make any material change in the accounting methods, policies or practices followed by the Business or, with respect to any Venture Entity, take or approve any action to make any such material change (in all cases, other than such changes that have been required by Law or GAAP);
- (j) (i) with respect to Tax matters of the Sold Companies and Subsidiaries, make or change any election relating to Taxes, change an annual accounting period or adopt or change any accounting method relating to Taxes, file any amended Tax Return, enter into any closing agreement, settle any Tax claim or assessment relating to the Sold Companies or the Subsidiaries, surrender any right to claim a refund, offset or credit of Taxes, consent to any extension or waiver of the limitation period applicable to any Tax claim or assessment relating to the Sold Companies or the Subsidiaries if such election, adoption, change, amendment, agreement, settlement, surrender or consent would have the effect of materially increasing the Tax liability of any Sold Company or Subsidiary for any period ending after the Closing Date and (ii) with respect to Tax matters of the Venture Entities, take or approve any action to do any of the activities in clause (i) above;
- (k) enter into, amend, supplement, waive, modify, terminate, annul, cancel, allow to lapse, assign, convey, encumber or otherwise transfer, in whole or in part, rights and interests in or under any Material Contracts or enter into any Contract that would be a Material Contract if in effect on the date of this Agreement, except Contracts made in the ordinary course of business;
- (1) (i) with respect to Sold Companies' and Subsidiaries' Proceedings, compromise, settle, grant any waiver or release relating to or otherwise adjust any right or claim with respect to any pending or threatened Proceeding (A) relating to the transactions contemplated by this Agreement or (B) against the Business, the Sold Companies or the Subsidiaries having a value in the aggregate in excess of the greater of (1) \$500,000 or (2) the reserve for any such Proceeding, or that imposes non-monetary relief against the Business, the Sold Companies or the Subsidiaries that would be materially adverse to the Business and the Sold Companies and Subsidiaries, taken as a whole and (ii) with respect to Venture Entities' Proceedings, take or approve any action to do any of the activities in clause (i) above;
- (m) (i) with respect to the Sold Companies and Subsidiaries, change, or agree to change, any business policies of the Sold Companies and Subsidiaries which relate to advertising, promotional activities, pricing, personnel, labor relations, sales, returns or warranties in each case in any material respect other than in the ordinary course of business and (ii) with respect to any such business policies of the Venture Entities, take or approve any action to do any of the activities in clause (i) above, in each case in any material respect other than in the ordinary course of business;
- (n) other than in the ordinary course of business consistent with past practices, sell or monetize any accounts receivable under any of the Factoring Agreements;
- (o) amend any organizational document of any Sold Company or Subsidiary or take or approve any action to amend any organizational document of any Venture Entity;
- (p) take or authorize any of, or commit or agree to take any action that would make any representation or warranty of Seller or any of its subsidiaries hereunder inaccurate in any respect at, or as of any time prior to, the Closing Date; or
- (q) offer, propose, authorize, agree or commit to do any of the foregoing.

- 5.2 ACCESS TO BOOKS AND RECORDS; CONFIDENTIALITY. (a) The Sellers shall cause the books and records, contracts, documents and other information of the Sold Companies and Subsidiaries to be in the sole possession and control of the Sold Companies and the Subsidiaries or the Buyer at or as soon as practicable after the Closing. To the extent that books and records, Contracts, documents and other information of the Sellers or Affiliates relevant to or affecting the Business and necessary to the operation of the Business after the Closing Date relate partially to the Sellers or their Affiliates other than the Sold Companies, the Subsidiaries or the Venture Entities, the Sellers or Affiliates shall provide the Buyer with full access to the portions of such information pertaining to the Business.
- (b) During the period commencing on the date hereof and ending on the Closing Date or the earlier termination of this Agreement in accordance with its terms, the Sellers shall, give the Buyer and its counsel, accountants, employees, agents, financing sources and other authorized representatives full access, during normal business hours, upon reasonable advance notice, to the officers, directors, employees, premises, properties, books, records, financial statements, Tax Returns relating solely to the Sold Companies, any Subsidiary and, to the extent the Sellers have authority and control to do so, the Venture Entities, data and Contracts of the Business and the Sold Companies and the Subsidiaries and, to the extent the Sellers have authority and control to do so, the Venture Entities; provided that such access does not interfere with normal business operations; and further provided that such access will not include sampling or testing of the Owned Real Property or the Leased Real Property without Sellers' prior written consent which shall not be unreasonably withheld or delayed. Any information provided to or obtained by the Buyer pursuant to this SECTION 5.2(b) will be subject to the Confidentiality Agreement and must be held by the Buyer in accordance with and be subject to the terms of the Confidentiality Agreement. The Buyer agrees to be bound by and comply with the provisions set forth in the Confidentiality Agreement as if such provisions were set forth herein, and such provisions are hereby incorporated herein by reference.
- (c) From and after the date hereof, except as (i) required by Law or
- (ii) as necessary for the preparation of any Tax Returns, the Sellers shall, and shall cause their Affiliates and respective officers, directors, employees, counsel, accountants, agents and other representatives to, (i) hold in strict confidence all information relating to the Business and the Sold Companies, Subsidiaries and Venture Entities as conducted before the Closing; provided that the foregoing shall not apply to information that is or becomes generally available to the public through no action of the Sellers. If any Seller or its Affiliates or their respective officers, directors, employees, counsel, accountants, agents and other representatives are legally compelled or required to disclose any such information, it is agreed that Sellers will promptly notify Buyer to permit Buyer to seek a protective order or take other appropriate action. Sellers will cooperate in Buyer's efforts to obtain a protective order or other reasonable assurance that confidential treatment will be accorded to such information. If, in the absence of a protective order, Sellers or any of their Affiliates and respective officers, directors, employees, counsel, accountants, agents and other representatives are, in the opinion of counsel, compelled as a matter of Law to disclose such information to a third party, Sellers may disclose to the third party compelling disclosure only the part of such information as is required by Law to be disclosed (in which case, prior to disclosure, Sellers will use reasonable efforts to advise and consult with Buyer and its counsel as to such disclosure and the nature and wording of such disclosure), and Sellers will use commercially reasonable efforts to obtain confidential treatment of any such information so disclosed. Sellers acknowledge and agree that other remedies cannot fully compensate Buyer for a violation by Sellers of the terms of this SECTION 5.2(c) and that Buyer shall be entitled to injunctive relief to prevent any such violation or continuing violation by any Seller. At the Closing, the Sellers and their subsidiaries shall assign their respective rights under any confidentiality agreement relating to the Business or the Sold Companies and Subsidiaries with a third party to Buyer or its Affiliates.

- (d) During the period commencing on the Closing Date and ending on March 31, 2005, the Buyer shall cause the employees of the Sold Companies, Subsidiaries and to the extent the Buyer has authority and control to do so, the Venture Entities to provide all cooperation reasonably necessary in connection with Cooper's regulatory reporting obligations, including, without limitation, assistance with preparing and filing reports required to be filed with the Securities and Exchange Commission.
- 5.3 SECTION 338 ELECTION. Neither Buyer nor any of its Affiliates shall make (i) any election under Section 338(h)(10) of the Code or (ii) any election under Section 338(g) with respect to the stock of the entities listed on SCHEDULE 5.3. Buyer shall be permitted to make any election under Section 338(g) of the Code with respect to any of the Sold Companies or Subsidiaries other than the entities that are listed on SCHEDULE 5.3, and the parties shall make or cause to be made all appropriate filings as required by applicable Law with respect thereto. Notwithstanding anything to the contrary in this SECTION 5.3 and subject to the following sentence, Buyer may make protective elections under Section 338(g) of the Code with respect to Cooper Standard Automotive (Deutschland) GmbH, Cooper Standard Automotive Espana S.A., and Cooper Standard Automotive Ceska' republika s.r.o. Prior to the filing of any such protective election, Buyer shall provide Cooper with a copy of each proposed protective election under Section 338(g) of the Code for its approval (which approval will not be unreasonably withheld or delayed).
- 5.4 TAX RETURNS, CONTESTS AND COOPERATION. (a) The Buyer and the Sellers agree to furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance (including powers of attorney and, at the expense of the requesting party, reasonable access to the other party's Tax Return preparer, provided that such other party may limit such access as it reasonably deems necessary to protect confidential information) relating to any of the Sold Companies or the Subsidiaries (including access to books and records, employees, contractors and representatives) as is reasonably necessary for the filing of all Tax Returns, the making of any election related to Taxes, the preparation for any audit by any Taxing Authority, and the prosecution or defense of any claim, suit or proceeding relating to any Tax or Tax Return. The Buyer and the Sellers shall retain all books and records that pertain to the Sold Companies or the Subsidiaries until the expiration of all relevant statutes of limitations (and, to the extent notified by the Buyer and the Sellers, any extensions thereof). At the end of such period, each party shall provide the other with at least thirty (30) days' prior written notice before destroying any such books and records, during which period the party receiving such notice can elect to take possession, at its own expense, of such books and records. Neither Buyer nor any of its Affiliates will revoke, amend or otherwise modify any existing powers of attorney effective for or with respect to any Current Tax Matter without the prior written consent of Cooper, which shall not be unreasonably withheld.
- (b) Seller shall prepare and timely file, or cause to be prepared and timely filed, all consolidated, combined or unitary Tax Returns in respect of any of the Sold Companies or Subsidiaries for any Pre-Closing Tax Period which will include the operations of any of the Sold Companies or any of the Subsidiaries, and shall timely pay all Taxes shown on such Tax Returns. All such Tax Returns shall be prepared in accordance with prior practices unless otherwise required by Law. The Buyer shall prepare and timely file, or cause to be prepared and timely filed, all other Tax Returns in respect of the Sold Companies and any of the Subsidiaries with respect to a Pre-Closing Tax Period, including for any Straddle Period (the "BUYER FILED TAX RETURNS"). The Buyer shall provide Cooper with a copy of each proposed Buyer Filed Tax Return (and such additional information regarding such Buyer Filed Tax Return as may reasonably be requested by Cooper) for its approval (which approval will not be unreasonably withheld or delayed) (i) at least fifteen (15) days prior to the filing of such Tax Return or (ii) in the case of a Tax Return that is required to be filed within twenty (20) days of the Closing Date, at least ten (10) days prior to the date such Tax Return is required to be filed; provided, that in the case of a

Tax Return that is required to be filed within ten (10) days of the Closing Date, the Buyer shall use its commercially reasonable efforts to afford Cooper a reasonable opportunity to review and approve such Buyer Filed Tax Return prior to filing such Tax Return. Any Buyer Filed Tax Return shall be prepared on a basis consistent with the last previous similar Tax Return; provided further that, if after consultation in good faith with the Sellers, the Buyer reasonably determines that it is required by Law to prepare any Tax Return for a Straddle Period on a basis inconsistent with the last previous similar Tax Return, then they shall so notify the Sellers and shall prepare such Tax Returns in accordance with applicable Law. For the avoidance of doubt, neither Buyer nor any of its Affiliates shall amend, refile, revoke or otherwise modify any Tax Return or Tax election of any of the Sold Companies or any Subsidiary in respect of a Pre-Closing Tax Period to the extent that such amendment, refiling, revocation or other modification would have an adverse effect on any of the Sellers or their Affiliates, without the prior written consent of Cooper, which consent shall not be unreasonably withheld or delayed.

(c) All transfer, documentary, sales, use, registration and other such Taxes (including all applicable real estate transfer Taxes, but excluding any Taxes based on or attributable to income or gains) and related fees (including any penalties, interest and additions to Tax) ("TRANSFER TAXES") arising out of or incurred in connection with this Agreement (other than in connection with the Restructuring) shall be borne by the Buyer. The party that is legally required to file a Tax Return relating to

Transfer Taxes shall be responsible for preparing and timely filing such Tax Return. All such Tax Returns shall be prepared on a basis

consistent with the last previous similar Tax Return and in accordance with applicable Law.

- (d) All Tax sharing agreements between any of the Sold Companies or Subsidiaries, on the one hand, and any of Sellers or their Affiliates (other than the Sold Companies or any of the Subsidiaries), on the other hand, shall be terminated as of the Closing Date and shall have no further effect for any taxable year (whether the current year, a future year or a past year).
- (e) (i) If any Taxing Authority asserts a claim against Buyer or any of its Affiliates with respect to the Taxes of any of the Sold Companies or any Subsidiary for a Pre-Closing Tax Period, which, if successful, would result in an indemnity payment by Sellers to Buyer pursuant to SECTION 5.5 hereof (a "TAX CLAIM"), Buyer shall promptly notify Sellers in writing of such Tax Claim stating the nature and basis of such Tax Claim and the amount thereof, to the extent known by Buyer, and shall include a copy of any correspondence received from the Taxing Authority. If notice of a Tax Claim is not given to the Sellers within a sufficient period of time to allow the Sellers to effectively contest such Tax Claim to the extent provided in this SECTION 5.4(e), the Sellers shall not be liable to the Buyer or any of its Affiliates to the extent the Sellers are actually prejudiced thereby.
- (ii) Except as otherwise provided in SECTION 5.4(e)(iii) or 5.4(e)(iv), Sellers may, at their own expense, participate in and, upon notice to Buyer, assume the defense, compromise or resolution of (A) any Tax Claim and (B) any other suit, action, audit, litigation or proceeding to the extent such claim, suit, action, litigation or proceeding relates to matters indemnifiable by Sellers pursuant to SECTION 5.5 hereof, including as a result of any breach by Sellers of any representation or warranty contained in SECTION 3.10 hereof, (including for the avoidance of doubt any Current Tax Matter) (each a "PRE-CLOSING TAX MATTER"), provided that Sellers acknowledge in writing their indemnification obligation for all Taxes relating to the Pre-Closing Tax Matter. If Sellers assume such defense, Buyer shall have the right (but not the duty) to participate in the defense thereof and to employ counsel, at its own expense, separate from counsel employed by Sellers. If Sellers do not assume the defense of such Pre-Closing Tax Matter, then Buyer may defend such Pre-Closing Tax Matter at the sole cost of Sellers and the Sellers may participate in, but not control, the defense of such Pre-Closing Tax Matter at the Sellers' sole expense. Sellers shall not enter into any settlement of or otherwise compromise any

such Pre-Closing Tax Matter to the extent that it relates to the Sold Companies or any Subsidiary and such settlement or compromise would materially and adversely affect the Tax liability of the Buyer, the Sold Companies or any Subsidiary for any Post-Closing Tax Period without the prior written consent of Buyer, which consent shall not be unreasonably withheld, conditioned, or delayed. Sellers shall, to the extent reasonably requested by Buyer, keep Buyer informed as to the status of such Pre-Closing Tax Matter, including all settlement negotiations and offers.

- (iii) Buyer shall have the right to control the defense, compromise, and resolution of any Tax claim, suit, action, audit, litigation or proceeding relating to a Straddle Period (a "STRADDLE PERIOD TAX MATTER"). If Buyer assumes such defense, Sellers shall have the right (but not the duty) to participate in the defense thereof and to employ counsel, at its own expense, separate from counsel employed by Buyer. If Buyer does not assume the defense of such Straddle Period Tax Matter, then Sellers may defend such Straddle Period Tax Matter and Buyer may participate in, but not control, the defense of such Straddle Period Tax Matter with each party bearing their own expenses. Buyer shall not enter into any settlement of or otherwise compromise any such Straddle Period Tax Matter to the extent that such settlement or compromise would materially and adversely affect the Tax liability of Sellers or any of their Affiliates for any Pre-Closing Tax Period without the prior written consent of Cooper, which consent shall not be unreasonably withheld, conditioned, or delayed. Buyer shall, to the extent reasonably requested by Sellers, keep Sellers informed as to the status of such Straddle Period Tax Matter, including all settlement negotiations and offers.
- (iv) Notwithstanding anything in this Agreement to the contrary, in the case of a Transfer Pricing Tax Matter, Sellers may defend such proceeding and Buyer may participate in the defense of such proceeding (including without limitation being present at all meetings, hearings or other proceedings), with each party bearing their own expenses. Seller shall not make any concession, settlement, closing or any other agreement with respect to any material item thereto without the consent of Buyer, which consent shall not be unreasonably withheld. Sellers shall promptly keep Buyer informed as to the status of all material items with respect to such proceedings, including without limitation all communications with the appropriate Taxing Authority, all notices (including time and location) for meetings, hearings or other proceedings, and all settlement discussions, negotiations and offers.
- (f) Other than as provided in SECTION 5.4(e) hereof, Buyer will promptly deliver to Sellers and Sellers will promptly deliver to Buyer copies of all communications sent to or received from any Taxing Authority in so far as they relate to any Pre-Closing Tax Period or Straddle Period of the Sold Companies or Subsidiaries.
- 5.5 TAX MATTERS. Sellers shall jointly and severally indemnify Buyer and its Affiliates and hold them harmless from all Losses attributable to (i) all Taxes imposed on or payable with respect to the Sold Companies or the Subsidiaries relating to a Pre-Closing Tax Period or the pre-Closing portion of a Straddle Period as determined under SECTION 5.5(c), (ii) all Taxes of any member of an affiliated, consolidated, combined or unitary group of which the Sold Companies or the Subsidiaries (or any predecessor of any of the foregoing) are or were a member on or prior to the Closing Date, including pursuant to Treasury Regulation Sections.1.1502-6 or any analogous or similar state, local, or foreign law or regulation, as a transferee or successor, by contract or otherwise (iii) any Taxes (including Transfer Taxes) arising from or in connection with the Restructuring, (iv) any Taxes arising from or in connection with any breach by Sellers of any representation, warranty or covenant contained in this Agreement and (v) all costs and expenses, including reasonable legal fees and expenses, attributable to any item in subclauses (i) (iv). Notwithstanding anything to the contrary herein, Sellers shall only be liable to Buyer for any particular Tax that is not an Income Tax pursuant to this SECTION 5.5, a breach of

representation or warranty contained in SECTION 3.10 hereof, or otherwise only to the extent the amount such Tax that is not an Income Tax exceeds the amounts reflected in the Tax Reserve.

- (b) Buyer shall indemnify Seller and its Affiliates and hold them harmless from all Losses attributable to (i) any Taxes imposed on or payable with respect to the Sold Companies or the Subsidiaries relating to a Post-Closing Tax Period or the post-Closing portion of a Straddle Period as determined under SECTION 5.5(c), (ii) any Taxes of any of the Sold Companies or Subsidiaries paid by Sellers or any of their Affiliates that is reflected in the Tax Reserve, (iii) any Transfer Taxes (other than in connection with the Restructuring), (iv) Taxes arising from or in connection with any breach by Buyer of any representation, warranty or covenant contained in this Agreement,
- (v) any and all Losses for additional Taxes owed by any of Sellers, any of its Affiliates, any of the Sold Companies or any Subsidiary (including Tax owed by any such Person due to this indemnification payment) resulting solely from any transaction engaged in by Buyer, the Sold Companies, the Subsidiaries or any of their Affiliates not in the ordinary course of business occurring on the Closing Date after Buyer's purchase of the Shares and (vi) all costs and expenses, including reasonable legal fees and expenses, attributable to any item in subclauses (i) (v). The Buyer and Cooper agree to report all transactions not in the ordinary course of business occurring on the Closing Date after Buyer's purchase of the Shares on Buyer's federal income Tax Return to the extent permitted by Treasury Regulation Section 1.1502-76(b)(1)(ii)(B).
- (c) The portion of Tax or Tax refund, including interest paid therewith, related to the pre-Closing portion of the Straddle Period shall (i) in the case of any Taxes other than sales or use taxes, value-added taxes, employment taxes, withholding taxes, and Income Taxes, be deemed to be the amount of such Tax for the entire taxable period multiplied by a fraction, the numerator of which is the number of days in the pre-Closing portion of the Straddle Period and denominator of which is the number of days in the Straddle Period, and (ii) in the case of any sales or use taxes, value-added taxes, employment taxes, withholding taxes, and Income Taxes, be deemed equal to the amount which would be payable if the Straddle Period ended on and included the Closing Date. In determining the amount of Income Tax of any of the Sold Companies or any Subsidiary for a Straddle Period attributable to an interest in a Venture Entity, the Parties shall use their commercially reasonable efforts to attribute the portion of such Income Tax related to the pre-Closing portion of the Straddle Period equal to the amount which would be payable if the Straddle Period ended on and included the Closing Date, and if the Parties are not able to so attribute, then such Income Taxes shall be apportioned in accordance with

SECTION 5.5(c)(i) above. The portion of Tax or Tax refund, including interest paid therewith, related to the post-Closing portion of the Straddle Period shall be calculated in a corresponding manner as that described above.

- (d) Any indemnity payment to be made pursuant to this SECTION 5.5 shall be paid no later than ten (10) days after the indemnified party makes written demand upon the indemnifying party, provided that no such payment shall be made later than three (3) days prior to when the Tax to which such payment relates is due and payable.
- (e) The indemnification provisions in this SECTION 5.5 shall survive the Closing until thirty (30) days after the expiration of the applicable statute of limitations.
- (f) The parties agree that any indemnification payments made pursuant to this Agreement shall be treated for all federal, state, local and foreign Tax purposes as an adjustment to the Purchase Price, unless otherwise required by applicable Law.
- (g) In calculating the amount of any indemnity payment to be made pursuant to this SECTION 5.5, such payment will be reduced by any Tax benefit actually realized by the party receiving such payment (net of any Tax detriment) in the Tax period in which the indemnification payment is made

in respect of any Losses for which such indemnification payment is made. For the avoidance of doubt, such payments shall not be reduced for additions to net operating losses.

- 5.6 REFUNDS AND TAX BENEFITS. (a) The Buyer agrees that it shall not cause or permit any of the Sold Companies or the Subsidiaries to carry back to any Straddle Period or Pre-Closing Tax Period any net operating loss, loss from operations or other Tax attribute that is attributable to a Post-Closing Tax Period (a "POST-CLOSING LOSS") except in the case of a Post-Closing Loss that would be forfeited under applicable Law unless it was first carried back to a Straddle Period or Pre-Closing Tax Period, in which case the Buyer may, at its sole expense, cause or permit any of the Sold Companies or Subsidiaries to carry back such Post-Closing Loss. The Sellers shall timely pay to the Buyer any refund, credit, offset or other Tax benefit actually realized by the Seller with respect to a Post-Closing Loss, provided that the Seller will have no obligation under this Agreement to return or remit any such refund or other Tax benefit attributable to a breach by the Buyer of the foregoing undertaking.
- (b) The Buyer shall give the Sellers at least ten (10) days notice before it files a claim for a Tax refund with respect to a Post-Closing Loss that it is permitted to carry back to a taxable period ending on or before the Closing Date pursuant to SECTION 5.6(a). Notwithstanding anything to the contrary in this Agreement, (i) the Sellers shall use any Post-Closing Loss that is carried back to a Pre-Closing Tax Period pursuant to SECTION 5.6(a) to reduce their Taxes for such taxable period and (ii) the Sellers shall in no event be obligated to reimburse or otherwise indemnify the Buyer for any Losses (including, without limitation, Losses for Taxes) resulting from the disallowance of a Post-Closing Loss.
- (c) Sellers shall be entitled to all refunds of Seller Taxes. Buyer shall promptly notify Sellers upon receipt of notice of the right to receive a refund of Seller Taxes and remit any such actual payment (or, in the case of a Straddle Period, the portion attributable to a Pre-Closing Tax Period as determined under SECTION 5.5(c) hereof) when received to Sellers promptly upon receipt. To the extent any refund of Seller Taxes is disallowed, Seller shall timely reimburse Buyer for any amount paid to it by Buyer, including any interest or penalties imposed thereon, provided that Buyer shall not be entitled to reimbursement for interest or penalties imposed with respect to a refund of Seller Taxes which is disallowed: (i) if such interest or penalties arise solely as a result of (A) actions taken (or failed to be taken) by Buyer or any of its Affiliates with respect to a Post-Closing Tax Period, or (B) any claim, suit, action, audit, litigation or proceeding relating to Taxes in respect of a Post-Closing Tax Period, or (ii) that accrue after Sellers pay to Buyer the amount due pursuant to this SECTION 5.6(c) calculated on the date of such payment.
- (d) Buyer will ensure that, to the extent permitted by Law, in respect of any Pre-Closing Tax Period or Straddle Period, that any Sold Company that can participate in the Group Relief provisions within the United Kingdom, that Buyer will reasonably assist and provide Sellers with all claims, elections, notices, and any other document that may reasonably be required to enable a Sold Company to participate in the Group Relief provisions with the remaining United Kingdom companies of Sellers. If as a result of the Closing, a Sold Company or Subsidiary is treated as having a chargeable gain, the Sellers shall procure that a relevant UK Affiliate shall, and the Buyer shall procure that the relevant Sold Company or Subsidiary shall, make a joint election to treat such chargeable gain as accruing to the Seller's UK Affiliate and not to the Sold Company or Subsidiary. For the avoidance of doubt, (i) no payment shall be made by the Sold Company or Subsidiary to the Sellers or its UK Affiliate as consideration for making the Group Relief election described above and (ii) no payment shall be made by Sellers or any of its Affiliates to Buyer or any of its Affiliates as consideration for making the Group Relief election described above.

5.7 EMPLOYEES: BENEFIT PLANS.

- (a) Employees. All (i) active employees of the Sold Companies and Subsidiaries on the Closing Date, and (ii) active employees of the Sellers who are (A) employed primarily in the Business on the Closing Date and (B) listed on SCHEDULE 5.7(a), are hereinafter referred to as the "ACTIVE COMPANY EMPLOYEES." Any employee of the Sold Companies or Subsidiaries or employee of the Sellers who is employed primarily in the Business and is listed on SCHEDULE 5.7(a) who is not actively at work on the Closing Date due to a short-term absence, whether paid or unpaid (e.g., vacation, holiday, jury duty, FMLA, pregnancy, parental and bereavement leave, military leave, emergency leave, compassionate leave, short term disability, scheduled time off, or illness or injury leave) in compliance with the applicable policies of the Sellers, Sold Companies or the Subsidiaries (or a long-term absence covered under a long-term disability Company Benefit Plan) or any employee of the Sold Companies or Subsidiaries or employee of the Sellers who is employed primarily in the Business and is listed on SCHEDULE 5.7(a) who is not actively at work on the Closing Date and is receiving worker's compensation payments as required by Law will be deemed an Active Company Employee. To the extent they are not employed by the Sold Companies or Subsidiaries on the Closing Date, the Buyer shall make offers of employment in the Business to all Active Company Employees who are employed by the Sellers primarily in the Business immediately prior to the Closing Date and are listed on SCHEDULE 5.7(a). The parties shall use their commercially reasonable efforts to cause all such employees who are offered ongoing employment with the Buyer as described in the preceding sentence to accept such offers of ongoing employment. The Buyer shall not involuntarily terminate the employment of any Active Company Employees on the Closing Date (i.e., termination other than due to death or resignation). Notwithstanding anything herein to the contrary, Seller shall retain or assume all liabilities, and the Buyer and its Affiliates shall assume no liabilities, with respect to (x) the Discontinued Business Employees, whether under the Company Benefit Plans or otherwise, and (y) any Company Benefit Plan which is not listed on SCHEDULE 3.11(a)(i)-1 and for which costs were not expressly included in the income statements of the Sold Companies or Subsidiaries.
- (b) Seller Benefit Plans. Effective as of immediately prior to the Closing Date, except as required by Law, collective bargaining agreement or otherwise contemplated by this SECTION 5.7 or the Transition Services Agreement, all Company Employees will cease active participation in, and any benefit accrual under, each of the Seller Benefit Plans other than the Assumed Plans. The Buyer acknowledges and agrees that each of the Sold Company Benefit Plans will continue to be sponsored or maintained by the Sold Companies or the Subsidiaries, as applicable, on and immediately following the Closing Date.
- (c) Continuation of Comparable Benefit Plans/Prior Service. For the period from the Closing Date through December 31, 2005 (or such other longer period required by Law or contract), the Buyer shall maintain, or shall cause the Sold Companies and the Subsidiaries to maintain and, except as contemplated by the Transition Services Agreement, perform employer's and administrators obligations arising under or relating to each of the employee benefit plans, programs, policies and arrangements for Active Company Employees (other than Active Company Employees who are subject to a collective bargaining agreement (such employees, the "UNION EMPLOYEES")), that are no less favorable in the aggregate than the employee benefit plans (excluding equity based incentives) in which such employees participated immediately prior to the Closing Date; provided, further, that nothing in this SECTION 5.7 shall be construed to mean that Buyer or any of its Affiliates cannot amend or terminate any particular Sold Company Benefit Plan or any other employee benefit, compensation or incentive plan, policy or arrangement so long as the requirements of this

SECTION 5.7 are otherwise satisfied. To the extent not otherwise required by or resulting from the operation of Law, the Buyer shall, or shall cause the Sold Companies and the Subsidiaries to, recognize each Active Company Employee's and Former Employee's service for purposes of eligibility to participate and vesting (but not for any other purpose, such as benefit

accrual or level of benefits) with the Sellers, any of the Sold Companies, the Subsidiaries or any of their respective Affiliates or their respective predecessors as of the Closing Date as service with the Buyer, the Sold Companies and the Subsidiaries, as applicable, to the extent that such service was credited under the corresponding Company Benefit Plans, under the Buyer's, the Sold Companies' and the Subsidiaries' employee welfare benefit plans, employee pension plans, vacation, disability, severance and other employee benefit plans or policies and any other such plans or policies in which the Active Company Employees and Former Employees become or may become eligible to participate on or after the Closing Date.

(d) Welfare Plans. Except as provided in this SECTION 5.7(d) and SECTIONS 5.7(i), (k), (l), (n) and (q), coverage for all Active Company Employees and their respective spouses and dependents under the Seller Benefit Plans that are welfare benefit plans within the meaning of Section 3(1) of ERISA ("SELLER WELFARE PLANS") shall cease to be effective immediately prior to the Closing Date. Sellers and the Seller Welfare Plans shall be liable for all claims incurred at any time for individuals who are not Active Company Employees or Former Employees (and their spouses and dependents) and for all claims incurred (whether or not reported) prior to the Closing Date with respect to Active Company Employees and Former Employees and their spouses and dependents under the Seller Welfare Plans. On and after the Closing Date, Buyer and the Sold Companies and Subsidiaries shall provide coverage and benefits for all Active Company Employees and their respective eligible spouses and dependents under Buyer's Welfare Plans (as defined below) and shall be responsible for all claims incurred (whether or not reported) with respect to Active Company Employees and their eligible spouses and dependents on and after the Closing Date under the Sold Company Benefit Plans and any plans sponsored by Buyer that are welfare benefit plans within the meaning of Section 3(1) of ERISA (such plans collectively, the "BUYER'S WELFARE PLANS"). For purposes of this SECTION 5.7(d) and SECTION 5.7(j), a claim will be deemed "incurred" on the date that the event that gives rise to the claim occurs (for purposes of life insurance, sickness, accident and disability programs) or on the date that treatment or services are provided (for purposes of health care programs); provided that in the event such an individual is in the hospital as an in-patient as of the Closing Date, Sellers shall remain responsible for claims and expenses incurred in connection therewith until release from such hospitalization. The Buyer shall, or shall cause the Sold Companies and the Subsidiaries to, waive any pre-existing condition limitations and eligibility waiting periods with respect to Active Company Employees and their respective spouses and dependents under the Buyer's Welfare Plans (but only to the extent such pre-existing condition limitations and eligibility waiting periods were satisfied under the Company Benefit Plans as of the Closing Date) and shall recognize (or cause to be recognized) the dollar amount of all deductibles, co-payments, co-insurance and out-of-pocket limitations incurred with respect to Active Company Employees and their respective spouses and dependents under the Company Benefit Plans during the calendar year in which the Closing occurs for purposes of satisfying the corresponding deductibles, co-payment, co-insurance or out-of-pocket limitations for such calendar year under the corresponding Buyer's welfare plans in which the Active Company Employees become entitled to participate after the Closing Date. The provisions of this SECTION 5.7(d) shall not apply in respect of any severance or termination plans, policies or arrangements.

(e) Pension Plan Transfer. Effective as of the Closing Date, the Buyer shall establish (or cause the Sold Companies or one of its Subsidiaries, as applicable, to establish) a defined benefit pension plan and trust intended to be qualified under Section 401(a) of the Code and tax-exempt under Section 501(a) of the Code ("BUYER'S PENSION PLAN") for the benefit of Active Company Employees and Former Employees (and beneficiaries thereof) who have an accrued benefit under the Cooper Spectrum Retirement Plan ("COOPER'S PENSION PLAN") at the Closing Date (collectively, the "PENSION PARTICIPANTS"). Pension Participants who are Former Employees are listed on SCHEDULE 5.7 (e) (which schedule shall be delivered on or prior to the Closing). Each Pension Participant participating in Cooper's Pension Plan at the Closing Date shall become a participant in Buyer's Pension Plan as of the Closing Date in

accordance therewith. Buyer's Pension Plan shall provide the Pension Participants with credit for their service with the Sellers, the Sold Companies and Subsidiaries, as applicable, and any of their respective Affiliates or their respective predecessors, prior to the Closing Date for all purposes for which such service was recognized under Cooper's Pension Plan including, without limitation, vesting, benefit accrual, eligibility to participate and eligibility for disability and early retirement benefits (including subsidies relating to such benefits); provided that such service credit shall not result in duplication of benefits or accruals, and shall be contingent upon the completion of the asset transfer described below in this SECTION 5.7(e)

- (ii) On the Closing Date, or as soon as practicable thereafter, following the establishment of the Buyer's Pension Plan, but in no event later than 45 days following the establishment of such plan, the Sellers and Buyer shall cause to be filed all required Forms 5310-A and any other required IRS or PBGC forms with the appropriate governmental agency in order for the Buyer's Pension Plan to receive a transfer of assets and liabilities from Cooper's Pension Plan on or following the Closing Date in accordance with applicable law and in accordance with the provisions described below in this SECTION 5.7(e).
- (iii) Effective as of the Closing Date, in accordance with the provisions of this SECTION 5.7(e), Sellers shall cause Cooper's Pension Plan and related trust to transfer to Buyer's Pension Plan and related trust, and Buyer shall cause Buyer's Pension Plan and related trust to accept from Cooper's Pension Plan and related trust, a transfer of all liabilities for benefits (including ancillary benefits) accrued under Cooper's Pension Plan by the Pension Participants, calculated as of the Closing Date in the manner described below; provided, however, that the acceptance of such liabilities shall be expressly conditioned on the completion of the asset transfer described below in this SECTION 5.7(e). Upon the completion of such transfer of liabilities, the Pension Participants shall become participants in Buyer's Pension Plan with respect to the liabilities so transferred. Without limiting the generality of the foregoing, subject to paragraph (vii) below, on and after the Closing Date, Buyer's Pension Plan shall provide to the Pension Participants all benefits (including ancillary benefits) earned by such individuals under Cooper's Pension Plan, up to the Closing Date, as well as any benefits accrued by such individuals under Buyer's Pension Plan on and after the Closing Date.
- (iv) As soon as practicable following the establishment of Buyer's Pension Plan and satisfaction of any applicable regulatory filing requirements and the requirements set forth in this SECTION 5.7(e), but not later than 180 days following satisfaction of such requirements unless the parties otherwise agree in writing (the "FINAL TRANSFER DATE"), Sellers shall cause a transfer of the assets in respect of the transferred liabilities from the trust(s) established pursuant to Cooper's Pension Plan to the trust(s) established pursuant to Buyer's Pension Plan to accept such transfer of assets, in a total amount determined in accordance with Code Section 414(l) and this SECTION 5.7(e). In no event shall the total amount so transferred be less than the amount that is necessary to satisfy the requirements of Section 414(l) of the Code. Such transfer of assets shall be in cash or in marketable securities designated by Cooper that are held in the trust established pursuant to Cooper's Pension Plan. Unless the Sellers and Buyer agree otherwise in writing, all transfers shall occur on the last business day of a month. Cooper's Actuary (as defined below) shall be responsible for the required actuarial certification under Section 414(l) of the Code.
- (v) Sellers shall cause the enrolled actuary for Cooper's Pension Plan ("COOPER'S ACTUARY") to calculate the amount of assets to be transferred in respect of such transferred liabilities, as if Cooper's Pension Plan were terminated on the Closing Date using the "safe harbor" assumptions (as in effect on the Closing Date) that would be used by the PBGC to

calculate such benefits upon the termination of such plan (assuming no ancillary benefits have been amended out of Cooper's Pension Plan prior to such calculation), (the "TRANSFERRED BENEFIT LIABILITY") on the basis of benefits accrued by the Pension Participants under Cooper's Pension Plan as of the Closing Date, in accordance with the requirements of Section 414(l) of the Code. The amount of the assets to be transferred from Cooper's Pension Plan to Buyer's Pension Plan in respect of such Transferred Benefit Liabilities, as calculated pursuant to this SECTION 5.7(e), shall be referred to as the "TRANSFER AMOUNT."

- (vi) Sellers shall be responsible for the payment of all fees and expenses incurred by Sellers, Cooper's Actuary or Cooper's Pension Plan in the calculation and transfer of the Transfer Amount to the extent such fees and expenses are not paid from the assets of Cooper's Pension Plan. Buyer shall be responsible for the payment of all fees and expenses incurred by Buyer, the enrolled actuary for Buyer's Pension Plan or Buyer's Pension Plan in connection with the calculation and transfer of the Transfer Amount. The actuarial calculation of the Transferred Benefit Liability and Transferred Amount as well as assumptions and methodologies used to calculate such Transferred Benefit Liability and Transferred Amount determined by Cooper's Actuary shall be subject to review and challenge by an actuarial firm designated by Buyer. In the event of a good faith dispute between those two actuaries as to the amount to be transferred under this SECTION 5.7(e), and such dispute remains unresolved for a period of thirty (30) days, the chief financial officers of Cooper and Buyer, respectively, will endeavor to resolve the issue. Should such dispute remain unresolved for a period of twenty (20) days, Sellers and Buyer shall select and appoint a third-party independent actuary who is mutually satisfactory to Sellers and Buyer. Such third-party actuary shall be instructed to render its decision within twenty (20) days and such decision shall be conclusive as to any dispute for which the third-party actuary was appointed. The cost of such third-party actuary shall be divided equally between Sellers and Buyer.
- (vii) From and after the Closing Date until the Final Transfer Date, any benefit payable to a Pension Participant shall be paid and continue to be paid out of Cooper's Pension Plan trust. On and after the Final Transfer Date, any such benefits payable to a Pension Participant shall be paid from Buyer's Pension Plan trust.
- (viii) All amounts to be transferred between the trust(s) established pursuant to Cooper's Pension Plan and the trust established pursuant to Buyer's Pension Plan pursuant to SECTION 5.7(e)(i) shall include investment gains or losses on the Transferred Amount from the Closing Date to the Final Transfer Date, such investment gains and losses being determined on the basis of the aggregate net investment experience of Cooper's Pension Plan during such period.
- (ix) Notwithstanding any provision of this SECTION 5.7(e) to the contrary, no transfer of assets from Cooper's Pension Plan shall be made prior to such time as Buyer has provided to Sellers (A) evidence reasonably satisfactory to Sellers that Buyer has timely completed all Governmental Authority filings or submissions needed in order for Buyer's Pension Plan to receive a transfer of assets from Cooper's Pension Plan; and (B)(1) a current and valid favorable IRS determination letter with respect to Buyer's Pension Plan, or (2) a representation from Buyer that Buyer's Pension Plan is intended to qualify under Code Section 401(a) and that Buyer will timely file (or has already filed) an application for such favorable determination letter with respect to Buyer's Pension Plan with the IRS and that Buyer will make any and all necessary amendments on a retroactive basis to Buyer's Pension Plan as are required by the IRS to obtain such favorable determination.

- (f) Savings Plan Transfer. (i) Effective as of the Closing Date, the Buyer shall establish (or cause the Sold Companies or one of its Subsidiaries, as applicable, to establish) a defined contribution pension plan intended to be qualified under Section 401(a) of the Code ("BUYER'S SAVINGS PLAN") for the benefit of Active Company Employees and Former Employees (and beneficiaries thereof) who have an account balance under the Cooper Spectrum Investment Savings Plan ("COOPER'S SAVINGS PLAN") at the Closing Date (collectively, the "SAVINGS PARTICIPANTS"). Savings Participants who are Former Employees are listed on SCHEDULE 5.7(f) (which schedule shall be delivered on or prior to the Closing). Buyer's Savings Plan shall provide such Savings Participants with credit for their service with the Sellers, Sold Companies and Subsidiaries, as applicable, and any of their respective Affiliates or their respective predecessors, prior to the Closing Date for all purposes for which service was credited under Cooper's Savings Plan); provided that such service credit shall not result in duplication of benefits or accruals, and shall be contingent upon the completion of the asset transfer described below in this SECTION 5.7(f).
- (ii) On the Closing Date, or as soon as practicable thereafter, the Sellers and Buyer shall cause to be filed all required Forms 5310-A and any other required IRS forms in order for the Buyer's Savings Plan to receive a transfer of assets from Cooper's Savings Plan on or following the Closing Date in accordance with applicable law and in accordance with the provisions described below in this SECTION 5.7(f).
- (iii) As soon as practicable following the establishment of the Buyer's Savings Plan and satisfaction of any applicable regulatory filing requirements and the requirements set forth in this SECTION 5.7(f), but not later than ninety (90) days following satisfaction of such requirements unless the parties otherwise agree in writing, Sellers shall cause a transfer of the plan accounts, valued as of the date of transfer, of the Savings Participants, from Cooper's Savings Plan and related trust to Buyer's Savings Plan and related trust, and Buyer shall cause Buyer's Savings Plan and related trust to accept such transfer of assets (the date on which such transfer occurs is referred to as the "SAVINGS TRANSFER DATE"), including outstanding loan notes and in-kind transfers of property designated by Cooper. Unless the Sellers and Buyer agree otherwise in writing, all transfers shall occur on the last Business Day of a month.
- (iv) From and after the Closing Date until the Savings Transfer Date, any benefit payable to a Savings Participant shall be paid and continue to be paid out of Cooper's Savings Plan trust (it also being understood that Cooper's Savings Plan shall continue to administer such accounts through the Savings Transfer Date, including participant investment directions). On and after the Savings Transfer Date, any such benefits payable to a Savings Participant shall be paid from Buyer's Savings Plan trust. Sellers and Buyer shall work together to develop a process whereby Savings Participants who have loans outstanding under Cooper's Savings Plan as of the Closing Date will be permitted to continue to make periodic repayments on such outstanding loans through reduction of salary paid by Buyer or the Sold Companies or Subsidiaries, and Buyer or the Sold Companies or Subsidiaries remitting such payments to Cooper's Savings Plan on a timely basis.
- (v) Notwithstanding any provision of this SECTION 5.7(f) to the contrary, no transfer of assets from Cooper's Savings Plan shall be made prior to such time as Buyer has provided to Sellers either (A) a current and valid favorable IRS determination letter with respect to Buyer's Savings Plan, or (B) a representation from Buyer that Buyer's Savings Plan is intended to qualify under Section 401(a) of the Code and that Buyer will timely file (or has already filed) an application for such favorable determination letter with respect to Buyer's Savings Plan with the IRS and that Buyer will make any and all necessary amendments on a retroactive basis to Buyer's Savings Plan as are required by the IRS to obtain such favorable determination.

- (g) Assumed Plans. Effective as of the Closing Date, Buyer shall assume sponsorship of and, except as otherwise expressly provided in this SECTION 5.7, all obligations under, Liabilities with respect to, and assets with respect to, the Company Benefit Plans set forth on SCHEDULE 5.7(g) (the "ASSUMED PLANS"). Sellers and Buyer shall take all actions necessary to transfer such sponsorship and assets to Buyer as of the Closing Date. Except as otherwise expressly provided in this SECTION 5.7, Seller shall retain and assume all obligations under and Liabilities with respect to any Seller Benefit Plan which is not an Assumed Plan.
- (h) Unused Vacation. The Buyer shall, or shall cause the Sold Companies and the Subsidiaries to, credit each Active Company Employee with the unused vacation days to which the Active Company Employee is entitled through the Closing Date; provided that if Sellers shall be required by applicable Law to pay any Active Company Employee the cash value of their unused vacation days described in this SECTION 5.7 (h), then Buyer shall reimburse Sellers for the amount so paid by Sellers and shall not be required to honor such days.
- (i) Severance Obligations; Stay Bonus.
- (i) Buyer's Obligations. For the period from the Closing Date until December 31, 2005, subject to the immediately following SECTION 5.7(i) (ii), the Buyer shall, or shall cause the Sold Companies and the Subsidiaries to, provide severance benefits to any Company Employee (that is not a Union Employee), in accordance with the severance procedures of the Sellers, the Sold Companies or the Subsidiaries set forth on EXHIBIT C, for (A) terminations of employment on or after the Closing, not including terminations due solely to this Agreement or to the consummation of the transaction contemplated hereby and (B) terminations of employment before the Closing with respect to which a Company Employee is receiving ongoing severance payments as of the Closing, but only to the extent such severance benefits are payable pursuant to arrangements both specifically identified on SCHEDULE 3.11(a)(i)-1 and accrued on the financial statements as of the Closing Date, which accruals will be computed in the ordinary course of business consistent with past practice. In addition, subject to SECTION 5.7(i) (ii), Buyer shall assume all obligations and Liabilities under the arrangements set forth on SCHEDULE 5.7(i)(i) with respect to any individuals listed on that Schedule who execute a Waiver prior to the Closing Date.
- (ii) Sellers' Obligations. Notwithstanding any other provision of this Agreement, (A) Sellers shall be responsible for and retain all obligations and Liabilities with respect to severance, stay bonus obligations, sale incentives and termination payments or benefits, including, without limitation, any tax gross-up benefits, under all employment related agreements, including, without limitation, those identified on SCHEDULE 3.11(g), or otherwise triggered solely by the Restructuring or the sale of the Shares, and all severance and termination payments or benefits incurred by Company Employees on or prior to the Closing Date, whether under Company Benefit Plans or otherwise, other than amounts that are payable by Buyer pursuant to SECTION 5.7(i)(i)(B) above; (B) Sellers shall be responsible for and retain all obligations and Liabilities under the Severance Plan with respect to individuals other than those listed on SCHEDULE 5.7(i)(i) and individuals on SCHEDULE 5.7(i)(i) who do not execute a Waiver prior to the Closing Date; and (C) with respect to the arrangements set forth on SCHEDULE 5.7(i)(i) with respect to individuals who execute a Waiver prior to the Closing Date, (x) Sellers shall indemnify, reimburse and hold Buyer harmless for, all obligations and Liabilities (including, without limitation, any tax gross-up benefits) under such arrangements (to the extent such obligations and liabilities (i) arise pursuant to the terms of such arrangements in effect immediately prior to the Closing and (ii) in the case of the Employment Agreement, arise from a termination of employment during the Term (as defined in the Employment Agreement) in effect as of the Closing Date, without regard to any extensions of the Term under such Employment

Agreement after the Closing Date, including any automatic extensions of the Term after the Closing Date pursuant to Section 3 of the Employment Agreement) arising in connection with termination of employment of any of the individuals party thereto, provided that Buyer shall be responsible solely for severance and termination payments made pursuant to Section 1(a) of Exhibit C to the Severance Plan (for the individuals listed on SCHEDULE 5.7(i)(i) or Section 6(a)(i) of the Employment Agreement and, to the extent accrued on the financial statements as of the Closing Date, which accruals will be computed in the ordinary course of business consistent with past practice, any amount paid pursuant to Section 1(c)(ii) of Exhibit C to the Severance Plan (for the individuals listed on SCHEDULE 5.7(i)(i) or Section 6(a)(iii)(B) of the Employment Agreement (but, for the avoidance of doubt, Buyer shall not be responsible for, and Seller shall indemnify Buyer with respect to, any tax gross-up benefits associated with such payments) and (y) Sellers shall, in accordance with the Rabbi Trust (as defined below), fully and irrevocably fund the trusts contemplated by Section 6 of the Severance Plan and Section 11 of the Employment Agreement (the "RABBI TRUST") with the aggregate present value on the Closing Date of the payments potentially payable pursuant to Sections 5 and 7 of the Severance Plan (for such individuals listed on SCHEDULE 5.7(i)(ii) and Sections 6 and 12 of the Employment Agreement and for which Sellers have responsibility pursuant to clause (x) of SECTION 5.7(i)(ii)(C), it being understood and agreed that any severance or termination benefits ultimately payable under such arrangements for which Sellers have responsibility pursuant to clause (x) of SECTION 5.7(i)(ii)(C) will be satisfied first, and exclusively, by the Rabbi Trust with any remaining balance paid by Buyer, subject to indemnity and reimbursement of Buyer by Sellers for any such amounts ultimately paid by Buyer.

- (iii) Cooperation. For purposes of complying with the terms set forth in SECTION 5.7(i), each party shall cooperate with and make available to the other parties and their respective representatives all information, records, data and working papers, and shall permit reasonable access to its facilities and personnel, as may be reasonably required in connection with the determination of (A) the amount Sellers are obligated to contribute to the Rabbi Trust prior to the Closing Date and (B) the amount of any 280G excise taxes incurred by Company Employees in connection with, or arising out of, the transactions contemplated hereby and the amount of any tax gross-up obligations related thereto. Without limiting the generality of the foregoing, Sellers will provide Buyer with calculations of the amounts Sellers contribute to the Rabbi Trust, including without limitation all data used to complete such calculations. Buyer and its advisors will have ten (10) Business Days to review such calculations and data. If Buyer reasonably objects to the calculations within the time period set forth above, then Cooper and Buyer jointly shall engage the Arbitration Firm to resolve such dispute and whose fees shall be borne equally by Cooper and Buyer. The determination by the Arbitration Firm shall be binding on the parties.
- (j) Retiree Welfare Benefits. Effective as of the Closing Date, the Buyer shall assume (or cause the Sold Companies and Subsidiaries, if applicable, to continue to honor), to the extent not otherwise required by or resulting from, operation of Law or collective bargaining agreements, all post-employment and post-retirement welfare benefit obligations with respect to all Active Company Employees and Former Employees, and their respective spouses and dependents for claims incurred on and after the Closing Date and Sellers shall retain or assume all such obligations for claims incurred prior to the Closing Date. Former Employees with respect to whom post-employment and post-retirement welfare benefit obligations exist are listed on SCHEDULE 5.7(j) (which schedule shall be delivered on or prior to the Closing). Buyer shall establish a post-retirement welfare benefit plan ("BUYER'S PRB PLAN") for the benefit of the Active Company Employees and Former Employees and their respective spouses and dependents that provides benefits that are substantially comparable to the benefits provided under Sellers' post-retirement welfare plan. Benefits under Buyer's PRB Plan may not be decreased or

terminated prior to December 31, 2005 (or in the case of Union Employees, until the expiration of the applicable collective bargaining agreement).

- (k) Health Care and Dependent Care Spending Accounts. Active Company Employees who have health care spending accounts or dependent care spending accounts, on the Closing Date, under any Seller Benefit Plan ("COOPER'S FLEXIBLE ACCOUNT PLAN") will continue to participate in Cooper's Flexible Account Plan with respect to the health care spending accounts or dependent care spending accounts for claims incurred through December 31, 2004. Effective on and after January 1, 2005, such Active Company Employees will participate in one or more comparable plans of Buyer.
- (l) Continuation Coverage. The Sellers shall have the sole responsibility for "continuation coverage" benefits provided on and after the Closing Date under Seller Benefit Plans for all Active Company Employees and Former Employees and "qualified beneficiaries" of Active Company Employees and Former Employees for whom a "qualifying event" occurs before or on the Closing Date, provided that with respect to Active Company Employees and "qualified beneficiaries" for whom a qualifying event occurs on the Closing Date and who elect continuation coverage under Seller Benefit Plans (i) Seller shall make such coverage available to such individuals on the same basis (including premium cost) as is generally provided to active employees of Seller and (ii) Buyer shall reimburse Seller for any such coverage actually provided at 100% of the applicable premium for such period. The terms "continuation coverage," "qualified beneficiaries" and "qualifying event" shall have the meanings ascribed to them under Section 4980B of the Code and sections 601-608 of ERISA or the relevant Seller Benefit Plan policy.
- (m) Nonqualified Plans. Effective as of the Closing Date, the Buyer shall assume (or cause the Sold Companies and Subsidiaries, if applicable, to assume) all liabilities under the nonqualified plans and arrangements identified on SCHEDULE 5.7(m) ("COOPER'S SERP") related to the Active Company Employees and Former Employees. Buyer shall establish (or cause the Sold Companies and Subsidiaries, if applicable, to establish) a non-qualified supplemental executive retirement plan for the benefit of such Active Company Employees and Former Employees, which shall provide the benefits under Cooper's SERP the liability for which is assumed by Buyer pursuant to this SECTION 5.7(m) to the extent accrued on the financial statements as of the Closing Date, which accruals will be computed in the ordinary course of business consistent with past practice, and Seller shall retain and assume any remaining liabilities under Cooper's SERP.
- (n) (i) U.S. Workers' Compensation. This SECTION 5.7(n)(i) applies solely to workers' compensation arrangements maintained in the United States. Notwithstanding anything herein to the contrary, (A) Buyer shall be solely responsible for all Liabilities and obligations arising under workers' compensation arrangements with respect to Active Company Employees and Former Employees to the extent such Liability or obligation relates to claims arising from accidents or illnesses occurring during the period from and including the Closing Date and thereafter and (B) Sellers and their subsidiaries (other than the Sold Companies and Subsidiaries) shall be solely responsible for (x) all Liabilities and obligations arising under workers' compensation arrangements for any individuals who are not Active Company Employees or Former Employees and (y) except as otherwise set forth in this SECTION 5.7(n)(i), all Liabilities and obligations arising under workers' compensation arrangements with respect to Active Company Employees and Former Employees to the extent such Liability or obligation relates to accidents or illnesses occurring during the period prior to the Closing Date, including, without limitation, Liability for any retroactive workers' compensation premiums attributable to such period ("PRE-CLOSING WORKERS' COMPENSATION CLAIMS"); provided that with respect to any Pre-Closing Workers' Compensation arrangements (including the deductible portion of any insured workers' compensation arrangements and any amount in excess of the per incident maximum under any insured workers' compensation arrangements) ("PRE-CLOSING SELF INSURED WORKERS'

COMPENSATION ARRANGEMENTS") which remain unsatisfied on the Closing Date, Buyer shall reimburse Sellers for (and shall indemnify and hold Sellers harmless from and against) the cost of satisfying such Pre-Closing Workers' Compensation Claims under such Pre-Closing Self Insured Workers' Compensation Arrangements, including a ratable portion (i.e., attributable to only Active Employees' and Former Employees' Pre-Closing Workers' Compensation Claims) of the cost of any self-insurance fee paid to be self-insured in states in which the Sellers maintain Pre-Closing Self Insured Workers' Compensation Arrangements and a ratable portion (i.e., attributable to only Active Employees' and Former Employees' Pre-Closing Workers' Compensation Claims) of the cost of any letter of credit securing the Seller's obligation to satisfy such Pre-Closing Workers' Compensation Claims on an "as paid basis" (or as soon as practicable thereafter) and use commercially reasonable efforts to cooperate with Seller and the existing third party administrators currently administering such Pre-Closing Self Insured Workers' Compensation Arrangements to establish Buyer escrow accounts with such third party administrators, as of, or as soon as practicable following the Closing Date, to enable Buyer to pre-fund approximately one month of such Pre-Closing Workers' Compensation Claims in lieu of Seller's payment of such Pre-Closing Workers' Compensation Claims subject to Buyer's reimbursement obligation. Sellers and Buyer shall cause the appropriate personnel of the Sellers and the Sold Companies and the Subsidiaries, respectively, to cooperate following the Closing in the administration of the workers' compensation arrangements described in this SECTION 5.7(n)(i). In addition, Buyer agrees to use commercially reasonable efforts to enter into negotiations for new arrangements, on commercially reasonable and market normative terms, with third party administrators and/or any applicable insurance companies and governmental authorities as may be necessary to enable the Buyer to assume liability for the administration and satisfaction of such Pre-Closing Workers' Compensation Claims under the Pre-Closing Self Insured Workers' Compensation Arrangements and, if permitted by such insurance companies, with respect to Pre-Closing Workers' Compensation Claims which are insured claims, without further involvement of, or liability of, the Seller, as soon as practicable following the Closing Date. Sellers agree to use commercially reasonable efforts to cooperate with Buyer in such negotiations and assumption of liability.

(ii) UK Employer's Liability. This SECTION 5.7(n)(ii) applies solely to employer's liability arrangements maintained in the United Kingdom. Notwithstanding anything herein to the contrary, (A) Buyer shall be solely responsible for all Liabilities and obligations arising in respect of claims made against a Sold Company or a Subsidiary by Active Company Employees and Former Employees to the extent such Liability or obligation relates to claims made against a Subsidiary or a Sold Company arising from accidents or illnesses occurring during the period from and including the Closing Date and thereafter and (B) Sellers and their subsidiaries (other than the Sold Companies or a Subsidiary) shall be solely responsible for (x) all Liabilities and obligations arising under employer's liability arrangements for any individuals who are not Active Company Employees or Former Employees and (y) except as otherwise set forth in this SECTION 5.7(n)(ii), for all Liabilities and obligations arising in respect of an employer liability insurance policy maintained by Sellers or one of their subsidiaries (other than the Sold Companies or a Subsidiary) with respect to claims made by Active Company Employees and Former Employees to the extent such Liability or obligation relates to accidents or illnesses occurring during the period prior to the Closing Date, including, without limitation, Liability for any retroactive employer liability insurance premiums attributable to such period ("PRE-CLOSING EMPLOYER'S LIABILITY INSURANCE CLAIMS"); provided that with respect to any Pre-Closing Employer's Liability Insurance Claims that are subject to an excess or deductible (whether individual or aggregate) which remain unsatisfied on the Closing Date, Buyer shall reimburse Sellers for (and shall indemnify and hold Sellers harmless from and against) the cost of satisfying the deductible or excess portion of any such Pre-Closing Employer's Liability Insurance Claims under such employer's liability insurance including a ratable portion (i.e., attributable to only Active Company Employees' and Former Employees' Pre-Closing Employers' Liability

Insurance Claims) of the cost of any letter of credit securing the Sellers' obligation to satisfy such Pre-Closing Employer's Liability Insurance Claims on an "as paid basis" (or as soon as practicable thereafter) and use commercially reasonable efforts to cooperate with Sellers and the existing third party administrators currently administering such Pre-Closing Employer's Liability Insurance Claims to establish Buyer escrow accounts with such third party administrators, as of, or as soon as practicable following the Closing Date, to enable Buyer to pre-fund approximately one month of the deductible or excess portion of such Pre-Closing Employer's Liability Insurance Claims in lieu of Sellers' payment of such deductible or excess portion of Pre-Closing Employer's Liability Insurance Claims subject to Buyer's reimbursement obligation. Sellers and Buyer shall cause the appropriate personnel of the Sellers and the Sold Companies and the Subsidiaries, respectively, to cooperate following the Closing in the administration of the Employer's Liability Insurance arrangements described in this SECTION 5.7(n)(ii). In addition, Buyer agrees to use commercially reasonable efforts to enter into negotiations for new arrangements, on commercially reasonable and market normative terms, with applicable insurance companies as may be necessary to enable the Buyer to assume liability for the administration and satisfaction of such deductible or excess portion of Pre-Closing Employer's Liability Insurance Claims under the employer's liability insurance arrangements and, if permitted by such insurance companies, with respect to Pre-Closing Employer's Liability Insurance Claims which are insured claims, without further involvement of, or liability of, the Sellers, as soon as practicable following the Closing Date. Sellers agree to use commercially reasonable efforts to cooperate with Buyer in such negotiations and assumption of liability.

- (iii) Sellers and Buyer shall cooperate following the Closing Date in the provision of all documents, information, evidence and materials in their possession (or in the possession of any of their Affiliates) reasonably requested by the other with respect to any Pre-Closing Workers' Compensation Claim or Pre-Closing Employer's Liability Claim, as applicable, for or with respect to which Buyer, the Sold Companies or any Subsidiary is responsible in accordance with this SECTION 5.7(n). Sellers shall provide Buyer with supporting documentation evidencing the calculation of Buyer's reimbursement obligation set forth in this SECTION 5.7(n), which calculation shall be subject to Buyer's review and approval. If Buyer disputes such calculation and such dispute cannot in good faith be resolved between Buyer and Sellers, then Buyer and Sellers jointly shall engage the Arbitration Firm to resolve such dispute and the Arbitration Firm's fees shall be borne equally by Buyer and Sellers. The determination by the Arbitration Firm shall be binding on the Buyer and Sellers.
- (o) Annual Bonuses. Sellers shall pay or cause to be paid to all Company Employees any bonuses that are earned and become due, or would be paid in accordance with Sellers' past practice and in the ordinary course, prior to the Closing Date under the Seller Benefit Plans or any other annual incentive compensation plans of the Sellers or their Affiliates.
- (p) Required Contributions. On or prior to the Closing Date, Sellers shall make, or shall cause to be made, all contributions (including without limitation, all employer matching or other contributions and employee salary reduction contributions) to and payments from any Seller Benefit Plan in respect of any Company Employees (except those distributions to be made from a trust qualified under Section 401(a) of the Code) that pursuant to the Seller Benefit Plans or in accordance with Sellers' past practice and in the ordinary course, become due or would normally be paid prior to the Closing Date.
- (q) Employees on Leave. Active Company Employees who are absent from work as of the Closing Date because of disability and who are participants in the long-term disability insurance plan maintained by Sellers (the "SELLER'S LONG-TERM DISABILITY PLAN") shall continue to be covered with

respect to such disability under the Seller's Long-Term Disability Plan following the Closing for as long as such coverage remains in effect under such plan as in effect from time to time.

(r) Cooperation. The parties agree to furnish each other with such information concerning employees, employee payroll and employee benefit plans, subject to confidentiality and privacy considerations, and to take all such other action, as is necessary and appropriate to effect the transactions contemplated hereby.

(s) Siebe Automotive Pension Plan. Cooper-Standard Automotive UK Fluid Systems Limited ("CSA UK Fluid") shall continue as the sponsoring company of the Siebe Automotive Pension Plan (the "SIEBE PLAN") subject to the Definitive Trust Deed and Rules of the Siebe Plan dated 31 March 2000, as subsequently amended (the "SIEBE RULES"), provided that the Sellers shall, subject to the provisions of this SECTION 5.7(s), reimburse and indemnify Buyer, and hold Buyer harmless, for the amount of the Applicable Percentage (as defined below) of the actual contributions made to the Siebe Plan on and after the Closing Date (other than the portions of such contributions that are made by CSA UK Fluid to fund the future service benefits of Active Company Employees of CSA UK Fluid) and of the reasonably and properly incurred out-of-pocket expenses of the Siebe Plan for external actuaries and legal advisors and the trustee on and after the Closing Date (collectively, the "PERIODIC COSTS"). The Applicable Percentage as of the Closing Date shall be determined by the actuary appointed by the trustee of the Siebe Plan as appointed under section 47 of the Pensions 1995 Act (the "SIEBE ACTUARY") as soon as practicable following the Closing Date and be approved by an actuary appointed by Buyer (the "BUYER'S ACTUARY") and, if requested by Sellers, an actuary appointed by Sellers (the "SELLERS' ACTUARY"). Seller shall establish an escrow account, as of the Closing Date, and the Sellers shall prefund approximately three months of the Periodic Costs (as reasonably determined by the then-current Siebe Actuary) from which the Buyer may draw funds to satisfy the Sellers reimbursement obligation as it becomes due from time to time. Sellers shall also retain liability for and shall reimburse and indemnify Buyer and hold Buyer harmless, for the amount of the Applicable Percentage (as of the later of the termination date specified for the purpose of Clause 28.2.2 of the Siebe Rules and the commencement of winding up of the Siebe Plan) of any wind-up or termination liability payable by CSA UK Fluid or any successor thereof as sponsoring company to the Siebe Plan on any wind-up or termination of the Siebe Plan on or after the Closing Date. If the Buyer's Actuary and/or the Sellers' Actuary does not approve of the Applicable Percentage as of the Closing Date within thirty (30) Business Days following its receipt of such calculation, then Sellers and Buyer shall jointly appoint a mutually acceptable independent actuary (failing agreement such actuary shall be appointed by the then President of the Institute of Actuaries) to determine the Applicable Percentage, and the fees of such independent actuary shall be borne equally by Sellers and Buyer. The Buyer's Actuary, the Sellers' Actuary and the independent actuary shall be supplied with such data by the Buyer as they may reasonably request for the purpose of the approval of the Applicable Percentage. The determination by such independent actuary shall in the absence of manifest error be binding on the parties. After the Closing, the Applicable Percentage and the Periodic Costs shall be determined by the Siebe Actuary at each actuarial valuation of the Siebe Plan subject to approval by the Sellers' Actuary and the Buyer's Actuary. If the Sellers' Actuary and the Buyer's Actuary do not approve of the Applicable Percentage or the Periodic Costs as determined from time to time after the Closing Date within thirty (30) Business Days following their receipt of such calculation, then Sellers and Buyer shall jointly appoint a mutually acceptable independent actuary (failing agreement such actuary shall be appointed by the then President of the Institute of Actuaries) to determine the Applicable Percentage and the Periodic Costs, and the fees of such independent actuary shall be borne equally by Sellers and Buyer. Any such independent actuary shall act as an expert and not as an arbitrator. The determination by such independent actuary shall, in the absence of manifest error, be binding on the parties.

- (ii) The Applicable Percentage shall be calculated (as a percentage) as the ratio of "A" divided by "B" where:
- "A" equals the liability in respect of Discontinued Business Employees and all other deferred pensioners or pensioners of the Siebe Plan who at Closing were pensioners or deferred pensioners; and
- "B" equals the liability in respect of all members of the Siebe Plan (including Discontinued Business Employees and all other deferred pensioners or pensioners).
- "A" and "B" shall be calculated using the assumptions and methods in the Ongoing Basis used in the most recent actuarial valuation, updated from time to time, or on wind-up, the actual costs of discharging the liabilities.
- (iii) Subject to the provisions of this SECTION 5.7(s), the Siebe Plan shall be administered as set forth in this clause (iii).
- (A) The Buyer and CSA UK Fluid shall have sole responsibility for the administration and conduct of the Siebe Plan and shall procure that the Siebe Plan is administered in all respects in accordance with applicable Law and the Siebe Rules.
- (B) Sellers shall be entitled (at their own expense) from time to time on reasonable notice to Buyer, to examine, or have their representative examine, the books and records of CSA UK Fluid and of the Siebe Plan to confirm the level of contributions made by CSA UK Fluid to the Siebe Plan and that the expenses of operating the Siebe Plan have been properly incurred.
- (C) The Buyer shall use commercially reasonable efforts to avoid a wind-up of the Siebe Plan.
- (D) The Buyer shall not amend the Siebe Plan if such amendment would have the effect of materially increasing the obligations of the Sellers under this SECTION 5.7(s) without the prior written consent of the Sellers, other than such amendments as are required by applicable Law.
- (t) Secunded Employees. Prior to the Closing Date, Cooper shall cause the employment of the employees listed on SCHEDULE 5.7(t) (who are, on the date of this Agreement, active employees of Cooper who are employed primarily in the Business) (the "SECUNDED EMPLOYEES") to be transferred to Cooper-Standard Automotive Inc. ("CSA"). From the time of such transfer until December 31, 2005 (the "SECUNDMENT PERIOD"), Buyer shall cause CSA to permit the secundment of the Secunded Employees to Cooper with respect to up to 50% of their working time (as requested by Cooper) to perform the duties for Cooper set forth on SCHEDULE 5.7(t). Cooper shall reimburse CSA for the pro rata portion of the employment costs incurred by CSA with respect to the Secunded Employees during the Secundment Period.
- (u) No Third Party Beneficiary Rights; No Right to Employment. Nothing herein expressed or implied shall confer upon any of the employees of the Sellers, the Buyer, the Sold Companies, the Subsidiaries or any of their Affiliates, any rights or remedies, including any right to benefits or employment, or continued benefits or employment, for any specified period, of any nature or kind whatsoever under or by reason of this Agreement.

- (v) Canadian Salaried Plan. Except as set forth on SCHEDULE 5.7(v) (which Schedule shall be delivered on or prior to the Closing), the Oliver Rubber Employee who participates in the Canadian Salaried Plan shall, immediately prior to Closing, cease to participate in and accrue benefits under all applicable Company Foreign Benefit Plans, including, for greater certainty, the Canadian Salaried Plan. The Sellers shall provide the Buyer (or cause to be provided to the Buyer) such information as may be reasonably requested by Buyer concerning service by the Oliver Rubber Employee with the Seller or Affiliate of the Seller, as applicable, for the purposes of calculating benefits payable to such Oliver Rubber Employee under the Canadian Salaried Plan, including without limitation, advice concerning the date the Oliver Rubber Employee retires or has his employment terminated by the Seller or Affiliate of the Seller, as applicable, and such advice is to be provided within five Business Days of such Oliver Rubber Employee's retirement or termination date; further, Sellers agree to indemnify Buyer and any applicable Sold Company or Subsidiary for any costs the Buyer or any applicable Sold Company or Subsidiary may suffer or incur (including additional funding costs and reasonable legal and actuarial fees) as a result of any required or proposed partial wind-up of the Canadian Salaried Plan that applies in relation to any pension entitlements of any Canadian Salaried Plan member where such partial wind-up is required or proposed by Sellers or a Governmental Authority as a result of action taken by Sellers prior to the Closing Date, including, without limitation, the termination of the Oliver Rubber Employee.
- 5.8 LABOR MATTERS. (a) The Buyer agrees that from and after the Closing the Sold Companies, the Subsidiaries and the Venture Entities shall be responsible for any notification required under the WARN Act or the Employment Standards Act, Ontario with respect to the Sold Companies, the Subsidiaries and the Venture Entities with respect to employment losses which occur after the Closing Date; provided that Buyer shall not cause any employment losses with respect to the Sold Companies and Subsidiaries within the ninety (90) days (16 weeks in Ontario) following the Closing where such employment losses would give rise to an obligation of the Sellers or the Sold Companies, Subsidiaries and the Venture Entities to have given any notification required under the WARN Act with the ninety (90) days (or the Employment Standards Act, Ontario with the sixteen
- (16) weeks) preceding the Closing. Each Seller agrees that between the date hereof and the Closing Date or the earlier termination of this Agreement in accordance with its terms, it will cause the Sold Companies and the Subsidiaries and with respect to the Venture Entities, to the extent the Sellers have the authority and control to do so, not to effect or permit a "plant closing" or "mass layoff" as these terms are defined in the WARN Act or "mass termination" as defined in the Employment Standards Act, Ontario with respect to any member of the Sold Companies, the Subsidiaries and the Venture Entities without notifying the Buyer in advance and without complying with the notice requirements and all other provisions of the WARN Act or the Employment Standards Act, Ontario. The Sellers will also notify the Buyer of, and obtain its consent to, prior to the Closing, all layoffs and terminations at any "single site of employment" or "facility or operating unit within a single site of employment" that occur within ninety (90) days of the Closing and that, in the aggregate, exceed twenty percent (20%) of the workforce or 25 or more employees at either the "single site of employment" or a "facility or operating unit", which consent shall not be unreasonably withheld. The Buyer will take no action or make any statement prior to Closing that will obligate the Sellers to provide a group of employees with WARN notification or notice under the Employment Standards Act, Ontario.
- (b) The Sellers and the Buyer shall reasonably cooperate in connection with any required notification to, or any required consultation with, or the provision of documents and information to, the employees, employee representatives, work councils, unions, labor boards and relevant government agencies and governmental officials concerning the transactions contemplated by this Agreement with respect to non-U.S. Employees of any of the Sold Companies, the Subsidiaries and the Venture Entities so that such Persons may render advice as required in accordance with Law.

(c) The parties agree that, in connection with the sale of stock pursuant to this Agreement (the "TRANSACTION"), the Sellers, Buyer and Sold Companies, Subsidiaries and Venture Entities shall comply with any and all obligations imposed under the National Labor Relations Act (the "NLRA"). With respect to each bargaining unit in which employees of the Sold Companies, Subsidiaries or Venture Entities are represented by a labor organization or labor organizations, Buyer agrees that the Sold Companies, Subsidiaries and Venture Entities as a result of the Transaction will satisfy the following requirements at the earliest time permitted by applicable Law: (i) to recognize the labor organization as the bargaining representative for the employees within the existing bargaining unit; and (ii) to comply with, for the remainder of its stated term, the existing collective bargaining agreement between each such labor organization and the respective Sold Company, Subsidiary or Venture Entity, or to be bound by any other collective bargaining agreement(s) negotiated and entered into between the labor organization and the Sold Company, Subsidiary or Venture Entity establishing the terms and conditions of employment for bargaining unit employees. Should a dispute arise between any labor organization(s) and the Sellers, the Buyer and/or any Sold Companies, Subsidiaries or Venture Entities concerning compliance with any sale, successorship or similar provision contained in any existing collective bargaining agreement, the Sellers and/or the Buyer (whichever is appropriate) agree that the respective Sold Company, Subsidiary or Venture Entity will process the dispute under any applicable grievance and arbitration provisions of the collective bargaining agreement covering the dispute, provided that the dispute is the subject of a valid grievance filed in accordance with the terms of the applicable collective bargaining agreement. This SECTION 5.8(c) shall not prevent the Sellers, Buyer or any Sold Company, Subsidiary or Venture Entity from initiating or defending against any proceedings at any time before the NLRB or any court concerning any such dispute or other matters properly within the jurisdiction of the NLRB or court; nor shall this SECTION 5.8(c) require the arbitration of any grievance that is not arbitrable under the applicable collective bargaining agreement or require the Buyer and/or any Sold Companies, Subsidiaries or Venture Entities, contrary to the parties' understanding of applicable Law, to recognize any labor organization or comply with or be bound by any collective bargaining agreement prior to the Closing Date. The Sellers and Buyer further agree that the Sellers will provide notice and the opportunity for bargaining with any labor organization(s) that represent employees of the Sold Companies, Subsidiaries or Venture Entities as may be required under applicable Law and nothing in this agreement shall prevent or preclude such bargaining from taking place in compliance with applicable Law. The phrase "existing collective bargaining agreement" for purposes of this SECTION 5.8(c) shall mean those collective bargaining agreements as set forth in SCHEDULE 3.12. The requirements of this SECTION 5.8(c) shall apply to Venture Entities only to the extent that Buyer controls such entities.

5.9 CONTACT WITH CUSTOMERS AND SUPPLIERS. Prior to the Closing or the earlier termination of this Agreement in accordance with its terms, the Buyer and its representatives may contact and communicate with the Company Employees (other than the executive officers), customers, suppliers and licensors of the Sold Companies, the Subsidiaries and the Venture Entities in connection with the transactions contemplated hereby only with the prior written consent of the Sellers, which may not be unreasonably withheld, delayed or conditioned, except that consent may be conditioned upon a designee of the Sellers being present at any such meeting or conference.

5.10 INTERCOMPANY DEBT. On or prior to the day before the Closing,

(a) each of the Sold Companies that has positive Net Intercompany Amounts shall distribute such Net Intercompany Amounts to its shareholders and (b) the Sellers shall, and shall cause their Affiliates to, eliminate any Net Intercompany Amounts of each of the Sold Companies that has negative Net Intercompany Amounts and such elimination shall be treated as a contribution of capital to such Sold Company by its parent with the result that there will be no intercompany receivables or intercompany payables between the Sold Companies, the Subsidiaries or the Venture Entities, on the one hand, and the Sellers or their Affiliates (other than the Sold Companies, the Subsidiaries and the Venture Entities), on the other hand, at the Closing Date. For the avoidance of doubt, any intercompany receivables or payables between the Sold

Companies, the Subsidiaries and/or the Venture Entities shall remain in place and not be cancelled. As used herein, "NET INTERCOMPANY AMOUNTS" means with respect to each Sold Company (i) all intercompany receivables due to such Sold Company, its Subsidiaries and the Venture Entities from the Sellers and their Affiliates (other than the Sold Companies, the Subsidiaries and the Venture Entities), less (ii) all intercompany payables of such Sold Company, its Subsidiaries and the Venture Entities to the Sellers and their Affiliates (other than the Sold Companies, the Subsidiaries and the Venture Entities). As soon as practicable after the date of this Agreement, Sellers shall deliver to Buyer a schedule that sets forth the approximate amount of the intercompany receivables or intercompany payables between a Sold Company, Subsidiary or Venture Entity, on the one hand, and another Sold Company, Subsidiary or Venture Entity, on the other hand, all as of the most recent practicable date.

- 5.11 MINIMUM CASH REQUIREMENT. Sellers shall ensure that, as of the Closing, each of the Sold Companies and the Subsidiaries will maintain sufficient cash to satisfy any requirements of Law.
- 5.12 CORPORATE NAMES. Except as otherwise set forth in this SECTION 5.12, from and after the Closing, the Buyer shall cause the Sold Companies and the Subsidiaries to remove or cover the name "Cooper Tire," "Cooper," "Cooper-Standard" and any trademarks, trade names, brandmarks, brand names, trade dress or logos confusingly similar to such name (the "RETAINED NAMES") from all signs, billboards, advertising materials, telephone listings, labels, stationery, office forms, packaging or other materials of the Sold Companies and the Subsidiaries (other than materials for purely internal circulation, for which removal or coverage is impracticable). Except as set forth in this SECTION 5.12, the Buyer shall neither use nor permit any of the Sold Companies or the Subsidiaries to use the Retained Names in connection with the businesses of the Sold Companies and the Subsidiaries or otherwise, except for neutral, non-trademark use to describe the history of the Business or as required by Law.
- (b) Notwithstanding anything to the contrary in SECTION 5.12(a), the Sellers hereby grant to the Buyer an exclusive worldwide, royalty-free license to use (i) "Cooper-Standard" and the "Cooper-Standard" logo and accompanying goodwill with respect to such Retained Names listed on SCHEDULE 5.12 (collectively, the "TRADEMARKS") as or as part of a trademark, service mark, domain name, trade name, corporate name, logo or other source-indicator and (ii) the name of Cooper SaiYang Wuhu Automotive Co., Ltd. as the corporate name of the 60%owned Chinese subsidiary of CSA (Barbados) Investment Co. Ltd. from the Closing Date until the earlier of (A) the second anniversary of the Closing Date, or (B) an initial public offering of the equity interests of the Business. The Buyer may sublicense the above rights as necessary exclusively to the Sold Companies and the Subsidiaries, and may assign the above rights to any purchaser or acquiror of the Business in any change-in-control or similar transaction, whether by a merger, sale, restructuring or otherwise. During the term of the license (A) Sellers will not (1) license or sell the Trademarks to any other Person (except in connection with a Change of Control of Cooper) or (2) license or sell (except in connection with a Change of Control of Cooper) the name "Cooper" to a Competitive Business other than, subject to SECTION 5.18 of this Agreement, any joint venture or similar partnership or entity to which either Seller or one of their Affiliates is a partner or similar participant so long as such joint venture or similar partnership or entity does not use the "Cooper" name as part of a Competitive Business, and (B) neither Sellers nor any of their Affiliates will use the Trademarks in connection with a Competitive Business; provided that nothing shall prohibit Sellers and their Affiliates from using the Trademarks in a neutral, non-trademark manner to describe the history of the Business or as required by Law. After this license expires, the Buyer, the Sold Companies and the Subsidiaries shall not use the Trademarks in connection with their businesses or otherwise, except for neutral, non-trademark use to describe the history of the Business or as required by Law.
- (c) Buyer and Sellers agree to report to the other any acts of infringement of the Trademarks by a third party. Buyer and Sellers further agree to cooperate in either party's efforts to cease

any such infringing activity. The Buyer acknowledges that the Sellers have the first right to enforce any and all rights relating to the Trademarks at Sellers' sole cost and expense; provided, that if (i) Sellers take any actions to enforce their rights to the Trademarks at the request of Buyer or, (ii) Sellers' actions to enforce their rights to the Trademarks solely benefit the Buyer, then Buyer will be responsible for all expenses incurred in connection therewith, except if the Proceeding is the result of or relates to a breach by the Sellers of any of the representations and warranties contained in SECTION 3.13. Sellers acknowledge that if Sellers choose not to enforce such rights, Buyer may do so; provided, that Buyer may not take any action that would impair Sellers' rights to the Trademarks or any portion thereof or otherwise bind Sellers without the prior written consent of Sellers. If Buyer enforces such rights, Buyer will be responsible for all expenses and fees incurred in connection therewith, except if the Proceeding is the result of or relates to a breach by Sellers of any of the representations and warranties contained in SECTION 3.13. Sellers and Buyer each agree to cooperate with the other as necessary in any such Proceeding.

- (d) The Buyer acknowledges that in order to preserve the distinctiveness and goodwill of the Trademarks, Sellers must have the right to ensure that the Buyer's use of the Trademarks meets minimum standards of quality. The Buyer shall at all times and in all respects continue to use the Trademarks consistent with the past practices of Sellers, including producing goods of substantially the same quality as currently produced, so as not to injure the Trademarks or the goodwill associated therewith, and any use consistent therewith is hereby deemed approved without submission to Sellers. The Buyer further agrees to furnish to the Sellers, at reasonable intervals and upon written request, any documents and/or information reasonably necessary to effectuate its review to determine that the standards of quality control have been consistent with the terms of this SECTION 5.12(d). In the event that the Sellers determine that any of the standards are not met, it shall provide notice to the Buyer and a reasonable opportunity to cure such defects.
- (e) For 180 days from the Closing Date, the Sellers will, in a reasonable manner on their websites, (i) inform all Internet users as to the Buyer's new ownership of the Business and (ii) direct interested Internet users to a website of the Buyer's selection. The parties will cooperate in good faith to (x) transfer to the Buyer and (y) allow the Buyer to use, in each case in accordance with applicable Law and the Sellers' privacy policy, all content and information collected from or used in connection with the Sellers' websites with respect to the Business.
- (f) Each of the parties hereto acknowledges and agrees that the monetary damages for any material breach of the requirements of SECTION 5.12 would be inadequate, and agrees and consents that without intending to limit any additional remedies that may be available, temporary and permanent injunctive and other equitable relief may be granted without proof of actual damage or inadequacy of legal remedy in any Proceeding that may be brought to enforce any of the provisions of SECTION 5.12.
- 5.13 RESTRUCTURING. Prior to the Closing, except as otherwise modified pursuant to SECTION 2.6 hereof, the Sellers shall cause the transactions described in SCHEDULE B (the "RESTRUCTURING") not yet completed as of the date of this Agreement to be consummated in a manner reasonably satisfactory to Buyer, and shall obtain any required written consent of any third party under, and comply with all provisions of, each applicable collective bargaining agreement or Law with respect to such Restructuring.
- 5.14 CONSENTS AND COMPETITION. (a) The Buyer and the Sellers shall,
- (i) as promptly as practicable, but in no event later than ten (10) Business Days following the date of this Agreement, within the applicable filing deadlines, submit all initial filings required by the HSR Act, EC Merger Regulation, the Canadian Competition Act and the Australian Foreign Acquisition and Takeover Act and (ii) as promptly as practicable, within the applicable filing deadlines, submit all initial filings required by or any other applicable Competition/Investment Law, and thereafter provide any supplemental

information requested in connection therewith. The Buyer and the Sellers shall furnish to the other such necessary information and reasonable assistance as the other may request in connection with its preparation of any filing or submission that is necessary under the HSR Act, EC Merger Regulation or any other applicable Competition/Investment Law. The Buyer and the Sellers shall request early termination of the applicable waiting period under the HSR Act, EC Merger Regulation or any other applicable Competition/Investment Law. The Buyer and the Sellers will promptly inform the other party of any material communication received by such party from any Governmental Authority in respect of any filing under the HSR Act, EC Merger Regulation or any other applicable Competition/Investment Law. Each of the parties will (a) use its respective commercially reasonable efforts to comply as expeditiously as possible with all requests of any Governmental Authority for additional information and documents, including, without limitation, information or documents requested under the HSR Act, the EC Merger Regulation or any other applicable Competition/Investment Law; (b) not (i) extend any waiting period under the HSR Act, the EC Merger Regulation or any other applicable Competition/Investment Law or (ii) enter into any agreement with any Governmental Authority not to consummate the transactions contemplated by this Agreement, except with the prior consent of the other parties; and (c) cooperate with the other parties and use commercially reasonable efforts to contest and resist any action, including legislative, administrative or judicial action, and to have vacated, lifted, reversed or overturned any Governmental Order (whether temporary, preliminary or permanent) that restricts, prevents or prohibits the consummation of the transactions contemplated by this Agreement. Without limiting the generality of the foregoing, the Buyer agrees to propose, negotiate and cooperate with the Sellers to effect, by consent decree, hold separate order or otherwise, the sale, divestiture or disposition of such assets or businesses of any of the Sold Companies or the Subsidiaries (or otherwise take any action that limits the freedom of action with respect to, or its ability to retain, any of the businesses, product lines, or assets of any of the Sold Companies or the Subsidiaries), so long as conditioned upon, and not occurring prior to, the Closing, as may be required in order to avoid the entry of, or to effect the dissolution of, any Governmental Order (whether temporary, preliminary or permanent) that would otherwise have the effect of preventing or delaying the consummation of the transactions contemplated hereby; provided that Buyer shall not be required to take any such actions to the extent it would have a material adverse effect on the Business after Closing; and provided further, that any sale, divestiture or disposition of such assets or businesses be on terms reasonably satisfactory to the Buyer.

(b) The Sellers shall use commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or advisable to obtain all third-party authorizations, approvals, consents or waivers in order to consummate the transactions contemplated hereby, including securing consents or waivers of third parties listed in SCHEDULE 2.5(a)(vi). Buyer shall use its commercially reasonable efforts to assist the Sellers in obtaining the consents or waivers of third parties listed in SCHEDULE 2.5(a)(vi).

5.15 INTENTIONALLY OMITTED.

5.16 INSURANCE. From and after the Closing Date, the Sellers shall use their commercially reasonable efforts, subject to the terms of the Insurance Policies, to retain the right to make claims and receive recoveries for the benefit of Buyer and the Sold Companies and Subsidiaries under any insurance policies maintained at any time prior to the Closing Date by the Sellers, their subsidiaries, the Sold Companies and Subsidiaries and/or their respective predecessors, covering any loss, liability, claim, damage or expense relating to the Business (including products and current and former employees and directors) arising out of occurrences prior to the Closing. Each Seller agrees to use its commercially reasonable efforts so that each of Buyer, the Sold Companies and the Subsidiaries shall have the right, power and authority in its own name or in the name of any Seller or any of their subsidiaries to make directly any claims under any such policies and to receive directly recoveries thereunder. Upon request of the Buyer, Seller shall provide historical loss information with respect to the Insurance Policies.

- 5.17 NOTICE OF EVENTS. (a) During the period from the date of this Agreement to the Closing Date or the earlier termination of this Agreement in accordance with its terms, the Buyer shall give prompt written notice to the Sellers of (i) the occurrence or non-occurrence of any event that has caused, or could reasonably be expected to cause, any representation or warranty made by it to be untrue or inaccurate in any material respect at any time after the date of this Agreement and prior to the Closing Date; and (ii) any material failure on its part to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder. In addition, the Buyer shall promptly notify the Sellers if the Buyer obtains knowledge that any of the representations and warranties of the Sellers in this Agreement and the schedules hereto (including the updated schedules) are not true and correct in all material respects, or if the Buyer obtains knowledge of any material errors in, or omissions from, the schedules (including the updated schedules) to this Agreement.
- (b) During the period from the date of this Agreement to the Closing Date or the earlier termination of this Agreement in accordance with its terms, the Sellers shall give prompt written notice to the Buyer if they become aware of (i) the occurrence or non-occurrence of any event to the Knowledge of the Sellers which has caused, or could reasonably be expected to cause, any representation or warranty made by the Sellers to be untrue or inaccurate in any material respect at any time after the date of this Agreement and prior to the Closing Date; and (ii) any material failure on the Sellers part to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder. If any such event requires any change to the Sellers' schedules to this Agreement with respect to ARTICLE III, the Sellers shall promptly deliver to the Buyer a supplement to the schedules specifying such change; provided that any such supplement shall not be taken into account for purposes of determining whether the conditions precedent in ARTICLE VI are satisfied or for purposes of indemnification as set forth in ARTICLE VIII (except as set forth below); and provided further, that if any supplement reflects any state of facts, change, occurrence or event that, individually or in the aggregate, would result in the failure of any condition precedent set forth in ARTICLE VI, Buyer may, at its option, cause this Agreement to be terminated without any liability to Buyer by giving written notice to Sellers upon the earlier of (i) a date within five (5) days after receipt of the supplement to the schedule and (ii) the Closing Date. If Buyer has a right to terminate this Agreement under this Section and elects not to exercise such right, and the Closing occurs, this Agreement shall be deemed to have been amended (including for purposes of indemnification as set forth in ARTICLE VIII) by supplementing the applicable schedule. Notwithstanding anything to the contrary in this SECTION 5.17, Buyer acknowledges and agrees that final SCHEDULES 5.7(e), 5.7(f), 5.7(j) and 5.7(v) will not be delivered or finalized until on or prior to the Closing. Buyer will not have the right to terminate this Agreement as the result of the delivery and updating of such schedules referred to in the preceding sentence and this Agreement shall be deemed to have been amended (including for the purposes of indemnification as set forth in ARTICLE VIII) by the updating and delivery of such schedules.
- 5.18 NONCOMPETITION. Each Seller covenants and agrees that for a period of the earlier of (i) three (3) years following the Closing Date or (ii) a Change of Control of Cooper, it shall not, and shall cause its subsidiaries not to indirectly or directly own, manage, operate, control or participate in the ownership, management, operation or control of or be otherwise connected in any material matter with any entity engaged in the manufacture or sale of products that are within the scope of the Business (a "COMPETITIVE BUSINESS"), whether as employer, proprietor, partner, stockholder, consultant, agent, lender or guarantor or otherwise.
- (b) Notwithstanding anything to the contrary contained in SECTION 5.18(a):
- (i) Sellers may directly or indirectly hold securities listed on a recognized stock exchange or automated quotation system of any Person to the extent that such investment does

not directly or indirectly confer on Seller more than five percent (5%) of the voting power with respect to, or interest in the profits of, such Person;

- (ii) Sellers may acquire interests in or securities of any Person that, together with Sellers, derived 25% or less of their combined total annual revenues in their most recent fiscal year from activities that constitute a Competitive Businesses;
- (iii) Sellers may acquire or use any product for internal uses or to conduct Sellers' or their subsidiaries' other businesses that consume, use, contain, depend upon or otherwise incorporate any such product; and
- (iv) Seller may perform any act or conduct any business expressly contemplated hereby or the Transaction Agreements.
- (c) Each of the parties hereto acknowledges and agrees that the monetary damages for any material breach of the requirements of SECTIONS 5.18(a) or (b) would be inadequate, and agrees and consents that without intending to limit any additional remedies that may be available, temporary and permanent injunctive and other equitable relief may be granted without proof of actual damage or inadequacy of legal remedy in any Proceeding that may be brought to enforce any of the provisions of SECTION 5.18(a) or (b).
- 5.19 NON-SOLICITATION AND NON-HIRE OF EMPLOYEES. (i) Each Seller covenants and agrees that for a period of three (3) years following the Closing Date it shall not, and shall cause its subsidiaries and controlled Affiliates not to, directly or indirectly, solicit to hire or hire, or enter into a consulting agreement with any Company Employee who is an executive officer as of the Closing Date or who becomes an executive officer of any of the Sold Companies or the subsidiaries within one year of the Closing Date; provided, however, that the foregoing prohibition as to solicitations to hire shall not be deemed violated by any general solicitation to the public by Sellers or their subsidiaries or controlled Affiliates and (ii) Buyer covenants and agrees that for a period of three (3) years following the Closing Date it shall not, and shall cause the Sold Companies and its other subsidiaries and controlled Affiliates not to, directly or indirectly, solicit to hire or hire, or enter into a consulting agreement with any employee of any Seller who is an executive officer; provided, however, that the foregoing prohibition as to solicitations to hire shall not be deemed violated by (x) any general solicitation to the public by Buyers, the Sold Companies, the Subsidiaries and their respective controlled Affiliates or (y) Buyer's employment of James S. McElya and Seller hereby agrees that such employment shall not be deemed to violate any noncompetition covenants or agreements otherwise applicable to James S. McElya with respect to Sellers and their Affiliates. As used in this SECTION 5.19(a), "executive officer" shall have the meaning set forth in Rule 3b-7 of the Securities Exchange Act of 1934.
- (b) Each of the parties hereto acknowledges and agrees that the monetary damages for any material breach of the requirements of SECTION 5.19(a) would be inadequate, and agrees and consents that without intending to limit any additional remedies that may be available, temporary and permanent injunctive and other equitable relief may be granted without proof of actual damage or inadequacy of legal remedy in any Proceeding that may be brought to enforce any of the provisions of SECTION 5.19(a).
- 5.20 ACQUISITION FINANCING. Sellers agree to provide, and shall cause its subsidiaries and the Sold Companies and Subsidiaries and their respective representatives to provide, all cooperation reasonably necessary in connection with the arrangement of the Acquisition Financing, including (i) participation in meetings, drafting sessions, due diligence sessions, management presentation sessions, road shows and sessions with rating agencies, (ii) preparation of business projections, financial

statements, offering memoranda, private placement memoranda, prospectuses and similar documents and (iii) execution and delivery of any underwriting or placement agreements, pledge and security documents, other definitive financing documents, including any indemnity agreements, or other requested certificates or documents, including a certificate of the chief financial officers of Sellers and/or the Business with respect to solvency matters and legal opinions as may be reasonably requested by Buyer, provided that the documents so executed and delivered provide that they are not effective until the Closing. Seller shall take such further action as may be required to cause Ernst & Young LLP, the independent auditors of the Business and the Sold Companies and Subsidiaries, to provide any unqualified opinions, consents or customary comfort letters with respect to the Financial Statements. Sellers agree to allow Buyer's accounting representatives the opportunity to review the Financial Statements and to allow such representatives access to the Sold Companies' and Subsidiaries' and the Business' supporting documentation with respect to the preparation of the Financial Statements and the independent auditors' working papers relating to procedures performed relating to the Financial Statements.

- 5.21 LETTERS OF CREDIT ON ACCOUNT OF SOLD COMPANIES OR SUBSIDIARIES. Cooper agrees that it will continue to maintain all letters of credit Cooper has for and in support of workers' compensation, workers entitlement guarantees or employers' liability obligations and arrangements of the Sold Companies and Subsidiaries after the Closing Date until the expiration of such letters of credit.
- 5.22 GUARANTEES. Prior to the Closing, Buyer shall use its commercially reasonable efforts to assist Sellers in obtaining releases with respect to the guarantees set forth as items 4, 5, 6, 9, 10, 14, 20 and 21 on SCHEDULE 3.6(b); provided, however, that such efforts shall not require Buyer to make any payment to the beneficiary of such guarantees to obtain such releases. To the extent Cooper is not released from its obligations under such guarantees at or prior to the Closing, Buyer shall indemnify Cooper for any amounts due after the Closing pursuant to SECTION 8.2(a)(iii).
- 5.23 BUYER FINANCING. Buyer shall use its commercially reasonable efforts to obtain the Acquisition Financing on the terms and conditions described in the Financing Commitments, including using commercially reasonable efforts (i) to negotiate definitive agreements with respect thereto on the terms and conditions contained in the Financing Commitments and (ii) to satisfy all conditions applicable to Buyer in such definitive agreements.

ARTICLE VI CONDITIONS TO CLOSING

- 6.1 CONDITIONS TO THE OBLIGATIONS OF THE PARTIES. The respective obligations of the Buyer and the Sellers to consummate the transactions contemplated by this Agreement are subject to the satisfaction or waiver (if permitted by applicable Law) at or prior to the Closing of each of the following conditions:
- (a) None of the parties hereto will be subject to any Governmental Order or Law that restrains or prohibits the consummation of the transactions contemplated by this Agreement (a "RESTRAINT"). If any such Restraint has been issued, enacted, enforced or promulgated by a Governmental Authority, each party shall use its commercially reasonable efforts to have any such Restraint overturned or lifted.
- (b) The waiting period (including any extension thereof) applicable to the consummation of the transactions contemplated by this Agreement under the HSR Act, EC Merger Regulation or any other applicable Competition/Investment Law will have expired or been terminated.

- 6.2 CONDITIONS TO THE OBLIGATION OF THE SELLERS. The obligation of the Sellers to consummate the transactions contemplated by this Agreement is subject to the satisfaction or waiver (if permitted by applicable Law) at or prior to the Closing of each of the following conditions:
- (a) The representations and warranties made by the Buyer in this Agreement will be true and correct in all material respects (provided that any representation or warranty of the Buyer contained herein that is subject to a materiality, material adverse effect or similar qualification will not be so qualified for purposes of determining the existence of any breach thereof on the part of the Buyer) as of the Closing Date as though made on and as of such date (except to the extent such representations and warranties speak as of an earlier date, in which case such representations and warranties shall be, subject to the qualifications set forth above, true and correct as of such earlier date).
- (b) The Buyer will have performed and complied in all material respects with all agreements and obligations required by this Agreement to be so performed or complied with by it at or prior to the Closing.
- (c) The Buyer will have delivered to the Sellers a certificate, dated as of the Closing Date and executed by an executive officer of the Buyer, certifying to the fulfillment of the conditions specified in SECTIONS 6.2(a) and 6.2(b).
- (d) The Buyer will have delivered, or caused to be delivered, all of the closing deliveries required by SECTION 2.5(b).
- 6.3 CONDITIONS TO THE OBLIGATIONS OF THE BUYER. The obligation of the Buyer to consummate the transactions contemplated by this Agreement is subject to the satisfaction or waiver (if permitted by applicable Laws) at or prior to the Closing of each of the following conditions:
- (a) The representations and warranties made by the Sellers in this Agreement will be true and correct in all material respects (provided that (i) any representation or warranty (other than those set forth in SECTION 3.9(c)(ii)) of the Sellers contained herein that is subject to a materiality, Sold Company Material Adverse Effect or similar qualification will not be so qualified for purposes of determining the existence of any breach thereof on the part of the Sellers and (ii) SECTION 3.9(c)(ii) will be true and correct in all respects) as of the Closing Date as though made on and as of such date (except to the extent such representations and warranties speak as of an earlier date, in which case such representations and warranties shall be, subject to the qualifications set forth above, true and correct as of such earlier date).
- (b) The Sellers will have performed and complied in all material respects with all agreements and obligations required by this Agreement to be performed or complied with by them at or prior to the Closing.
- (c) Each of the Sellers will have delivered to the Buyer a certificate, dated as of the Closing Date and executed by an executive officer of each such Seller, certifying to the fulfillment of the conditions specified in SECTIONS 6.3(a) and 6.3(b).
- (d) Cooper will have delivered, or caused to be delivered, all of the closing deliveries required by SECTION 2.5(a).
- (e) Buyer shall have received the proceeds of the Debt Financing on the terms and conditions set forth in the Debt Financing Commitment or upon terms and conditions which are, in the

reasonable judgment of Buyer, comparable or more favorable (to Buyer) in the aggregate, and to the extent that any terms and conditions are not set forth in the Debt Financing Commitment, on terms and conditions reasonably satisfactory to Buyer.

(f) Since the date of this Agreement, the Business, Sold Companies and Subsidiaries shall not have become subject to any additional regulation of any Governmental Authority or become subject to the jurisdiction or regulation of additional Governmental Authorities as a result of the Restructuring.

ARTICLE VII TERMINATION

- 7.1 TERMINATION. This Agreement may be terminated at any time prior to the Closing Date:
- (a) by the mutual written consent of Cooper and the Buyer;
- (b) by Cooper or the Buyer, upon written notice to the other party, if the Closing has not occurred on or before December 15, 2004 (the "OUTSIDE DATE"), unless the failure of such consummation is due to the failure of the party wishing to terminate to comply in all material respects with the agreements and covenants contained herein;
- (c) by Cooper or the Buyer, upon written notice to the other party, if any Restraint is in effect and has become final and nonappealable, provided that the party seeking to terminate this Agreement pursuant to this SECTION 7.1(c) has used its commercially reasonable efforts to remove such Restraint;
- (d) by the Buyer, if any Seller materially breaches any of its representations, warranties or obligations under this Agreement and such breach is not cured within thirty (30) days after written notice to the Sellers by the Buyer; or
- (e) by Cooper, if the Buyer materially breaches any of its representations, warranties or obligations under this Agreement and such breach is not cured within thirty (30) days after written notice to the Buyer by Cooper.
- 7.2 EFFECT OF TERMINATION. In the event of termination of this Agreement pursuant to SECTION 7.1, this Agreement will become void and have no effect, and no party will have any liability or any further obligation to any other party, except as provided in this SECTION 7.2 and except that nothing herein releases, or may be construed as releasing, any party hereto from any liability or damage to any other party hereto arising out of the breaching party's willful breach of its representations and warranties or breach in the performance of any of its covenants, agreements, duties or obligations arising under this Agreement. The obligations of the parties to this Agreement under SECTION 5.2 (as such Section relates to confidentiality), SECTION 7.2 and ARTICLE IX will survive any termination of this Agreement.

ARTICLE VIII INDEMNIFICATION

8.1 INDEMNIFICATION BY THE SELLERS. Subject to the terms and conditions set forth in this ARTICLE VIII, from and after the Closing, the Sellers shall, jointly and severally, indemnify and hold harmless the Buyer, its Affiliates (including, after the Closing, the Sold Companies, the

Subsidiaries and the Venture Entities) and their respective officers, directors, shareholders, employees, agents and representatives (collectively, the "BUYER INDEMNIFIED PERSONS") from and against any and all Losses incurred or suffered by any Buyer Indemnified Person arising out of or related to (i) any breach of any representation or warranty by any Seller contained in this Agreement (reading such representation or warranty without regard to materiality, Sold Company Material Adverse Effect or similar qualification) whether made upon execution of this Agreement or at Closing as contemplated by SECTIONS 6.3(a) and the certificate to be delivered at Closing pursuant to SECTION 6.3(a), (ii) any non-fulfillment or breach by any Seller of any covenant, obligation or agreement hereunder, (iii) in respect of any Seller Benefit Plan as a result of the acts or omissions of Sellers or any of their subsidiaries or Affiliates in connection with the administration of such Seller Benefit Plans after the Closing Date and before the transfer of assets and liabilities contemplated by SECTIONS 5.7(e), 5.7(f) or 5.7(g), (iv) payments to Company Employees, including former employees, triggered in whole or in part by the transactions contemplated hereby (whether alone or in connection with other events) except to the extent liability for such payment is expressly assumed by Buyer under SECTION 5.7, (v) the Schenectady Facility Matter, (vi) (A) any Release or threatened Release, transporting, or arranging for disposal of Hazardous Materials or any violation of Environmental Law at or resulting from operations of any property or facility owned or operated by the Sold Companies or the Subsidiaries at any time prior to, but not on or after, the Closing Date, (B) Liabilities arising out of the disposal or arranging for disposal, prior to the Closing Date, of Hazardous Materials generated or otherwise originating at the Owned Real Property or the Leased Real Property at or to a location other than the Owned Real Property or the Leased Real Property (C) Liabilities arising from death, personal injury, or property damage relating to, resulting from, caused by or arising out of, directly or indirectly, the actual or alleged presence of asbestos or asbestos containing materials in any of the products (or any part or component) designed, manufactured, serviced or sold, or services performed by any Sold Company or Subsidiary or the Business, in each case to the extent such products or services were designed, manufactured, serviced, sold or performed prior to the Closing, including any such Liabilities for negligence, workers' compensation, employers' liability, strict liability, design (but not with respect to products designed before the Closing and not manufactured as of the Closing Date that are manufactured after the Closing) or manufacturing defect, conspiracy, failure to warn, or breach of express or implied warranties of merchantability or fitness for any purpose or use or (D) Liabilities relating to Environmental Laws to the extent arising from the matters set forth on SCHEDULE 8.1(a)(vi) (D), (vii) any matter that, but for the consummation of the transactions contemplated under this Agreement or any action or omission of the Sellers or their Affiliates (other than the Sold Companies after the Closing Date), would have been subject to indemnification pursuant to the Share and Business Sale and Purchase Agreement, dated as of November 22, 1999 including Schedule 15 thereof (the "SIEBE PURCHASE Agreement"), by and among Invensys plc, Cooper and the other parties named therein, but subject to the same survival period, monetary and other contractual limitations expressly set forth therein (except for assignment limitations) as if Buyer Indemnified Persons were parties thereto, (viii) any liabilities in respect of the Agreement and Plan of Merger, dated as of June 22, 2000, by and among Holm Parent, Inc., Holm Acquisition Company, The Standard Products Company and Holm Industries, Inc. and the Asset Purchase Agreement, dated as of April 20, 2000, by and between The Standard Products Company and Plastech Engineered Products, Inc., (ix) the Restructuring, including, without limitation, any severance, termination benefits or other employee benefit costs associated with the Restructuring, (x) any Losses resulting from Claims with respect to any Seller Benefit Plan, or liabilities thereunder, which are not expressly assumed by the Buyer pursuant to SECTION 5.7 or with respect to any Company Benefit Plans that are not disclosed on SCHEDULE 3.11(a)(i)-1 and for which costs were not expressly included in the income statements of the Sold Companies or Subsidiaries, (xi) obligations of the Sold Companies or Subsidiaries with respect to any repurchase of receivables or any other obligations under the Factoring Agreements as a result of any transactions under such factoring agreement prior to the Closing Date, (xii) any Tax, fine, lien, penalty or other liability imposed on any Sold Company or Subsidiary pursuant to Title IV of ERISA with respect to any defined benefit pension plan subject to Title IV of ERISA sponsored by Sellers or any member of their Controlled Group at any time on or prior to the

Closing Date, except with respect to the plans sponsored by a Sold Company or Subsidiary as of the Closing Date or plans covered by SECTION 5.7(e) and SECTION 5.7(g), (xiii) any Losses (including in respect of funding obligations) resulting from Claims in respect of Cooper's Pension Plan relating to its status as a cash balance plan and (xiv) any Losses related to claims that Retained Names (including the Trademarks) infringe on the intellectual property rights of a third party (excluding those Losses related to the Buyer's use of the Retained Names (including the Trademarks) in noncompliance with SECTIONS 5.12(a) and (B)).

- (b) None of the Buyer Indemnified Persons will be entitled to recover from the Sellers for any Losses indemnifiable pursuant to SECTION 8.1 (a)(i), unless and until the total of all Losses indemnifiable under SECTION 8.1(a)(i) exceeds \$11,650,000 and then only for the amount by which such Losses exceed such amount (the "DEDUCTIBLE"); provided that in calculating whether the Deductible has been obtained, only Losses under SECTION 8.1(a)(i) in excess of \$100,000 shall be considered. The Buyer Indemnified Persons will not be entitled to recover more than an aggregate of \$174,750,000 (the "MAXIMUM AMOUNT"), from the Sellers with respect to all Losses indemnifiable pursuant to SECTION 8.1(a)(i); provided that the foregoing limitations not apply to any obligation or liability under SECTION 8.1(a)(i) arising out of or related to a breach of any of the representations set forth in SECTIONS 3.2, 3.3, 3.4(a) (other than the second sentence thereof) and 3.25.
- 8.2 INDEMNIFICATION BY THE BUYER. Subject to the terms and conditions set forth in this ARTICLE VIII, from and after the Closing, the Buyer shall indemnify and hold harmless each of the Sellers, their respective Affiliates and their respective officers, directors, shareholders, employees, agents, and representatives (collectively, the "SELLER INDEMNIFIED PERSONS") from and against any and all Losses incurred or suffered by any Seller Indemnified Person arising out of or related to (i) any breach of any representation or warranty by Buyer contained in this Agreement (reading such representation or warranty without regard to materiality, material adverse effect or similar qualification) whether made upon execution of this Agreement or at Closing as contemplated by SECTION 6.2(a) and the certificate to be delivered at Closing pursuant to SECTION 6.2(c), (ii) any non-fulfillment or breach by Buyer of any covenant, obligation or agreement hereunder, (iii) any payments required to be made by the Sellers after the Closing under the guarantees set forth as items 4, 5, 6, 9, 10, 14, 20 and 21 on SCHEDULE 3.6(b) and (iv) obligations arising following the Closing under the letters of credit to be maintained by Cooper pursuant to SECTION 5.21.
- (b) None of the Seller Indemnified Persons will be entitled to recover from the Buyer for any Losses indemnifiable pursuant to SECTION 8.2 (a)(i), unless and until the total of all Losses indemnifiable under SECTION 8.2(a)(i) exceeds the Deductible, and then only for the amount by which such Losses exceed such amount. The Seller Indemnified Persons will not be entitled to recover more than the Maximum Amount from the Buyer with respect to all Losses indemnifiable pursuant to SECTION 8.2(a)(i); provided that the foregoing limitations not apply to any obligation or liability under SECTION 8.2(a)(i) arising out of or related to a breach of any of the representations set forth in SECTIONS 4.2, and 4.5.
- 8.3 INDEMNIFICATION AS EXCLUSIVE REMEDY. Except as otherwise provided below, the indemnification provided for in this ARTICLE VIII, subject to the limitations set forth herein, shall be the exclusive post-Closing remedy available to any party in connection with any Losses arising out of the matters set forth in this Agreement or the transactions contemplated hereunder; provided, however, that nothing herein will limit in any way any such party's remedies in respect of fraud by the other party in connection herewith or with any Transaction Agreement or the transactions contemplated hereby or thereby. Notwithstanding anything in this Agreement to the contrary, all matters relating to Taxes shall

be governed by SECTIONS 5.4, 5.5 and 5.6 hereof and no provision of this ARTICLE VIII shall apply to limit, modify or offset the rights or obligations of the parties thereunder.

8.4 SURVIVAL. Except as set forth below, all claims for indemnification under SECTION 8.1(a) must be asserted on or prior to the date of the termination of the respective survival periods set forth in this SECTION 8.4, except such claims may be pursued thereafter if written notice thereof (specifying in reasonable detail the basis for such claim) was duly given within such period. Claims for fraud may be asserted at any time prior to the expiration of the applicable statute of limitations. The representations and warranties contained in this Agreement will survive the Closing Date until the first anniversary of the Closing Date, except that the representations and warranties (a) set forth in SECTIONS 3.2, 3.3, 3.4(a) (other than the second sentence thereof), 3.25, 4.2, 4.3 and 4.5 shall survive indefinitely, (b) set forth in SECTION 3.10 shall survive as set forth in SECTION 5.5(e), (c) set forth in SECTION 3.11 shall survive the Closing until the third anniversary of the Closing Date, and (d) set forth in SECTION 3.15 shall survive the Closing until the fourth anniversary of the Closing Date. All claims for indemnification under subclauses (ii)-(xiv) of SECTION 8.1(a) may be asserted at any time; provided, however that all claims for indemnification under subclause (vi)(D) of

SECTION 8.1(a) must be asserted before the tenth anniversary of the Closing Date, after which date no claims for indemnity may be asserted under SECTION 8.1(a)(vi)(d).

- 8.5 LIMITATIONS ON INDEMNIFICATION. (a) The amount of any Losses for which indemnification is provided under SECTION 8.1(a) or SECTION 8.2(a) shall be computed net of any third party insurance proceeds received by, or payable to, an Indemnified Party in connection with such Losses (net of any expenses incurred by the Indemnified Party in obtaining such insurance proceeds including the cost of maintaining any insurance policy) and any Tax benefit actually realized by the Indemnified Party (net of any Tax detriment) in the Tax period in which the indemnification payment is made in respect of any Losses for which such indemnification payment is made. For the avoidance of doubt, such payments shall not be reduced for additions to net operating losses. The parties agree that any indemnification payments made pursuant to this Agreement shall be treated for tax purposes as an adjustment to the Final Purchase Price, unless otherwise required by Law.
- (b) Notwithstanding anything herein to the contrary, no party shall be liable to any Indemnified Party for (i) punitive or exemplary Losses or (ii) special, incidental, indirect or consequential Losses, except to the extent that applicable Law would permit a party to recover lost or anticipated profits of the Business that are reasonably foreseeable to Sellers; provided, however, that the foregoing limitations shall not apply to the extent that such Losses consist of a payment by the Indemnified Party in defense of or in respect of a Third Party Claim.
- (c) For matters covered by Schedule 15: Environmental Covenant to the Siebe Purchase Agreement Buyer Indemnified Persons to the extent such Buyer Indemnified Persons have direct rights to indemnification pursuant to Schedule 15: Environmental Covenant to the Siebe Purchase Agreement shall not be entitled to indemnification for Environmental Losses pursuant to SECTION 8.1(a) unless and until such Buyer Indemnified Persons have fully complied with the provisions of such Schedule 15 (it being understood that this condition shall not apply to any such noncompliance that first occurred prior to the Closing Date, except to the extent such noncompliance may be remedied following the Closing Date without any significant cost or adverse impact to the Buyer Indemnified Persons) and the relevant Buyer Indemnified Person has made all reasonable efforts to obtain indemnification under the Siebe Purchase Agreement and has failed to obtain such indemnification despite such efforts. Buyer shall use all commercially reasonable efforts to cause those Sold Companies and Subsidiaries that, as of the date of this Agreement, have rights to indemnification pursuant to Schedule 15 of the Siebe Purchase

Agreement to materially comply with all of their obligations under Schedule 15 of the Siebe Purchase Agreement.

- (d) Seller shall use all commercially reasonable efforts to maintain and to assign to Buyer any third-party indemnities, settlements or cost sharing agreements available or potentially available to Buyer.
- (e) Without limiting SECTION 8.5(c) of this Agreement or Schedule 15 of the Siebe Purchase Agreement in any way:
- (i) The Buyer Indemnified Persons shall cooperate with the Sellers and take all commercially reasonable actions to avoid and minimize Environmental Losses that would otherwise be subject to the indemnification under SECTION 8.1(a), including not soliciting or importuning any Governmental Authority to require any environmental correction, investigation, monitoring or remediation unless affirmatively required to do so by Law, including Environmental Laws or unless the Buyer Indemnified Persons reasonably believe that such actions will likely reduce the overall extent of such Losses (subject to Sellers' consent which shall not be unreasonably withheld or delayed). Buyer Indemnified Persons shall not conduct invasive investigations of the Real Property (including sampling of soil, surface water or ground water) for the sole purpose of identifying an Environmental Loss subject to the indemnification under SECTION 8.1(a).
- (ii) In addition to any other limitations on indemnification that may apply, with respect to any claim for indemnification any of the Buyer Indemnified Persons may assert regarding Environmental Laws or Hazardous Materials, none of the Sellers shall have any obligation with respect to such claim to the extent of the Losses for which indemnification is sought:
- (A) arise out of any breach of Buyer Indemnified Persons' covenants under this SECTION 8.5(e);
- (B) without limiting SECTION 8.5(e)(i), arise out of any action (to address any Release, threat of Release or violation of Environmental Law) not required by any Governmental Authority or under Law, including Environmental Law or any action to meet a cleanup standard under Environmental Law that is more stringent or costly than necessary for the continued use of any property as it was similarly used by the Sold Companies, the Subsidiaries or the Venture Entities prior to the Closing Date, unless the Buyer Indemnified Persons agree to reimburse the Sellers for costs necessary to meet such cleanup standard;
- (C) are ordinary costs of any post-Closing construction, demolition, renovation or maintenance of facilities on the Owned Real Property by a Buyer Indemnified Party, including any asbestos abatement obligations arising from such activities, but not including Losses that arise from a Release or threatened Release of Hazardous Materials or violation of Environmental Law relating thereto, that exists as of, or occurred or existed prior to, the Closing Date; or
- (D) are, pursuant to any written agreement under which any of the Sold Companies, the Subsidiaries or the Venture Entities are entitled to assert a claim, subject to indemnification by any other Person, except where a Buyer, a Sold Company, a

Subsidiary or a Venture Entity has in good faith sought and failed to obtain such indemnification.

- (iii) In addition, with respect to any claim that is the subject of this SECTION 8.5(e):
- (A) Buyer Indemnified Persons shall notify the Sellers of any claim for indemnification under SECTION 8.1(a) and afford the Sellers the opportunity to evaluate the conditions giving rise to such claim prior to incurring substantial costs with respect to such claim for which it may seek indemnification; provided, however, that in the event of an emergency that presents an imminent likelihood of personal injury or property damage, the Buyer Indemnified Parties may take action to address such emergency immediately, but must provide verbal notice to Sellers as soon as possible with written follow-up notice promptly thereafter; however, Sellers may exercise their rights under SECTION 8.5(e)(iii)(B) as soon as feasible in the event of an emergency;
- (B) the Sellers, at their sole expense, shall be entitled, but not obligated, to undertake, with the Buyer Indemnified Persons' oversight, participation and approval (not to be unreasonably withheld or delayed or to unreasonably result in material increased costs beyond those required by Law or Environmental Law), any investigation, remediation or other action required by Environmental Law (and any negotiation with Governmental Authorities regarding same) with respect to such matter, using all commercially reasonable efforts to avoid any interference with operations of the Buyer Indemnified Persons, and the Buyer Indemnified Persons shall afford the Sellers reasonable access to the relevant Real Property to undertake any such investigation, remediation or other action, subject, where such property is Leased Real Property, to the requirements of the relevant landlord; provided, however, that the Sellers shall promptly repair, restore or reimburse the Buyer Indemnified Persons for any damages or other costs incurred by the Buyer Indemnified Persons as a result of providing access under this SECTION 8.5(b):
- (C) if Sellers do not assume responsibility for undertaking actions pursuant to SECTION 8.5(e)(iii), the Buyer Indemnified Persons shall undertake, with the Sellers' oversight and participation, in good faith, to complete such actions in a reasonably cost effective manner;
- (D) to the extent there is a risk of litigation and sharing relevant information or communications between Sellers and Buyer Indemnified Persons could affect their status as privileged or otherwise protected, the Sellers and Buyer Indemnified Persons shall exchange information pursuant to the protection of a joint defense agreement and shall otherwise cooperate in order to facilitate the expeditious and cost-effective resolution of such claim; and
- (E) where reasonably necessary to achieve the purpose of SECTION 8.5(e)(II)(b), the Buyer Indemnified Parties shall, at Sellers' request, impose environmental deed or use restrictions, and/or accept Sellers' installation of engineering controls on the Owned Real Property, provided that such restrictions or controls are acceptable to the relevant Governmental Authority and are in compliance with Law and do not prevent or inhibit any uses of such Owned Real Property similar to those that occurred as of the Closing Date.

8.6 PROCEDURES. Notice of Losses. As soon as is reasonably practicable after the Sellers or the Buyer has actual knowledge of any Losses for which indemnification is available under SECTION 8.1(a) or SECTION 8.2(a) (a "CLAIM"), such party shall give written notice thereof (a "CLAIM NOTICE") to the other party. A Claim Notice must describe the Claim in reasonable detail, and must indicate the amount (estimated to the extent feasible) of the Loss that has been or will be suffered by the Indemnified Party. No delay in or failure to give a Claim Notice by the Indemnified Party to the Indemnifying Party will adversely affect any other rights or remedies that the Indemnified Party has under this Agreement, or alter or relieve the Indemnifying Party of its obligations to indemnify the Indemnified Party to the extent that such delay or failure has not materially prejudiced the Indemnifying Party. Each Indemnifying Party to whom a Claim Notice is given shall respond to any Indemnified Party that has given a Claim Notice (a "CLAIM RESPONSE") within thirty (30) days (the "RESPONSE PERIOD") after the date the Claim Notice is given. Any Claim Response must specify whether or not the Indemnifying Party disputes the Claim described in the Claim Notice. If any Indemnifying Party fails to give a Claim Notice. If any Indemnifying Party elects not to dispute a Claim described in a Claim Notice, whether by failing to give a timely Claim Response or otherwise, then the amount of such Claim will be conclusively deemed to be an obligation of such Indemnifying Party and such Indemnifying Party shall pay to the Indemnified Party within thirty (30) days after the last day of the applicable Response Period the amount to which such Indemnified Party is entitled.

- (b) Third Party Claims. If any Claim Notice identifies any Claim brought by a third person (a "THIRD PARTY CLAIM"), the Indemnifying Party will have the right, exercisable by written notice to the Indemnified Party within sixty (60) days of receipt of such Claim Notice, to assume the defense of such Third Party Claim, with counsel selected by the Indemnifying Party that is reasonably satisfactory to the Indemnified Party. If the Indemnifying Party assumes the defense of, and the full responsibility for paying or otherwise discharging such Third Party Claim, the Indemnifying Party will not be liable to the Indemnified Party for any legal expenses of other counsel or any other expenses subsequently incurred by such Indemnified Party in connection with the defense thereof and the Indemnified Party will have the right to participate at its own expense in the defense of such Third Party Claim. If the Indemnifying Party does not assume the defense of such Third Party Claim, the Indemnified Party and the Indemnifying Party and the Indemnifying Party and the Indemnifying Party sole cost and expense.
- (ii) The party responsible for the defense of any Third Party Claim (a "RESPONSIBLE PARTY") shall, to the extent reasonably requested by the other party, keep such other party informed as to the status of such Third Party Claim, including, without limitation, all settlement negotiations and offers. With respect to a Third Party Claim for which a Seller is the Responsible Party, the Buyer shall use all reasonable efforts to make available to such Seller all books and records of the Buyer relating to such Third Party Claim and shall cooperate with such Seller in the defense of the Third Party Claim. No settlement or compromise or consent to the entry of any judgment that does not relate solely to monetary damages arising from any such Third Party Claim may be effected (A) by the Indemnifying Party without the written consent of the Indemnified Party (which consent shall not be unreasonably withheld or delayed) unless all relief provided is paid or satisfied in full by the Indemnifying Party and the Indemnified Party receives a full release in respect of the Third Party Claim or (B) by the Indemnified Party without the consent of the Indemnifying Party (which consent shall not be unreasonably withheld or delayed). In no event shall an Indemnifying Party be liable for any settlement effected without its prior written consent.

ARTICLE IX MISCELLANEOUS

- 9.1 FEES AND EXPENSES. Except as otherwise provided in this Agreement and the Transaction Agreements, Cooper, on behalf of the Sellers, on the one hand, and the Buyer, on the other hand, shall bear its own expenses and the expenses of its Affiliates in connection with the preparation and negotiation of this Agreement and the consummation of the transactions contemplated by this Agreement. Cooper, on behalf of the Sellers, on the one hand, and the Buyer, on the other hand, shall bear the fees and expenses of any broker or finder retained by such party or parties and their respective Affiliates in connection with the transactions contemplated by this Agreement and the Transaction Agreements.
- 9.2 GOVERNING LAW. This Agreement will be construed under and governed by the Laws of the State of Delaware applicable to contracts made and to be performed in that State.
- 9.3 PROJECTIONS. In connection with the Buyer's investigation of the Sold Companies, the Subsidiaries and the Business, Buyer may have received, or may receive, from the Sellers, the Sold Companies, the Subsidiaries and/or their respective representatives certain projections and other forecasts for the Business, and certain business plan and budget information. Buyer acknowledges that (a) there are uncertainties inherent in attempting to make such projections, forecasts, plans and budgets, (b) Buyer is familiar with such uncertainties, (c) Buyer is taking full responsibility for making its own evaluation of the adequacy and accuracy of all estimates, projections, forecasts, plans and budgets so furnished to them and (d) Buyer will not assert any claim against the Sellers or any of their respective directors, officers, employees, Affiliates or representatives, or hold the Sellers or any such Persons liable, with respect thereto. Buyer acknowledges that the Sellers make no representation or warranty with respect to such projections, forecasts or plans and that the Sellers make only those representations and warranties explicitly set forth in ARTICLE III. Nothing in this SECTION 9.3 shall constitute a waiver of any rights of Buyer to assert a breach of representation or warranty hereunder in accordance with SECTION 5.5 and ARTICLE VIII or to assert a claim of fraud.
- 9.4 AMENDMENT. This Agreement may be amended, supplemented or modified, and any provision hereof may be waived with the written consent of the parties hereto.
- 9.5 ASSIGNMENT. Neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned by any party hereto without the prior written consent of the other parties; provided that Buyer may assign its rights or obligations hereunder to any Affiliate of Buyer (other than an Affiliate that is not a United States Person within the meaning of Section 7701(a)(30) of the Code) or, for collateral security purposes, to any lender providing financing to the Buyer, any of the Sold Companies or any of their Affiliates, without the prior written consent of the other parties hereto; provided, further, that no such assignment by Buyer shall relieve Buyer of its obligations hereunder. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of and be enforceable by, the parties hereto and their respective successors and permitted assigns, and is not intended to confer upon any Person other than the parties hereto and their respective successors and permitted assigns any rights or remedies hereunder (except for Indemnified Parties in ARTICLE VIII).
- 9.6 WAIVER. Any of the terms or conditions of this Agreement that may be lawfully waived may be waived in writing at any time by each party, at its sole discretion, that is entitled to the benefits thereof. Any waiver of any of the provisions of this Agreement by any party hereto will be binding only if set forth in an instrument in writing signed on behalf of such party. No failure to enforce any provision of or right under this Agreement will be deemed to or will constitute a waiver of such provision or right and no waiver of any of the provisions of this Agreement will be deemed to or will

constitute a waiver of any other provision hereof (whether or not similar) nor will such waiver constitute a continuing waiver.

9.7 NOTICES. (a) Any notice, demand, or communication required or permitted to be given by any provision of this Agreement will be deemed to have been sufficiently given or served for all purposes if (i) personally delivered,

(ii) sent by a nationally recognized overnight courier service to the recipient at the address below indicated, (iii) sent by registered or certified mail, return receipt requested, postage prepaid, or (iv) delivered by facsimile with confirmation of receipt:

If to the Buyer:

CSA Acquisition Corp. c/o The Cypress Group L.L.C. 65 East 55th Street New York, New York 10022

Attn: David P. Spalding Telephone: (212) 705-0150 Telecopy: (212) 705-0199

and c/o GS Capital Partners 2000, L.P. 85 Broad Street

New York, New York 10004 Attn: Gerald Cardinale Telephone: (212) 902-1000 Telecopy: (212) 357-5505

With a copy to:

Simpson Thacher & Bartlett LLP 425 Lexington Avenue New York, New York 10017 Attn: Gary I. Horowitz Telephone: (212) 455-2000 Telecopy: (212) 455-2502

and Fried, Frank, Harris, Shriver & Jacobson LLP One New York Plaza New York, New York 10004 Attn:Steven J. Steinman Telephone: (212) 859-8000 Telecopy: (212) 859-4000

If to any of the Sellers:

c/o Cooper Tire & Rubber Company 701 Lima Avenue Findlay, Ohio 45840

Attn: James E. Kline

Telephone: (419) 427-4757 Telecopy: (419) 420-6052

With a copy to:

Jones Day North Point 901 Lakeside Avenue Cleveland, Ohio 44114 Attn: Charles W. Hardin, Jr.

Telephone: (216) 586-3939

Telecopy: (216) 579-0212

or to such other address as any party hereto may, from time to time, designate in a written notice given in like manner.

- (b) Except as otherwise provided herein, any notice under this Agreement will be deemed to have been given (i) on the date such notice is personally delivered or delivered by facsimile, (ii) the next succeeding Business Day after the date such notice is delivered to the overnight courier service if sent by overnight courier, or (iii) five (5) Business Days after the date such notice is sent by registered or certified mail; provided that in each case notices received after 4:00 p.m. (local time of the recipient) will be deemed to have been duly given on the next Business Day.
- (c) For convenience only, the parties agree that all notices, consents, directions or other actions that may be given or taken hereunder by the Sellers may be given by Cooper on behalf of the Sellers pursuant to a written instruction or document duly executed by Cooper and that the Buyer will treat any such instrument or document as the action of the Sellers hereunder.
- 9.8 COMPLETE AGREEMENT. This Agreement (together with the Schedules, Exhibits and other agreements referenced herein), the Confidentiality Agreement and the other documents and writings referred to herein or delivered pursuant hereto contain the entire understanding of the parties with respect to the subject matter hereof and thereof and supersede all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter hereof and thereof.
- 9.9 COUNTERPARTS. This Agreement may be executed in one or more counterparts, all of which will be considered one and the same agreement and each of which will be deemed an original.
- 9.10 PUBLICITY. The Sellers and the Buyer will consult with each other and will mutually agree upon any publication or press release of any nature with respect to this Agreement or the transactions contemplated hereby and shall not issue any such publication or press release prior to such consultation and agreement except as may be required by Law or by obligations pursuant to any listing agreement with any securities exchange or any securities exchange regulation, in which case the party proposing to issue such publication or press release shall make all reasonable efforts to consult in good faith with the other party or parties before issuing any such publication or press release and shall provide a copy thereof to the other party or parties prior to such issuance.
- 9.11 HEADINGS. The headings contained in this Agreement are for reference only and do not affect in any way the meaning, interpretation or effect of this Agreement.
- 9.12 SEVERABILITY. Any provision of this Agreement that is invalid, illegal or unenforceable in any jurisdiction will, as to that jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof in

such jurisdiction or rendering that or any other provision of this Agreement invalid, illegal or unenforceable in any other jurisdiction.

- 9.13 THIRD PARTIES. Nothing herein expressed or implied is intended or will be construed to create, confer upon or give to any Person, other than the parties hereto and their permitted successors or assigns, any rights or remedies under or by reason of this Agreement.
- 9.14 FURTHER ASSURANCES. The parties shall execute such further instruments and take such further actions as may be reasonably necessary to carry out the intent of this Agreement. Each party hereto shall cooperate affirmatively with the other parties, to the extent reasonably requested by such other parties, to enforce rights and obligations herein provided.
- 9.15 CONSENT TO JURISDICTION; WAIVER OF JURY TRIAL. Each of the parties irrevocably submits to the exclusive jurisdiction of the courts of the State of Delaware and of the United States District Court for the District of Delaware, for the purposes of any Proceeding arising out of this Agreement or any transaction contemplated hereby. Each of the parties hereto irrevocably and fully waives the defense of an inconvenient forum to the maintenance of such Proceeding. Each of the parties further agrees that service of any process, summons, notice or document to such party's respective address listed above in one of the manners set forth in SECTION 9.7 will be deemed in every respect effective service of process in any such Proceeding, and waives any objection it might otherwise have to service of process under Law. Nothing herein will affect the right of any Person to serve process in any other manner permitted by Law. The parties hereto hereby irrevocably and unconditionally waive trial by jury in any Proceeding relating to this Agreement or any other agreement entered into in connection therewith and for any counterclaim with respect thereto.
- 9.16 SPECIFIC PERFORMANCE. The parties agree that irreparable damages would occur if any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent actual breaches or threatened breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court specified in SECTION 9.15, in addition to any other remedy to which they are entitled at law or in equity and without the necessity of proving damages or posting a bond or other security.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, each of the parties hereto has caused this Agreement to be executed by its duly authorized officer, in each case as of the date first above written.

COOPER TIRE & RUBBER COMPANY

By: /s/ Thomas A. Dattilo

Name: Thomas A. Dattilo

Title: Chairman of the Board, President

and Chief Executive Officer

COOPER TYRE & RUBBER COMPANY UK LIMITED

By: /s/ P.G. Weaver

Name: Philip G. Weaver

Title: Director

CSA ACQUISITION CORP.

By: /s/ David P. Spalding

Name: David P. Spalding Title: Chief Executive Officer

Exhibit 31.1

CERTIFICATION

- I, Thomas A. Dattilo, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Cooper Tire & Rubber Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
- a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 5, 2004

/s/ Thomas A. Dattilo

Thomas A. Dattilo Chairman, President and Chief Executive Officer

Exhibit 31.2

CERTIFICATION

- I, Philip G. Weaver, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q of Cooper Tire & Rubber Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
- (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- (b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- (c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 5, 2004

/s/ Philip G. Weaver

Philip G. Weaver Vice President and Chief Financial Officer

Exhibit 32

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of Cooper Tire & Rubber Company (the "Company") for the period ended September 30, 2004, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), each of the undersigned officers of the Company certifies, pursuant to 18 U.S.C.

Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to such officer's knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company as of the dates and for the periods expressed in the Report.

Date: November 5, 2004

/s/ Thomas A. Dattilo

Name: Thomas A. Dattilo
Title: Chief Executive Officer

/s/ Philip G. Weaver

Name: Philip G. Weaver Title: Chief Financial Officer

The foregoing certification is being furnished solely pursuant to 18 U.S.C. Section 1350 and is not being filed as part of the Report or as a separate disclosure document.

-1-

End of Filing



© 2005 | EDGAR Online, Inc.